

# TDS Telecom Winterhaven Last Mile Underserved Broadband Project Imperial County, California

**Proponent's Environmental Assessment** 

*Prepared by:* Tim Jordan Theresa Knoblock, M.A.

Prepared For: TDS Telecommunications Corporation Attn: Nate Stanislawski 525 Junction Road Madison, Wisconsin 53717

Prepared By: Tierra Right of Way Services, Ltd. 1575 East River Road, Suite 201 Tucson, Arizona 85719

April 21, 2015

# TDS Telecom Winterhaven Last Mile Underserved Broadband Project Imperial County, California

# **Proponent's Environmental Assessment**

Prepared by: Tim Jordan Theresa Knoblock, M.A.

*Prepared For:* California Public Utilities Commission 505 Van Ness Avenue San Francisco, California 94102

Prepared By: Tierra Right of Way Services, Ltd. 1575 East River Road, Suite 201 Tucson, Arizona 85719

# TABLE OF CONTENTS

List of	Acronyms	v
1.0	Summary	1
2.0	Project Purpose and Need	2
2.1	Overview	2
2.2	Project Objectives	2
3.0	Project Description	2
3.1	Project Location	2
3.2	Existing System	3
3.3	Proposed Project	3
3.4	Project Components	6
3.5	ROW Requirements	6
3.6	Construction	6
3.	6.1 Staging Areas	6
3.	6.2 Communications Line Installation	6
3.	6.3 Node Installation	8
3.	6.4 Surface Restoration	8
3.	6.5 Construction Workforce and Equipment	8
3.	6.6 Construction Schedule	9
3.7	Operation and Maintenance	9
3.8	Applicant-Proposed Measures	9
3.9	Key Permits and Approvals	12
4.0	Environmental Setting and Impact Assessment Summary	.13
4.1	Aesthetics	13
4.	1.1 Affected Environment	13
4.	1.2 Environmental Effects	14
4.	1.3 References	15
4.2	Agricultural Resources	16
4.	2.1 Affected Environment	16
4.	2.2 Environmental Effects	19
4.	2.3 References	20
4.3	Air Ouality and Greenhouse Gases	.20
4.	3.1 Affected Environment	20
4.	3.2 Environmental Effects	22
4.	3.3 References	.27
4.4	Biological Resources	.27
4.	4.1 Affected Environment	27
4.	4.2 Environmental Effects	35
4.	4.3 References	38
4.5	Cultural Resources	39
4.	5.1 Affected Environment	39
4.	5.2 Environmental Effects	56
4.	5.3 References	.58
4.6	Geology, Soils, and Seismic Potential	.60
4.	6.1 Affected Environment	.60
4.	6.2 Environmental Effects	.62

4.6.3	References	64
4.7 H	lazards and Hazardous Materials	65
4.7.1	Affected Environment	65
4.7.2	Environmental Effects	68
4.7.3	References	71
4.8 H	lydrology and Water Quality	71
4.8.1	Affected Environment	71
4.8.2	Environmental Effects	73
4.8.3	References	75
4.9 L	and Use and Planning	75
4.9.1	Affected Environment	75
4.9.2	Environmental Effects	77
4.9.3	References	78
4.10 N	lineral Resources	79
4.10.1	Affected Environment	79
4.10.2	Environmental Effects	80
4.10.3	References	81
4.11 N	loise	81
4.11.1	Affected Environment	
4.11.2	Environmental Effects	
4.11.3	References	
4.12 P	opulation and Housing	
4.12.1	Affected Environment	
4.12.2	Environmental Effects	
4.12.3	References	
4.13 P	ublic Services/Utilities and Service Systems	
4.13.1	Affected Environment	
4.13.2	Environmental Effects	
4.13.3	References	91
4.14 R	ecreation	91
4.14.1	Affected Environment	91
4.14.2	Environmental Effects	91
4.15 T	ransportation and Traffic	
4.15.1	Affected Environment	
4.15.2	Environmental Effects	
4.15.3	References	97
4.16 G	browth-Inducing and Cumulative Impacts	97
4.16.1	Growth-Inducing Impacts	97
4.16.2	Indirect Impacts	97
4.16.3	Cumulative Impacts	
4.16.4	References	
5.0 NEI	PA-Specific Resource Area Impact Analysis	
5.1 Ir	ndian Trust Assets	100
5.1.1	Affected Environment	
5.1.2	Environmental Consequences	
5.2 E	nvironmental Justice	101
5.2.1	Affected Environment	
5.2.2	Environmental Consequences	
	÷	

5.3	Socioeconomics	
5.	3.1 Affected Environment	
5.	3.2 Environmental Consequences	
5.4	References	
6.0	List of Preparers	
6.1	Tierra Right of Way Services, Ltd.	
6.2	TDS Telecommunications, Inc	

# LIST OF FIGURES

Figure 1. Project location	4
Figure 2. General project area	5

# LIST OF TABLES

Table 3.1	. Cable Installation Lengths	6
Table 3.2	. Canal Bore Locations	7
Table 3.3	. Permits and Approvals Required for Construction	12
Table 4.1	. ICAPCD Significance Thresholds for Emissions from Construction Activities	20
Table 4.2	. Modeled Construction Schedule	25
Table 4.3	. Estimated Daily Construction Emissions—Criteria Pollutants	25
Table 4.4	. Estimated Greenhouse Gas Emissions	26
Table 4.5	. Special Status Species with the Potential to Occur in the Study Area	34
Table 4.6	. Bird Conservation Region 33 Migratory Bird List	35
Table 4.7	. Common Sound Levels	82
Table 4.8	. Existing Noise Sources in the Project Area	83
Table 4.9	. Construction Equipment Noise Levels	85
Table 5.1	BIA NEPA Resource Areas	99

# LIST OF APPENDICES

Appendix A. Project Plans	A.1
Appendix B. CalEEMod Results	B.1
Appendix C. Biological Resources Evaluation	C.1
Appendix D. Waterway Delineation and Assessment Report	D.1
Appendix E. Class III Cultural Resources Survey Report and Cultural Resources	
Correspondence	E.1
Appendix F. Allands Data and Research, Inc., Report	F.1

# LIST OF ACRONYMS

AIRFA	American Indian Religious Freedom Act
AMSL	Above Mean Sea Level
APE	Area of Potential Effect
APM	Applicant's Proposed Measure
ARPA	Archaeological Resources Preservation Act
ARPI	Arizona Rural Policy Institute
BIA	Bureau of Indian Affairs
BLM	Bureau of Land Management
BMP	Best Management Practice
BOR	Bureau of Reclamation
BRE	Biological Resources Evaluation
BWD	Bard Water District
САА	Clean Air Act
CAAQS	California Ambient Air Quality Standards
CARB	California Air Resources Board
CASF	California Advanced Services Fund
CBG	Census Block Group
ССАА	California Clean Air Act
CCR	California Code of Regulations
CDFA	California Department of Food and Agriculture
CDFG	California Department of Fish and Game
CDFW	California Department of Fish and Wildlife
CDMV	California Department of Motor Vehicles
CDOC	California Department of Commerce
CDOT	California Department of Transportation
CDWR	California Department of Water Resources
CEQA	California Environmental Quality Act
CERCLA	Comprehensive Environmental Response, Compensation, and Liability Act
CESA	California Endangered Species Act
CFR	Code of Federal Regulations
CGC	California Government Code
CGP	Construction General Permit
CHRIS	California Historic Resources Information System
CIPC	California Invasive Plant Council
CNDDB	California Natural Diversity Database
CNPS	California Native Plant Society
CPUC	California Public Utilities Commission
CRHR	California Register of Historic Resources
CUPA	Certified Unified Program Agency

dB	Decibel
dB A	Decibel A weighted
	Digital Loop Carrier
DEC	Digital Loop Carrier
DTSC	Department of Toxic Substances Control
	Environmentel Assessment
EA FEZ	Environmental Assessment
EID	Earliquake Fault Zone
	Environmental Impact Statement
	Environmental Drotection Agency
	Endensered Species Act
ESA	Endangered Species Act
	Federal Aviation Administration
FHWA	Federal Highway Administration
FMMP	Farmland Mapping and Monitoring Program
FONSI	Finding of No Significant Impact
FWS	U.S. Fish and Wildlife Service
GHG	Greenhouse Gas
GLO	General Land Office
GP	General Plan
GWP	Global Warming Potential
HAZWOPER	Hazardous Waste and Operations and Emergency Response
НСР	Habitat Conservation Plan
HDPE	High Density Polyethylene
HUC	Hydrologic Unit Code
ICAPCD	Imperial County Air Pollution Control District
IID	Imperial Irrigation District
IPAC	Information, Planning, and Conservation
IS	Initial Study
ITA	Indian Trust Asset
LCR MSCP	Lower Colorado River Multi-Species Conservation Plan
LRA	Local Responsibility Area
MBPS	Megabits Per Second
MBTA	Migratory Bird Treaty Act
MND	Mitigated Negative Declaration
MRZ	Mineral Resource Zone
MSDS	Material Safety Data Sheet
MUTCD	Manual on Uniform Traffic Control Devices
NAAQS	National Ambient Air Quality Standards
NAGPRA	Native American Graves Protection and Repatriation Act
NAHC	Native American Heritage Council
NEPA	National Environmental Policy Act

NHPA	National Historic Preservation Act
NMFS	National Marine Fisheries Service
NPDES	National Pollutant Discharge Elimination System
NPL	National Priorities List
NPS	National Park Service
NRCS	National Resource Conservation Service
NRHP	National Register of Historic Places
O&M	Operations and Maintenance
OHWM	Ordinary High Water Mark
OSHA	Occupational Safety and Health Administration
PEA	Proponent's Environmental Assessment
PM10, 2.5	Particulate Matter 10, and 2.5 Micron
PRC	California Public Resources Code
RCRA	Resource Conservation and Recovery Act
ROG	Reactive Organic Gas
ROU	Right of Use
ROW	Right of Way
RWQCB	Regional Water Quality Control Board
SCIC	South Central Information Center
SDR	Standard Dimension Ratio
SED	Substitute Environmental Document
SHPO	State Historic Preservation Office
SMARA	Surface Mining and Reclamation Act
SMGB	State Mining and Geology Board
SPRR	Southern Pacific Railroad
SPVUSD	San Pasqual Valley Unified School District
SWPPP	Stormwater Pollution Prevention Plan
SWRCB	State Water Resources Control Board
TDS	TDS Telecom, Inc.
UPRR	Union Pacific Railroad
USACE	U.S. Army Corps of Engineers
USC	United States Code
USDOT	U.S. Department of Transportation
USGS	U.S. Geological Survey
UST	Underground Storage Tank
VDSL2	Very High Bit Rate Digital Subscriber Line, 2nd Generation
WDR	Waste Discharge Requirement
WRCC	Western Regional Climate Center
WUS	Waters of the U.S.
YCIPTA	Yuma County Intergovernmental Public Transportation Authority
YCWUA	Yuma County Water Users' Association

# 1.0 SUMMARY

Winterhaven Telephone Company d.b.a. TDS Telecom, Inc. (TDS), proposes to construct the Winterhaven Last Mile Underserved Broadband Project (the project), which would extend high-speed internet service to an area approximately 40.59 km<sup>2</sup> (15.67 miles<sup>2</sup>) in size, including the community of Winterhaven, a portion of the Fort Yuma-Quechan Indian Reservation, and other areas of unincorporated Imperial County in southeastern California.

The proposed project involves the construction of a second-generation, very-high-bit-rate digital subscriber line (VDSL2) fiber-optic network capable of 25 MBPS/5 MBPS (download/upload) speeds. In total, approximately 24.65 km (15.31 miles) of new fiber-optic cable would be buried within protective conduit along existing roads in the project area, and approximately 2.25 km (1.40 miles) of existing buried copper line would be used in the new system.

Public involvement has been an integral part of the project since its inception. The Fort Yuma– Quechan Tribal Council (the Council) was notified of the project prior to TDS's submittal of a California Advanced Services Fund (CASF) grant application to the California Public Utilities Commission (CPUC) on February 1, 2013, and the Council released a letter in support of TDS's efforts to secure funding on January 24, 2013. In addition to coordination with the Council, two public meetings were held at the Paradise Casino on July 17, 2014, where presentations describing the project were given to members of the public and questions and concerns regarding the project were discussed.

Because a portion of the project area is located on the Fort Yuma–Quechan Indian Reservation and a right-of-way (ROW) grant from the Bureau of Indian Affairs (BIA) is required prior to construction, the project must demonstrate National Environmental Policy Act (NEPA) compliance. This environmental compliance is in addition to California Environmental Quality Act (CEQA) compliance that is required by the project's partial funding source, the CPUC. Consequently, this Proponent's Environmental Assessment (PEA) is a combined environmental document that complies with NEPA requirements for the preparation of an Environmental Analysis (EA) and with CEQA requirements for an Initial Study (IS). An agreement between the CPUC and BIA was developed where CPUC would be the State lead agency for the project and BIA would serve as the Federal lead. A second Federal agency, the Bureau of Reclamation (BOR), is involved with the project because irrigation canals under their jurisdiction would be crossed by the project alignments; the BOR has agreed to act as a cooperating agency.

This PEA includes the information required by the CPUC PEA Guidelines (*CPUC Information and Criteria List*, Appendix B, Section V) as well as an impact analysis for resource areas specified by the BIA in the *BLA NEPA Handbook*. This PEA includes a discussion of the purpose and need for the proposed project in Section 2; the project description in Section 3; the environmental setting, impacts, mitigation measures, and cumulative and growth-inducing impacts in Section 4; and an analysis of NEPA-specific resource areas in Section 5. No alternatives are being considered in this document other than the "No-Action Alternative." The proposed project alignment was chosen for analysis because of the presence of existing roads, ROWs, and supporting facilities. Other locations would require the preparation of new sites, which would result in increased environmental impacts. Under NEPA guidelines, in an EA where there are no unresolved conflicts with respect to

alternative uses of available resources, only the Proposed Action needs to be considered (43 CFR 46.310(b)).

No cumulative impacts, growth-inducing effects, or indirect effects were identified for the proposed project. Lists of references are included after each resource area in Sections 4 and 5, and a list of the PEA preparers is included in Section 6 of this document.

# 2.0 PROJECT PURPOSE AND NEED

# 2.1 Overview

The CPUC approved funding in the amount of \$2,063,967.00 from the CASF for the Winterhaven Last Mile Underserved Broadband Project. The project, which would be constructed and operated by the Winterhaven Telephone Company d.b.a. TDS, would extend high-speed internet service to an area approximately 40.59 km<sup>2</sup> (15.67 miles<sup>2</sup>) in size that includes the community of Winterhaven and other areas of unincorporated Imperial County, California, including a portion of the Fort Yuma–Quechan Indian Reservation.

TDS has been building broadband networks utilizing technologies similar to this project for the past decade and currently provides both voice and limited broadband services in the project area. The project would implement second-generation VDSL2 technology at its central office as well as at numerous existing and proposed digital loop carrier (DLC) sites in order to provide high-speed internet service across the project area. The U.S. Census Block Groups (CBGs) impacted by the project area include 060259400001, 060259400002, and 060259400003.

TDS has targeted the area for broadband deployment because of existing customer demand and because they determined that the project is economically feasible with the assistance of a CASF grant of \$2,063,967.00 (60 percent of the project costs) to match TDS's funding of \$1,375,978.00. When completed, the project would reach an estimated 961 households at maximum advertised speeds of 25 MBPS/5 MBPS, which is above the served threshold of 6 MBPS/1.5 MBPS.

TDS estimates the project would initially yield 233 potential subscriber households in the project area. In addition to residential customers, the project area includes five anchor institutions which may benefit from the project, including San Pasqual Valley High, San Pasqual Valley Elementary, Bill M. Manes High, San Pasqual Middle School, and San Pasqual Vocational Academy.

# 2.2 Project Objectives

The proposed project's objective is to make affordable broadband Internet services available to currently underserved areas in Imperial County, including a portion of the Fort Yuma–Quechan Reservation.

# 3.0 **PROJECT DESCRIPTION**

# 3.1 Project Location

The project area is located in southeastern Imperial County, California, just north of Yuma, Arizona, and the Colorado River. Baseline Road, which runs north-south, marks the boundary between the Fort Yuma–Quechan Reservation and private land; the Reservation is west of Baseline Road, and

private land lies to the east (Figures 1 and 2). The southern edge of the project area is roughly bounded by the Union Pacific Railroad (UPRR) tracks, the community of Winterhaven, and the Paradise Casino on Picacho Road. The Cocopah Canal runs along the eastern boundary of the project area and the community of Bard is located at the northeastern limits of the project area. Stalnacker and Ross Roads, along with the community of Ross Corner, make up the approximate northern limits of the project area, and the western edge of the project area is near Arnold Road, where the road approaches the UPRR. Specifically, the project area is located in portions of Section 2, Township 15 South, Range 24 East; Sections 11, 14, and 21–27, Township 16 South, Range 22 East; and Sections 4, 5, 7–9, 18, and 19 Township 16 South Range 23 East, San Bernardino Baseline and Meridian (SBB&M), as depicted on the Araz, Bard, Yuma East, and Yuma West, AZ/CA, 7.5-minute U.S. Geological Survey (USGS) topographic quadrangle maps.

# 3.2 Existing System

TDS's existing land-based telecommunications system in the project area consists of direct-buried copper lines and is able to provide basic telephone and 911 services. The copper lines in the project area are connected to one of four DLCs, the first of which is located at the TDS Central Office in Winterhaven and serves the 35100 Digital Serving Area (DSA). The second DLC, located just north of the Paradise Casino on Picacho Road, serves the 35109 DSA, and the third DLC, located in Bard, serves the 35102 DSA. The fourth DLC is located just east of the intersection of Arnold and Flood Roads and serves the 35103 DSA. Dial-up Internet services are available in all four DSAs, but the data transfer rate is limited to a non-broadband speed of 56 kbps under the International Telecommunications Union V92 standard.

# 3.3 Proposed Project

The proposed project involves the construction of a second-generation VDSL2 fiber-optic network capable of 25 MBPS/5 MBPS (download/upload) speeds. In total, approximately 24.65 km (15.31 miles) of new fiber-optic cable would be buried within protective conduit along existing roads in the project area, and approximately 2.25 km (1.40 miles) of existing buried copper line would be used to connect a proposed DLC site on Arnold Road to the new system. A summary of the associated lengths to be installed on and off the Fort Yuma-Quechan Reservation can be found in Table 3.1. The buried line installation, which consists of the telecommunications cable and its protective conduit, would be performed using plowing construction techniques, and a directional boring machine would be used to install the line at canal and road crossings. Ancillary equipment to be installed includes 10 new equipment cabinets at DLC sites that would serve as connecting "nodes" for customers, splice boxes, and line markers. The equipment cabinets would be approximately 0.6 by 1.0 by 1.2 m (2.0 by 3.0 by 4.0 feet) in size and would be installed on top of buried vaults within an approximately 6-m-square (20-foot-square) area. Splice boxes are small, rectangular metal enclosures that would be installed between lengths of cable. Line markers, which would be installed at intervals of approximately five per mile, are approximately 1.2 m (4.0 feet) tall and made of flexible fiberglass. Electrical power for the new DLC sites would be provided by existing aerial distribution lines located immediately adjacent to each site. Project plans are included in Appendix A.



Figure 1. Project location.



Figure 2. General project area.

Installation	Length (m)	Length (km)	Length (feet)	Length (miles)
On-Reservation	10,139	10.14	33,264	6.30
Off-Reservation	14,507	14.51	47,595	9.01
Total	24,646	24.65	80,859	15.31

Table 3.1. Cable Installation Lengths

# 3.4 Project Components

The proposed project would consist of the following components:

- Installation of approximately 24,646 m (80,860 feet) of 96-count, shielded fiber-optic telecommunications cable within protective 3.20-cm-diameter (1.25-inch-diameter), high-density polyethylene (HDPE), standard dimension ratio (SDR)–11 conduits.
- Installation of 10 equipment cabinets on top of buried epoxy composite vaults at DLC sites that would serve as telecommunications nodes.
- Clean-up and site restoration following construction.

# 3.5 ROW Requirements

The portions of the proposed project located on Tribal land are located on allotments that would require ROW grants from the associated landowners prior to the telecommunications line installation. The remaining portions of the project located on non-Tribal land would require County road ROW encroachment permits from Imperial County.

# 3.6 Construction

# 3.6.1 Staging Areas

No staging areas would be required in the project area during construction of the proposed project. All equipment and material staging would take place either at the Winterhaven Central Office or individual contractor's off-site yards.

# 3.6.2 Communications Line Installation

The line installation would be performed in three steps. First, protective conduit for the fiber-optic cable would be installed by either plowing or directional boring construction methods. Second, the conduit would be prepared for receiving the fiber-optic cable by "pigging." This process involves forcing a cleaning sponge, or "pig," through the conduit using compressed air to clean and lightly lubricate the inside of the conduit. Third, the fiber-optic cable would be "blown" through the conduit using compressed air. The total combined ground disturbance associated with the project, including both the plowed and bored installations, would not exceed an area approximately 5.1 ha (12.5 acres) in size.

# 3.6.2.1 Plowed Installation

Approximately 20,757 m (68,101 feet) of the proposed installations would be performed using plowing construction techniques. Plowed conduit is installed using a track-type bulldozer equipped with a specialized single ripper that loosens the soil along the installation path. Conduit is fed either from the plow bulldozer or from a separate truck-mounted reel through a plow chute attached to

the ripper and laid directly at a nominal depth of 1.0 m (3.3 feet). A compaction machine follows directly behind the plow bulldozer and restores the ground surface to its original contour. The installation path may be "pre-ripped" by a second bulldozer, if necessary, to loosen the soil in areas where subsurface rock or other buried obstructions may be present. This second bulldozer may also, in some cases, be attached to the plow bulldozer to provide additional pulling power for the plowing operation. Ground disturbance associated with the plowed installation would be limited to an approximately 2.4-m-wide (8.0-foot-wide) corridor.

# 3.6.2.2 Directional Bore Installation

Approximately 3,889 m (12,758 feet) of the proposed installations would be performed using directional boring construction techniques. Directional boring is a method used to install utility lines under waterways, roads, and other areas where the avoidance of surface disturbance is desirable (Figure 3). Directional boring machines are essentially horizontal drilling rigs with a steerable drill bit. Each bore begins with creating a pilot hole, where the drill bit is guided by the operator as it progresses along the desired boring path. After boring the pilot hole, conduit is attached to the end of the drill string and the conduit is pulled back through the bore.

Two boring pits for bore ingress and egress would be required for each canal and road crossing installation, one on each side of the canal or road. These bore pits would be approximately 2.4 m (8.0 feet) square and would be located at varying distances from the canals or roads. The depth of the bore would be a minimum of 1.5 m (5.0 feet) below the bottom of the canals and roads, and the bore lengths would be variable. The bores would be of sufficient diameter to accommodate the 3.20-cm-diameter (1.25-inch-diameter) conduit and would be drilled using drilling fluid "mud" consisting of sodium bentonite and water. The drilling mud serves two purposes: first, it lubricates the drill bit; second, it seals the bore with an impermeable layer of sodium bentonite, keeping the bore from collapsing. As drilling mud accumulates in the bore pits, it would be evacuated using a trailer-mounted "mud-sucker" pump for reuse and/or appropriate disposal. In some cases, such as directional bores located beneath earthen canals, the entire bore would be grouted after conduit installation with a drilling mud/concrete mixture to provide a solid barrier that would prevent seepage flow from the canal in accordance with BOR guidelines.

Following the installation of the conduit beneath the canal or road, the bore pits would be filled in, compacted, and the ground surface restored to its original contour. The locations of all canal bores associated with the project are summarized in Table 3.2. Ground disturbance associated with the bored conduit installations would occur within the same 2.4-m-wide (8.0-foot-wide) corridor as the plowed installations.

Map No.	Canal Name	Location	<b>Canal Width</b> <sup>a</sup>
1	Reservation Main Drain	Stalnacker Road	20.5 m (67 feet)
2	Unnamed canal	Fisher and Parkman Roads	3.6 m (12 feet)
3	Reservation Main Drain	Fisher Road	19.6 m (64 feet)
4	Hopi Canal	Bard and Whitmore Roads	6.3 m (21 feet)
5	Cocopah Canal	Ross Road	9.0 m (30 feet)
6	Unnamed canal	Fisher and Ross Roads	5.3 m (17 feet)

Table 3.2. Canal Bore Locations

Map No.	Canal Name	Location	Canal Width <sup>a</sup>
7	Papago Canal	Perez Road	4.5 m (15 feet)
8	Pima Canal	Haughtelin and Perez Roads	4.5 m (15 feet)
9	Cocopah Canal	Flood and Arnold Roads	7.0 m (23 feet)
10	Navajo Canal	Picacho and Jackson Roads	7.3 m (24 feet)
11	Reservation Main Drain	Picacho Road	27.3 m (90 feet)
12	Pima Canal	Picacho and Haughtelin Roads	3.7 m (12 feet)
13	Pueblo Canal	Picacho and Indian Rock Roads	3.6 m (12 feet)
14	Cocopah Canal	Picacho Road	8.3 m (27 feet)
15	Reservation Main Drain	Arnold Road	27.3 m (90 feet)
16	Yuma Main Canal	Arnold Road	46.0 m (151 feet)
17	Walapai Canal	Arnold Road	2.4 m (8 feet)

<sup>a</sup> Includes width of canal and any associated vegetation at edges of canal (see Section 4.4 for details).

### 3.6.3 Node Installation

Communications node (DLC) installation would begin with excavating a hole measuring 1.0 m long by 2.0 m wide by 1.2 m deep (3.0 feet long by 6.0 feet wide by 4.0 feet deep) using a backhoe. An epoxy composite vault would then be placed, backfilled, and covered with gravel after the subsurface connections to the associated telecommunications lines are made. The vault cover would then be installed, onto which an equipment cabinet would be bolted to serve as the connecting point between the new fiber-optic lines and customers' copper service drops.

#### 3.6.4 Surface Restoration

Following the telecommunications line and DLC installations, TDS and/or their contractors would promptly perform site clean-up and surface restoration. Clean-up would include removing all construction debris, and surface restoration would involve returning the surface contours of disturbed areas to their pre-construction condition.

#### 3.6.5 Construction Workforce and Equipment

Preliminary construction workforce estimates indicate that one plow crew, two directional-boring crews, one splice crew, and one clean-up crew would be required to install the telecommunications lines associated with the project; each of these crews would consist of three to four workers. An additional two-person crew would be needed to construct the node sites. All work crews are anticipated to work standard eight-hour days, five days a week. Construction equipment necessary to complete the installations is anticipated to consist of:

- Two D5-class bulldozers for the plowed installations.
- Two directional boring machines (Vermeer D20x22 S3 or equivalent).
- Two trailer-mounted mud-sucker pumps for drilling mud evacuation and recovery.
- Two backhoes (Case 580x or equivalent).
- One medium-duty (5-ton), spray-bar-equipped water truck for dust control.
- One medium-duty (2.5–5.0-ton) flatbed truck for reel and underground vault delivery.
- Two trailer-mounted air compressors for conduit pigging and blowing fiber-optic line.

• Three to four light-duty pickups (0.5- and 0.75-ton) for crew transport.

# 3.6.6 Construction Schedule

The anticipated start date for the proposed project is mid-January 2016, and construction would take approximately two months.

# 3.7 Operation and Maintenance

Operation and maintenance (O&M) activities associated with the new telecommunications network are expected to be minimal because, once installed, fiber-optic cable is essentially maintenance-free. Occasional visits by TDS technicians to the DLC sites would be required to disconnect and connect customers, and air filters in the DLC equipment cabinets would require periodic inspections and cleaning. None of these O&M activities would involve ground disturbance.

# 3.8 Applicant-Proposed Measures

TDS has incorporated Applicant-Proposed Measures (APMs) into the proposed project to avoid significant impacts on the environment and to reduce any potential impacts to less-than-significant levels. Implementation of theses APMs, together with the limited nature of TDS' construction activities and their location along highly disturbed County road ROWs, ensures that the proposed project would not significantly affect the environment.

**APM AQ-1**: TDS will require all construction contractors to implement the following Imperial County Air Pollution Control District (ICAPCD) standard measures for fugitive PM10 control:

- All disturbed areas, including bulk-material storage that is not being actively utilized, shall be effectively stabilized, and visible emissions shall be limited to no greater than 20 percent opacity for dust emissions by using water, chemical stabilizers, dust suppressants, tarps, or other suitable material, such as vegetative ground cover.
- All on- and off-site unpaved roads would be effectively stabilized, and visible emissions shall be limited to no greater than 20 percent opacity for dust emissions by paving, chemical stabilizers, dust suppressants, and/or watering.
- All unpaved traffic areas 1 acre or more in size with 75 or more average vehicle trips per day would be effectively stabilized, and visible emissions shall be limited to no greater than 20 percent opacity for dust emissions by paving, chemical stabilizers, dust suppressants, and/or watering.
- The transport of bulk materials shall be completely covered unless six inches of freeboard space from the top of the container is maintained with no spillage or loss of bulk material. In addition, the cargo compartment of all haul trucks is to be cleaned and/or washed at the delivery site after removal of bulk material.
- All track-out and carry-out would be cleaned at the end of each workday or immediately when mud or dirt extends a cumulative distance of 15 linear m (50 linear feet) or more onto a paved road within an urban area.
- Bulk material shall be stabilized prior to movement or at points of transfer with the application of sufficient water, the application of chemical stabilizers, or by sheltering or enclosing the operation and transfer line.
- The construction of any new unpaved road is prohibited within any area with a population of 500 or more, unless the road meets the definition of a temporary

unpaved road. Any temporary unpaved road shall be effectively stabilized, and visible emissions shall be limited to no greater than 20 percent opacity for dust emissions by paving, chemical stabilizers, dust suppressants, and/or watering.

In addition, the following discretionary measures would be implemented:

- Watering of exposed soil with adequate frequency for continued moist soil.
- Replacing ground cover in disturbed areas as quickly as possible.
- Installing an automatic sprinkler system on all soil piles.
- Vehicle speed for all construction vehicles shall not exceed 15 mph on any unpaved surface at the construction site.

**APM BIO-1:** All irrigation canals in the project area will be bored beneath and avoided during construction.

**APM BIO-2:** Bore pits will be placed a minimum distance of 5 m (16 feet) beyond either the top of the canal bank or the maximum extent of any vegetation present along the canal's margin.

**APM BIO-3:** All agricultural fields will be avoided during construction.

**APM BIO-4:** No trees will be removed during project construction. If vegetation trimming is required to complete the installations, it will be kept to the absolute minimum necessary.

**APM BIO-5:** All equipment and vehicles will be thoroughly cleaned to remove dirt and weed seeds prior to being transported or driven to or from the project area.

**APM CR-1:** The Pilot Knob-Tap Drop 4 161kV Transmission Line will be avoided during construction.

**APM CR-2:** The UPRR will be bored beneath and avoided during construction.

**APM CR-3:** All construction activities will be monitored by a qualified archaeologist and/or Tribal member. If buried cultural resources are inadvertently discovered during ground-disturbing activities, work will stop in that area and within 100 feet of the discovery until the archaeological monitor can assess the significance of the find and, if necessary, develop appropriate measures in consultation with the CPUC, SHPO, and other appropriate agencies.

**APM CR-4:** If human remains are discovered or recognized in any location other than a dedicated cemetery, TDS will suspend further excavation or disturbance of the site and any nearby areas reasonably suspected to overlie adjacent human remains until the coroner of the County has been informed and has determined that no investigation of the cause of death is required.

**APM CR-5:** If human remains of Native American origin are discovered on Federal land during ground-disturbing activities, pursuant to the Native American Graves Protection and Repatriation Act (NAGPRA), the contractor will:

• Notify the County coroner or the Sheriff;

- Notify, in writing, the responsible Federal agency; and
- Cease activity in the area of discovery and protect the human remains.

**APM CR-6:** In the event that fossil remains are encountered, either by the cultural resources monitor or by construction personnel, qualified paleontological specialists will be contacted. Construction within 30.5 m (100.0 feet) of the find in non-urban areas and 15 m (50 feet) in urban areas will be temporarily halted or diverted until a qualified vertebrate paleontologist examines the discovery.

**APM GEO-1:** TDS will require the contractor to manage construction-induced sediment and excavated spoils in accordance with the requirements of the State Water Resources Control Board (SWRCB) and Environmental Protection Agency (EPA) National Pollution Discharge Elimination System (NPDES) permits for storm water runoff associated with construction activities.

**APM GEO-2:** Prior to the onset of construction, TDS or its authorized contractor will complete a Stormwater Prevention Pollution Plan (SWPPP) that outlines Best Management Practices (BMPs) to control discharges from construction areas.

**APM GEO-3:** No construction-related materials, wastes, spills, or residues will be discharged from the project.

**APM GEO-4:** The staging of construction materials, equipment, and excavation spoils will be performed outside of drainages.

**APM GEO-5:** Excavated or disturbed soil will be kept within a controlled area surrounded by a perimeter barrier that may entail silt fence, hay bales, straw wattles, or a similarly effective erosion-control technique that prevents the transport of sediment from a given stockpile.

**APM GEO-6:** All stockpiled material will be covered or contained in such a way that eliminates offsite runoff from occurring.

**APM GEO-7:** Upon completion of construction activities, excavated soil will be replaced and graded so that post-construction topography and drainage matches pre-construction conditions.

**APM GEO-8:** Surplus soil will be transported from the site and disposed of appropriately.

**APM HAZ-1:** TDS will ensure proper labeling, storage, handling, and use of hazardous materials in accordance with best management practices and the Occupational Safety and Health Administration's (OSHA's) Hazardous Waste Operations and Emergency Response (HAZWOPER) requirements.

**APM HAZ-2:** TDS will ensure that employees are properly trained in the use and handling of hazardous materials and that each material is accompanied by a material safety data sheet (MSDS).

**APM HAZ-3:** Any small quantities of hazardous materials stored temporarily in staging areas will be stored on pallets within fenced and secured areas and protected from exposure to weather. Incompatible materials will be stored separately, as appropriate.

**APM HAZ-4:** All hazardous waste materials removed during construction will be handled and disposed of by a licensed waste disposal contractor and transported by a licensed hauler to an appropriately licensed and permitted disposal or recycling facility, to the extent necessary to ensure the area can be safely traversed.

**APM HAZ-5:** Significant releases or threatened releases of hazardous materials will be reported to the appropriate agencies.

**APM NOI-1**: All construction equipment operation shall be limited to the hours of 7 a.m. to 7 p.m. Monday through Friday and 9 a.m. to 5 p.m. on Saturday. No construction operations shall occur on Sunday or holidays.

**APM TRA-1:** TDS will require the project contractor to obtain all necessary local, State, and BIA road encroachment permits prior to construction and will comply with all the applicable conditions of approval.

**APM TRA-2:** As deemed necessary by the applicable jurisdiction, the road encroachment permits may require the contractor to prepare a traffic control plan in accordance with professional engineering standards prior to construction.

**APM TRA-3:** TDS will develop circulation and detour plans to minimize impacts to local street circulation. This will include the use of signing and flagging to guide vehicles through and/or around the construction zone.

**APM TRA-4:** TDS will schedule truck trips outside of peak morning and evening commute hours.

**APM TRA-5:** TDS will limit lane closures during peak hours to the extent possible.

**APM TRA-6:** TDS will include detours for bicycles and pedestrians in all areas potentially affected by project construction.

**APM TRA-7:** TDS will install traffic-control devices as specified in the *California Department of Transportation Manual of Traffic Controls for Construction and Maintenance Work Zones.* 

**APM TRA-9:** The contractor will coordinate with local transit agencies for the temporary relocation of routes or bus stops in work zones as necessary.

# 3.9 Key Permits and Approvals

Key permits and approvals necessary for the construction of the proposed project are presented below in Table 3.3.

Agency	Permit/Approval	Status	
California Public Utilities Commission	MND	Pending	
Bureau of Indian Affairs	FONSI and ROW Grant	Pending	

Table 3.3. Permits and Approvals Required for Construction

Agency	Permit/Approval	Status	
Bureau of Reclamation	ROU Authorization	Pending	
Imperial County	Encroachment Permit	Pending	
Union Pacific Railroad	Encroachment Permit	Pending	

*Key:* FONSI = Finding of No Significant Impact (NEPA), MND = Mitigated Negative Declaration (CEQA), ROU = Right of Use.

# 4.0 ENVIRONMENTAL SETTING AND IMPACT ASSESSMENT SUMMARY

This PEA is a combined environmental document that complies with NEPA requirements for the preparation of an EA and with CEQA requirements for an IS. It is important to note that the use of the term "significant" in the resource sections below differs under these two laws. Under NEPA, an EA is prepared to determine whether a "Proposed Action" would have any "significant effects on the quality of the human environment," and significance is defined in terms of the impact's context and intensity. If a Proposed Action would result in one or more significant impacts, an Environmental Impact Statement (EIS) rather than an EA must be prepared. Under CEQA, each "significant effect on the environment" resulting from a "proposed project" must be identified in an IS along with ways to mitigate these effects. The manner in which the differences between the two processes are addressed in this PEA must therefore take into account that NEPA does not require mandatory findings of significance in an EA, and that some impacts determined to be significant under CEQA may not necessarily be determined significant under NEPA.

Environmental factors and mandatory findings of significance under CEQA are presented below. NEPA-specific resource areas that do not require consideration under CEQA are presented in Section 5.

# 4.1 Aesthetics

#### 4.1.1 Affected Environment

#### 4.1.1.1 Regulatory Setting

#### Federal

There are no applicable Federal regulations or policies related to aesthetics.

#### State

#### California Scenic Highway Program

In 1963, the California Legislature created the Scenic Highway Program to preserve and protect scenic highway corridors from changes that would diminish the aesthetic value of lands adjacent to the highways. The state regulations and guidelines governing the Scenic Highway Program are found in Section 260 et seq. of the Streets and Highways Code. A highway may be designated as scenic depending on how much of the natural landscape can be seen by travelers, the scenic quality of the landscape, and the extent to which development intrudes upon the travelers' enjoyment of the view.

#### Local

The Imperial County General Plan (ICGP) has goals and objectives related to visual resources (Imperial County 2008). These goals and objectives are listed below.

- Conservation and Open Space Element Goal 7: The aesthetic character of the region shall be protected and enhanced to provide a pleasing environment for residential, commercial, recreational, and tourist activity.
  - Objective 7.1—Encourage the preservation and enhancement of the natural beauty of the desert and mountain landscape.
- GP Circulation and Scenic Highways Goal 4: The County shall make every effort to develop a circulation system that highlights and preserves the environmental and scenic amenities of the area.

# 4.1.1.2 Project Setting

According to the ICGP, important visual resources include desert areas, sand hills, mountains, and the Salton Sea. Scenic visual resources that are visible from the project area are limited to mountains to the north and northwest.

Four areas within the County have potential as State-designated scenic highways, including Interstate 8 (I-8) from between the San Diego County line and its junction with State Route 98; this segment is not located in or near the project area.

The topography of the project area is relatively flat, allowing for mostly unobstructed views. The proposed project alignment is located along existing roads in an area used primarily for agriculture. Accordingly, the views in the project area are dominated by agricultural fields and irrigation canals with views of distant mountains to the north and northwest, primarily. In addition to roads, other linear features in the project area include aerial electrical distribution lines that parallel most of the roads in the project area. Scattered rural residences and associated planted trees are also present in rural portions of the project alignment. Within the community of Winterhaven, buildings range from one to two stories and distant mountain views are obstructed by buildings and landscaping trees.

The primary viewers of the proposed telecommunications facilities would include local residents, agricultural workers, and employees of existing businesses.

# 4.1.2 Environmental Effects

# 4.1.2.1 Significance Criteria

An impact related to aesthetics was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice.

Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Have a substantial adverse effect on a scenic vista.
- Substantially damage scenic resources including, but not limited to, trees, rock outcroppings, and historic buildings within a state scenic highway.

- Substantially degrade the existing visual character or quality of a site and its surroundings.
- Create a new source of substantial light or glare which would adversely affect day or nighttime views in the area.

#### 4.1.2.2 Impacts and Mitigation Measures

#### Impact AES-1: Adverse Impact on a Scenic Vista (Less than Significant)

The lack of topographic relief in the project area and presence of large areas dominated by agriculture allows mostly unobstructed views of distant mountains, which are considered a scenic visual resource in Imperial County. Construction of the proposed project would result in temporary impacts to the visual resources of the project area. These short-term impacts would be due to the presence of equipment and work crews during the installations. The equipment used would be similar in character to the agricultural equipment that is currently used in the fields adjacent to the project corridors. Following construction, aboveground facilities, including 10 new equipment cabinets and several splice pedestals painted in neutral colors, would be visible along the roads in the project area. These new facilities would be in character with the existing utility cabinets and pedestals found along the roads. These impacts to scenic vistas would be less than significant.

# Impact AES-2: Substantially Damage Scenic Resources including, but not limited to, Trees, Rock Outcroppings, and Historic Buildings within a State Scenic Highway (No Impact)

There are no State-designated scenic highways in the project area (CDOT 2014), and the project would not require removal of trees, rock outcroppings, historic buildings or other scenic resources; therefore, there would be no impact to scenic resources.

# Impact AES-3: Substantially Degrade the Existing Visual Character or Quality of a Site and its Surroundings (Less than Significant)

Construction of the proposed project would result in temporary impacts to the visual resources of the project area. These short-term impacts would be due to the presence of equipment and work crews during the installations. The equipment used would be similar in character to the agricultural equipment that is currently used in the fields adjacent to the project corridors.

Following construction, aboveground facilities, including 10new equipment cabinets and several splice pedestals painted in neutral colors, would be visible along the roads in the project area. These new facilities would be in character with the existing utility cabinets and pedestals found along the roads. These impacts to the visual character of the area would be less than significant.

# Impact AES-4: Create a New Source of Substantial Light or Glare which would Adversely Affect Day or nighttime Views in the Area (No Impact)

The proposed project does not include the installation of new sources of light or glare. Installation would occur during daylight hours and would not require lighting the work area. Therefore, there would be no impacts related to light or glare.

# 4.1.3 References

California Department of Transportation (CDOT)

2014 Officially Designated State Scenic Highways and Parkways. Available at: http://www.dot.ca.gov/hq/LandArch/scenic\_highways/index.htm. Accessed on December 1, 2014.

# 4.2 Agricultural Resources

# 4.2.1 Affected Environment

#### 4.2.1.1 Regulatory Setting

#### Federal

No Federal regulations or policies related to agricultural resources apply to the proposed project.

State

### Farmland Mapping and Monitoring Program

California established the Farmland Mapping and Monitoring Program (FMMP) in 1982 to continue the Important Farmland Inventory efforts begun by the Natural Resources Conservation Service (NRCS) in 1975. The FMMP is a non-regulatory program intended to aid in assessing the location, quality, and quantity of agricultural lands and the conversion of such lands over time. The FMMP provides consistent and impartial data for the analysis of agricultural land uses and land use changes in California. Under the FMMP, the first Important Farmland Maps were produced in 1984, covering 38 of the state's 58 Counties. Current maps, released every 2 years, cover almost 98 percent of the State's privately held land (California Department of Conservation 2014). The FMMP rates agricultural land according to soil quality and irrigation status within the designations discussed below.

**Prime Farmland:** Prime farmland is land that has the best combination of physical and chemical characteristics for producing food, feed, fiber, forage, oilseed, and other agricultural crops with minimum inputs of fuel, fertilizer, pesticides, and labor, and without intolerable soil erosion.

Unique Farmland: Unique farmland is land other than prime farmland that is used for the production of specific, high-value food and fiber crops such as citrus, tree nuts, olives, cranberries, fruits, and vegetables.

**Farmland of Statewide Importance:** Farmland of statewide importance is land of statewide or local importance, but not of national significance, that has been identified by State or local agencies for agricultural use.

Farmland of Local Importance: Farmland of local importance is land identified as important to the local agricultural economy by each County's board of supervisors and a local advisory committee.

#### Williamson Act

The California Land Conservation Act of 1965, commonly referred to as the Williamson Act, is a State policy administered at the local government level. The Williamson Act is intended to preserve agricultural and open-space lands through contracts with private landowners. By entering into a Williamson Act contract, the landowner foregoes the possibility of converting agricultural land to nonagricultural use for a rolling period of 10 years in return for lower property taxes. Local governments receive an annual subvention of foregone property tax revenues from the State via the Open Space Subvention Act of 1971.

Of California's 58 Counties, 53 have adopted the Williamson Act program, including Imperial County. However, beginning in budget year 2008–2009, California drastically reduced subvention reimbursements to Counties as part of a plan to phase out the program. In 2009–2010, California Governor Arnold Schwarzenegger cut State subvention funding to \$1,000, essentially eliminating State support for the program. In response to these funding cuts, Imperial County filed non-renewal on all Williamson Act contracts, effective January 2011 and covering 117,246 acres; however, pursuant to California Government Code (CGC) Section 51246, the contracts remain in full force and effect until their termination dates.

#### Local

# Imperial County General Plan

The Agricultural Element of the ICGP serves as the primary policy statement for implementing development policies for agricultural land use in Imperial County. The goals, objectives, implementation programs, and policies found in the Agricultural Element provide direction for new development as well as government actions and programs. Imperial County's Goals and Objectives are intended to serve as long-term principles and policy statements to guide agricultural use decision-making and uphold the community's ideals.

The County's Agricultural Element identifies several Implementation Programs and Policies for the preservation of agricultural resources. The Agricultural Element recognizes that the County can and should take additional steps to provide further protection for agricultural operations, while at the same time it should provide for the logical, organized growth of urban areas. The County must be specific and consistent about which lands will be maintained for the production of food and fiber and for support of the County's agricultural economy. The County's strategy and overall framework for maintaining agriculture includes the following policy directed at the preservation of Important Farmland:

The overall economy of the County is expected to be dependent upon the agricultural industry for the foreseeable future. As such, all agricultural land in the County is considered Important Farmland, as defined by Federal and State agencies, and should be reserved for agricultural uses. Agricultural land may be converted to nonagricultural uses only where a clear and immediate need can be demonstrated, such as requirements for urban housing, commercial facilities, or employment opportunities. All existing agricultural land will be preserved for irrigation agriculture, livestock production, aquaculture, and other agriculture-related uses except for nonagricultural uses identified in this General Plan or in previously adopted City General Plans.

The following program is provided in the Agricultural Element:

No agricultural land designated except as provided in Exhibit C shall be removed from the Agriculture category except where needed for use by a public agency, for geothermal purposes, where a mapping error may have occurred, or where a clear long term economic benefit to the County can be demonstrated through the planning and environmental review process. The Board (or Planning Commission) shall be required to prepare and make specific findings and circulate same for 60 days (30 days for parcels considered under Exhibit C of this element) before granting final approval of any proposal which removes land from the Agriculture category.

Also, the following policy addresses Development Patterns and Locations on Agricultural Land:

"Leapfrogging" or "checkerboard" patterns of development have intensified recently and result in significant impacts to the efficient and economic production of adjacent agricultural land. It is a policy of the County that leapfrogging will not be allowed in the future. All new nonagricultural development will be confined to areas identified in this plan for such purposes or in Cities' adopted Spheres of Influence, where new development must adjoin existing urban uses. Nonagricultural residential, commercial, or industrial uses will only be permitted if they adjoin at least one side of an existing urban use, and only if they do not significantly impact the ability to economically and conveniently farm adjacent agricultural land.

Agricultural Element Programs that address "leapfrogging" or "checkerboard" development include:

All nonagricultural uses in any land use category shall be analyzed during the subdivision, zoning, and environmental impact review process for their potential impact on the movement of agricultural equipment and products on roads located in the Agriculture category, and for other existing agricultural conditions which might impact the projects, such as noise, dust, or odors.

The Planning and Development Services Department shall review all proposed development projects to assure that any new residential or nonagricultural commercial uses located on agriculturally zoned land, except land designated as a Specific Plan Area, be adjoined on at least one entire property line to an area of existing urban uses. Developments that do not meet these criteria should not be approved.

# 4.2.1.2 Project Setting

According to the ICGP, agriculture has been the most important economic activity in the County throughout its history. The County recognizes the area as one of the finest agricultural areas in the world due to several environmental and cultural factors, including good soils, a year-round growing season, the availability of adequate water transported from the Colorado River, extensive areas committed to agricultural production, a gently sloping topography, and a climate that is well-suited for growing crops and raising livestock (Imperial County 2008).

The proposed project is located in an agricultural area classified as Prime Farmland by the FMMP. The project alignment is located within existing ROW that is used for transportation. Surrounding zoning in the majority of the project area is General Agricultural (see Section 4.9, Land Use). Crops observed in the project area during the biological evaluation survey include Sudangrass, wheat, cotton, alfalfa, dates, citrus, and other commodities.

#### 4.2.2 Environmental Effects

# 4.2.2.1 Significance Criteria

An impact related to agriculture was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice.

Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Convert Prime Farmland, Unique Farmland, or Farmland of Statewide Importance (Farmland), as shown on the maps pursuant to the Farmland Mapping and Monitoring Program of the California Resources Agency, to nonagricultural use.
- Conflict with existing zoning for agricultural use or a Williamson Act contract.
- Conflict with existing zoning for, or cause rezoning of, forest land (as defined in Public Resources Code section 122220[g]), timberland (as defined by Public Resources Code section 4526), or timberland zoned Timberland Production (as defined by Government Code section 51104[g]).
- Result in the loss of forest land or conversion of forest land to non-forest use.
- Result in other changes in the existing environment that, due to their location or nature, could result in the conversion of farmland to nonagricultural use or conversion of forest land to non-forest use.

#### 4.2.2.2 Impacts and Mitigation Measures

Impact AG-1: Conversion of Prime Farmland, Unique Farmland, or Farmland of Statewide Importance (Farmland), as Shown on the Maps Pursuant to the FMMP of the California Resources Agency, to Nonagricultural Use (No Impact)

Impact AG-2: Conflict with Existing Zoning for Agricultural Use or a Williamson Act contract (No Impact)

Impact AG-3: Conflict with Existing Zoning for, or Cause Rezoning of, forest Land, Timberland, or timberland Zoned as Timberland Production (No Impact)

Impact AG-4: Result in the Loss of Forest Land or Conversion of Forest Land to Non-Forest Use (No Impact)

Impact AG-5: Involve Other Changes in the Existing Environment which, Due to their Location or Nature, Could Result in conversion of Farmland to nonagricultural Use or Conversion of forest Land to Non-Forest Use (No Impact)

The proposed project would not result in the conversion of farmland to a nonagricultural use because all of the proposed installations would occur along existing County roads and the agricultural fields located next to the project alignment would be avoided (see APM BIO-3). For the same reason, there would be no conflicts with existing zoning regulations for agricultural areas or Williamson Act contracts. There is no forested land in the project area; therefore, the proposed project would have no effect on either forested land or any zoning regulations designating forested land. There would be no impacts to agricultural resources.

#### 4.2.3 References

California Department of Conservation

2014

California Important Farmland Finder. Available at: http://maps.conservation.ca.gov /ciff/ciff.html. Accessed on August 18, 2014.

#### Imperial County

2008 *Imperial County General Plan.* County of Imperial Planning/Building Department, El Centro, California.

### 4.3 Air Quality and Greenhouse Gases

#### 4.3.1 Affected Environment

#### 4.3.1.1 Regulatory Setting

Air quality and climate change are addressed by the Federal Clean Air Act (CAA) and California Clean Air Act (CCAA) and by local air district planning pursuant to the Acts. At the Federal level, the EPA administers the CAA. In California, the CCAA is administered by the California Air Resources Board (CARB) at the State level and by Air Quality Management Districts at the regional and local levels. The Imperial County Air Pollution Control District (ICAPCD) has local jurisdiction over the proposed project area.

#### Criteria Pollutants

The EPA and CARB have established national ambient air quality standards (NAAQS) and California Ambient Air Quality Standards (CAAQS), respectively, for the following six criteria pollutants: carbon monoxide (CO), nitrogen dioxide (NO<sub>2</sub>), sulfur dioxide (SO<sub>2</sub>), ozone (O<sub>3</sub>), lead (Pb), and particulate matter (PM), including PM less than 10 microns in diameter (PM10) and less than 2.5 microns in diameter (PM 2.5).

The local air districts develop air quality and air pollutant regulations and prepare air quality plans that set goals and measures for achieving attainment with NAAQS and CAAQS. The districts also develop emissions inventories, collect air-monitoring data, and perform dispersion modeling simulations to establish strategies that will reduce emissions and improve air quality. As part of an effort to attain and maintain NAAQS and CAAQS, the ICAPCD has established and adopted thresholds of significance for criteria pollutants of greatest concern within the district (ICAPCD 2007). The thresholds for ozone precursors (reactive organic gas [ROG] and nitrogen oxides [NO<sub>x</sub>]), PM10, and CO emissions from construction activities can be found in Table 4.1.

Pollutant	Threshold		
PM10	150 lbs./day		
ROG	75 lbs./day		
NO <sub>x</sub>	100 lbs./day		
СО	550 lbs./day		

Table 4.1. ICAPCD Significance Thresholds for Emissions from Construction Activities

#### **Greenhouse Gases**

Gases that have the ability to trap heat in the atmosphere are called greenhouse gases (GHGs). The  $CO_2$  produced from the burning of fossil fuels is a GHG, one of five principal GHGs entering the atmosphere due to human activities identified by the EPA and other Federal agencies. The other four gases are methane (CH<sub>4</sub>), nitrous oxide (N<sub>2</sub>O), and fluorinated gases. Since the time of the Industrial Revolution, the concentrations of GHGs in the Earth's atmosphere have risen and have been correlated with rising average temperatures. Increased atmospheric temperature, often called global warming, is only one aspect of climate change; other influences on climate can include human causes, such as deforestation and the development of land, and natural causes, such as changes in ocean and atmospheric circulation, the Earth's orbit, solar intensity, and volcanic activity.

GHGs such as  $CH_4$  and  $N_2O$  have a greater potential to produce global warming effects relative to  $CO_2$ . This phenomenon is known as Global Warming Potential (GWP) and is related to the gases' abilities to absorb energy and also persist in the atmosphere. The GWP of  $CO_2$  is 1, which serves as a baseline for other GWP values;  $CH_4$  has a GWP of 25, and the GWP of  $N_2O$  is 298 (EPA 2014a). The metric measure used to compare the emissions of various GHGs based upon their relative GWP is known as  $CO_2$  Equivalent ( $CO_2$  Eq.), which is customarily expressed in metric tons.

### **Fugitive Dust**

In Imperial County, all construction activities must be in compliance with Regulation VIII (ICAPCD 2007). The main purpose of this regulation is to reduce the amount of PM10 released into the atmosphere as a result of manmade fugitive dust sources. Compliance with the regulation does not constitute mitigation and it is presumed that all projects occurring in Imperial County will be implemented in compliance with Regulation VIII. Standard measures for fugitive PM10 control outlined in Regulation VIII include:

- All disturbed areas, including bulk material storage that is not being actively utilized, shall be effectively stabilized and visible emissions shall be limited to no greater than 20 percent opacity for dust emissions by using water, chemical stabilizers, dust suppressants, tarps, or other suitable material such as vegetative ground cover.
- All on- and off-site unpaved roads will be effectively stabilized, and visible emissions shall be limited to no greater than 20 percent opacity for dust emissions by paving, chemical stabilizers, dust suppressants, and/or watering.
- All unpaved traffic areas 1 acre or more in size with 75 or more average vehicle trips per day will be effectively stabilized, and visible emissions shall be limited to no greater than 20 percent opacity for dust emissions by paving, chemical stabilizers, dust suppressants, and/or watering.
- The transport of bulk materials shall be completely covered, unless 15 cm (6 inches) of freeboard space from the top of the container is maintained with no spillage or loss of bulk material. In addition, the cargo compartment of all haul trucks is to be cleaned and/or washed at the delivery site after removal of bulk material.
- All track-out and carry-out will be cleaned at the end of each workday or immediately when mud or dirt extends a cumulative distance of 15 linear m (50 linear feet) or more onto a paved road within an urban area.
- Bulk material shall be stabilized prior to movement or at points of transfer with the application of sufficient water, the application of chemical stabilizers, or by sheltering or enclosing the operation and transfer line.

• The construction of any new unpaved road is prohibited within any area with a population of 500 or more unless the road meets the definition of a temporary unpaved road. Any temporary unpaved road shall be effectively stabilized, and visible emissions shall be limited to no greater than 20 percent opacity for dust emissions by paving, chemical stabilizers, dust suppressants, and/or watering.

In order to provide a greater degree of PM10 reductions, above that required by Regulation VIII, the ICAPCD recommends the following discretionary mitigation measures for fugitive PM10 control:

- Watering of exposed soil with adequate frequency for continued moist soil.
- Replacing ground cover in disturbed areas as quickly as possible.
- Installing an automatic sprinkler system on all soil piles.
- Vehicle speed for all construction vehicles shall not exceed 15 mph on any unpaved surface at the construction site.
- Develop a trip reduction plan to achieve a 1.5 average vehicle ridership (AVR) for construction employees.
- Implement a shuttle service to and from retail services and food establishments during lunch hours.

# 4.3.1.2 Project Setting

The Western Regional Climate Center (WRCC) recorded seasonal climatic data from 1993–2013 at the Yuma Quartermaster Depot, located just south of the project area (WRCC 2014). These data include average maximum temperature, average minimum temperature, average total precipitation, and average snowfall. The average annual maximum temperature within the project area is 90.1° F (32.2° C), with the hottest month of the year being July with an average maximum temperature of 109.4° F (43.0° C). The average annual minimum temperature within the project area is 59.0° F (15.0° C), with December having the coldest average temperature of 43.4° F (6.3° C). The project area receives an average of 6.80 cm (2.67 inches) of precipitation annually, with February having the highest average precipitation at 1.20 cm (0.48 inches). The project area receives no snowfall in the average year.

The proposed project area is located within the Salton Sea air basin. Review of the 2013 CAAQS criteria pollutant attainment status for the basin indicates that it was in attainment for PM 2.5, CO, NO<sub>2</sub>, SO<sub>2</sub>, SO<sub>x</sub>, and PB and non-attainment for PM10 and O<sub>3</sub>. Review of the NAAQS criteria pollutant attainment status for the same year indicates that the air basin was in non-attainment for 8h O<sub>3</sub>, attainment for SO<sub>2</sub>, and unclassified/attainment for PM 2.5, CO, Pb, and NO<sub>2</sub> (CARB 2014).

# 4.3.2 Environmental Effects

# 4.3.2.1 Significance Criteria

An impact related to air quality or GHGs was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice.

Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Conflict with or obstruct implementation of the applicable air quality plan.
- Violate any air quality standard or contribute substantially to an existing or projected air quality violation.
- Result in a cumulatively considerable net increase of any criteria pollutant for which the project region is in non-attainment under an applicable Federal or state ambient air quality standard (including releasing emissions that exceed quantitative thresholds for ozone precursors).
- Expose sensitive receptors to substantial pollutant concentrations.
- Create objectionable odors affecting a substantial number of people.
- Generate GHG emissions, either directly or indirectly, that may have a significant impact on the environment.
- Conflict with an applicable plan, policy, or regulation adopted for the purpose of reducing the emissions of GHGs.

# 4.3.2.2 Impacts and Mitigation Measures

The proposed project would incorporate ICAPCD required dust control measures as detailed in the APM below and would not result in significant impacts on air quality in the project area.

**APM AQ-1:** TDS will require all construction contractors to implement the following ICAPCD standard measures for fugitive PM10 control:

- All disturbed areas, including bulk material storage that is not being actively utilized, shall be effectively stabilized, and visible emissions shall be limited to no greater than 20 percent opacity for dust emissions by using water, chemical stabilizers, dust suppressants, tarps, or other suitable material, such as vegetative ground cover.
- All on- and off-site unpaved roads will be effectively stabilized, and visible emissions shall be limited to no greater than 20 percent opacity for dust emissions by paving, chemical stabilizers, dust suppressants, and/or watering.
- All unpaved traffic areas 1 acre or more in size with 75 or more average vehicle trips per day will be effectively stabilized, and visible emissions shall be limited to no greater than 20 percent opacity for dust emissions by paving, chemical stabilizers, dust suppressants, and/or watering.
- The transport of bulk materials shall be completely covered unless 15 cm (6 inches) of freeboard space from the top of the container is maintained with no spillage or loss of bulk material. In addition, the cargo compartment of all haul trucks is to be cleaned and/or washed at the delivery site after removal of bulk material.
- All track-out and carry-out will be cleaned at the end of each workday or immediately when mud or dirt extends a cumulative distance of 15 linear m (50 linear feet) or more onto a paved road within an urban area.
- Bulk material shall be stabilized prior to movement or at points of transfer with the application of sufficient water, the application of chemical stabilizers, or by sheltering or enclosing the operation and transfer line.
- The construction of any new unpaved road is prohibited within any area with a population of 500 or more unless the road meets the definition of a temporary unpaved road. Any temporary unpaved road shall be effectively stabilized, and visible

emissions shall be limited to no greater than 20 percent opacity for dust emissions by paving, chemical stabilizers, dust suppressants, and/or watering.

In addition, the following ICAPCD-recommended discretionary measures will be implemented:

- Watering of exposed soil with adequate frequency for continued moist soil.
- Replacing ground cover in disturbed areas as quickly as possible.
- Installing an automatic sprinkler system on all soil piles.
- Vehicle speed for all construction vehicles shall not exceed 15 mph on any unpaved surface at the construction site.

# Impact AQ-1: Conflict with or Obstruct Implementation of the Applicable Air Quality Plan (Less than Significant).

The proposed project area is located in the Salton Sea air basin, which is currently in non-attainment for PM10 and  $O_3$  (CAAQs) and for 8h  $O_3$  (NAAQS). The ICAPCD adopted an Air Quality Management Plan for  $O_3$  on July 13, 2010, and a State implementation plan for PM10 on August 11, 2009. The ICAPCD plans estimate future emissions and describe strategies necessary for emissions reductions through regulatory controls. Emissions projections in the plans are based on population, vehicle, and land-use trends developed by the ICAPCD and CARB.

A proposed project would be considered inconsistent with air quality plans if it would result in population and/or employment growth that exceeds estimates used to develop applicable air quality plans. Projects that propose development that is consistent with the growth anticipated by the relevant land use plans would be consistent with the current ICAPCD air quality plans. Similarly, projects that propose development that is less dense than anticipated within a General Plan or other applicable land use plan would be consistent with the air quality plans because emissions would be less than estimated for the region.

The purpose of the proposed project is to make affordable broadband Internet services available to currently underserved areas in Imperial County, including a portion of the Fort Yuma–Quechan Reservation. It would not induce population or employment growth and would not conflict or obstruct the implementation of the applicable air quality plans. The proposed project would generate minor amounts of emissions during construction; however, no emissions would be generated during operation, and the emissions generated are not anticipated to impede attainment or maintenance of the NAAQS or CAAQS by the ICAPCD. Therefore, this impact would be considered less than significant.

# Impact AQ-2: Violate any air quality standard or contribute substantially to an existing or Projected Air Quality Violation (Less than Significant).

Potential impacts from the proposed project on the air quality of the project area were modeled using the California Emissions Estimator Model (CalEEMod) version 2013.2.2 (Appendix B). Construction equipment indicated in Section 3.6.5 operated under the schedule in Table 4.2 below were used as inputs for the model, which provided estimates for the ICAPCD criteria pollutants as well as an estimate for the amount of GHG that would be released during construction of the proposed project.

Table 4.2. Modeled Construction Schedule

Construction Phase	Days of Construction
Plowed Conduit Installation	7
Bored Conduit Installation	32
Node Installation	5
Total	44

Reactive Organic Gas (ROG), NOx, CO, and PM10 and PM2.5 (exhaust) estimates for all construction phases include unmitigated on- and off-site emissions (Table 4.3). PM10 and PM2.5 estimates only include dust from equipment exhaust because all on-site fugitive dust will be controlled through the implementation of standard measures in compliance with Imperial County Regulation VIII (APM AQ-1).

	Criteria Pollutant Emissions (lbs./day)						
Construction Phase	ROG	NO <sub>x</sub>	СО	PM10		PM2.5	
	On+Off-Site			Dust <sup>a</sup>	Exhaust	Dust <sup>a</sup>	Exhaust
Plowed Conduit	1.90+0.69	22.24+0.43	8.20+0.84	(0.22	1.00+0.01	6.01	0.94 + 0.01
Installation	2.59	22.67	9.04	60.22	1.01	0.01	0.95
Bored Conduit	3.21+0.05	29.98+0.33	19.16+0.67	50.00	$1.78 \pm 0.01$	5.09	$1.72 \pm 0.01$
Installation	3.26	30.31	19.83	50.90	1.79	5.08	1.73
Node	0.34+0.05	3.24+0.33	2.40+0.67	50.00	0.25+0.01	5.09	$0.23 \pm 0.01$
Installation	0.39	3.57	3.07	50.90	0.26	5.08	0.24
Maximum Daily	3.26	30.31	10.93	60.22	1.70	6.01	1 72
Emission	5.20	30.31	19.63	00.22	1.79	0.01	1.75
ICAPCD	75	100	500		150	none	
Thresholds	75	100	500		150		
Exceeds	20						
Threshold?	110	110	110		110	1	1/a

Table 4.3. Estimated Daily Construction Emissions—Criteria Pollutants

<sup>a</sup> Off-site fugitive dust only; all on-site fugitive dust will be controlled per Regulation VIII.

As shown in Table 4.3, the proposed project's estimated emissions would be below the ICAPCD maximum daily emission thresholds for all criteria pollutants. On-site fugitive dust will be controlled through the implementation of standard measures in compliance with Imperial County Regulation VIII (APM AQ-1). Therefore, the criteria pollutant emissions impacts associated with the proposed project would be less than significant.

# Impact AQ-3: Result in a Cumulatively Considerable Net Increase of Any Criteria Pollutant for which the Project Region is in Non-Attainment under an Applicable Federal or State Ambient Air Quality Standard (including Releasing Emissions which Exceed Quantitative Thresholds for Ozone Precursors) (Less than Significant).

The project area is currently in non-attainment for the criteria pollutants PM10 and O<sub>3</sub>; however, the estimated emissions levels from the proposed project during construction for both PM10 and ROG

are both well below the ICAPCD thresholds. Consequently, because the proposed project's anticipated emissions of these two criteria pollutants that are in non-attainment are below what ICAPCD would consider significant, any cumulative impacts would be considered less than significant.

# Impact AQ-4: Expose Sensitive Receptors to Substantial Pollutant Concentrations (Less than Significant).

Sensitive receptors located along the project corridors include residences and schools. Equipment used for the proposed installations would release diesel exhaust as the installations proceed; however, this equipment would not remain in any one location for a prolonged period of time. Therefore, substantial pollutant concentrations would not occur in the vicinity of the sensitive receptors along the project corridors, and impacts would be less than significant.

# Impact AQ-5: Create Objectionable Odors Affecting a Substantial Number of People (Less than Significant).

None of the facilities to be installed during construction of the proposed project are known to have odor impacts; however, equipment used for the proposed installations would release diesel exhaust, which some people may consider to have an objectionable odor, as the installations proceed. Because the proposed project area is primarily located in an open, rural area with relatively few people, and the construction equipment would not remain in any one location for a long period of time, odor impacts would be less than significant.

# Impact AQ-6: Generate GHG Emissions, Either Directly or Indirectly, that May Have a Significant Impact on the Environment (Less than Significant).

The proposed project's GHG emissions in  $CO_2$  Eq. were estimated using CalEEMod in lbs/day and extrapolated for the entire duration of each construction phase in metric tons (Table 4.4). No GHG emissions would be released during operation of the telecommunications system; therefore, the only emissions of GHG that require consideration are those from construction. The 68.4 MT of  $CO_2$  Eq. emissions that would be released by the proposed project is the same amount released by 14.4 average passenger vehicles in a year (EPA 2014b), which, given the 23.8 million registered passenger vehicles in California in 2014 (CDMV 2015), would be in comparison less than significant.

Construction Phase	CO <sub>2</sub> Equivalent (Lbs./Day), On+Off-Site	CO <sub>2</sub> Equivalent (Metric Tons)
Plowed Conduit Installation	1,987+115, <b>2,102</b>	6.7
Bored Conduit Installation	4,103+90, <b>4,193</b>	60.8
Node Installation	324+90, <b>414</b>	0.9
Project Total		68.4

#### Table 4.4. Estimated Greenhouse Gas Emissions

Impact AQ-7: Conflict with an Applicable Plan, Policy, or Regulation Adopted for the Purpose of Reducing the Emissions of GHGs (No Impact).

The ICAPCD currently has no adopted plan for reducing the emissions of GHGs. There would be no impacts related to emissions of GHGs.

### 4.3.3 References

California Air Resources Board (CARB)

2014 California Environmental Protection Agency Air Resources Board website. Available at: http://www.arb.ca.gov/homepage.htm. Accessed on August 18, 2014.

California Department of Motor Vehicles (CDMV)

2015 State of California Department of Motor Vehicles Statistics for Publication, January through December 2014. Available at: https://www.dmv.ca.gov/portal/wcm/connect/5aa16cd3-39a5-402f-9453-0d353706cc9a/official.pdf?MOD=AJPERES. Accessed on March 31, 2015.

Imperial County Air Pollution Control District (ICAPCD)

2007 CEQA Air Quality Handbook, Guidelines for the Implementation of the California Environmental Quality Act as 1970, as amended. ICAPCD, El Centro, California.

U.S. Environmental Protection Agency (EPA)

2014a EPA Climate Change website. Available at: http://epa.gov/climatechange/. Accessed on January 12, 2015.

2014b EPA Greenhouse Gas Equivalencies Calculator website. Available at: http://www.epa.gov/cleanenergy/energy-resources/calculator.html. Accessed on January 16, 2014.

Western Regional Climate Center (WRCC)

2014 Yuma Quartermaster Depot, Arizona—Climate Summary. Available at: http://www.wrcc. dri.edu/cgi-bin/cliMAIN.pl?nm8668. Accessed on June 19, 2014.

# 4.4 Biological Resources

# 4.4.1 Affected Environment

# 4.4.1.1 Regulatory Setting

Federal

# Endangered Species Act

The U.S. Fish and Wildlife Service (FWS) and the National Oceanographic and Atmospheric Administration's National Marine Fisheries Service (NMFS) enforce the provisions stipulated within the ESA of 1973 (16 USC Section 1531 et seq.). Threatened and Endangered species on the Federal list (50 CFR Section 17.11, and 17.12) are protected from take, defined as direct or indirect harm, unless a Section 10 permit is granted to an entity other than a Federal agency or a Biological

Opinion with incidental-take provisions is rendered to a Federal lead agency via a Section 7 consultation. Pursuant to the requirements of the ESA, an agency reviewing a proposed project within its jurisdiction must determine whether any Federally listed species may be present in the project site and determine whether the proposed project will have a potentially significant impact upon such species. Under the ESA, habitat loss is considered to be an impact to a species. In addition, the agency is required to determine whether the project is likely to jeopardize the continued existence of any species that is proposed for listing under the ESA or to result in the destruction or adverse modification of critical habitat proposed or designated for such species (16 USC 1536[3], [4]). Therefore, project-related impacts to these species or their habitats would be considered significant and would require mitigation.

#### Executive Order 13186: Migratory Bird Treaty Act

The Migratory Bird Treaty Act (MBTA) of 1918 (United States Code, Title 16, Chapter 7, Subchapter II) prohibits the "pursuit, hunt, take, capture, kill, attempt to take, capture, or kill, possess, offer for sale, sell, offer to barter, barter, offer to purchase, purchase, deliver for shipment, ship, export, import, cause to be shipped, exported, or imported, deliver for transportation, transport or cause to be transported, carry or cause to be carried, or receive for shipment, transportation, carriage, or export, any migratory bird, any part, nest, or eggs of any such bird, or any product, whether or not manufactured, which consists, or is composed in whole or part, of any such bird or any part, nest, or egg thereof." The ensuing Executive Order 13186, signed January 10, 2001, by President Clinton "directs executive departments and agencies to take certain actions to further implement the (MBTA)." Such actions include the responsibility that Federal agencies "taking actions that have, or are likely to have, a measurable negative effect on migratory bird populations ... develop and implement, within 2 years, a Memorandum of Understanding with the Fish and Wildlife Service, that shall promote the conservation of migratory bird populations."

#### Executive Order 11990: Protection of Wetlands

Executive Order 11990, signed May 24, 1997, directs Federal agencies to refrain from assisting in or giving financial support to projects that encroach on publicly or privately owned wetlands. It further requires that Federal agencies support a policy to minimize the destruction, loss, or degradation of wetlands. A project that encroaches on wetlands may not be undertaken unless the agency has determined that (1) there are no practicable alternatives to construction, (2) the project includes all practicable measures to minimize harm to wetlands affected, and (3) the impact will be minor.

#### Executive Order 13112: Invasive Species Prevention

On Feb 3, 1999, Executive Order 13112 was signed, establishing the National Invasive Species Council. Executive Order 13112 required that each Federal agency whose actions may affect the status of invasive species will, to the extent practicable and permitted by law, (1) identify such actions; (2) subject to the availability of appropriations, and within Administration budgetary limits, use relevant programs and authorities to: (i) prevent the introduction of invasive species, (ii) detect and respond rapidly to and control populations of such species in a cost-effective and environmentally sound manner, (iii) monitor invasive species populations accurately and reliably, (iv) provide for the restoration of native species and habitat conditions in ecosystems that have been invaded, (v) conduct research on invasive species and develop technologies to prevent introduction and provide for environmentally sound control of invasive species, and (vi) promote public education on invasive species and the means to address them; and (3) not authorize, fund, or carry out actions that it believes are likely to cause or promote the introduction or spread of invasive species in the United States or elsewhere unless, pursuant to guidelines that it has prescribed, the
agency has determined and made public its determination that the benefits of such actions clearly outweigh the potential harm caused by invasive species; and that all feasible and prudent measures to minimize risk of harm will be taken in conjunction with the actions. In addition, it requires that Federal agencies will pursue the duties set forth in this section in consultation with the Invasive Species Council, consistent with the Invasive Species Management Plan and in cooperation with stakeholders, as appropriate, and, as approved by the Department of State, when Federal agencies are working with international organizations and foreign nations.

#### State

#### California Endangered Species Act/California Environmental Quality Act

The California Endangered Species Act (CESA) of 1970 (Fish and Game Code Section 2050 et seq., and CCR Title 14, Subsection 670.2, 670.51) prohibits the take (interpreted to mean the direct killing of a species) of species listed under CESA (14 CCR Subsection 670.2, 670.5). Under CESA, State agencies are required to consult with the California Department of Fish and Wildlife (CDFW, formerly California Department of Fish and Game [CDFG]) when preparing CEQA documents. Consultation ensures that proposed projects or actions do not have a negative effect on State-listed species. During consultation, CDFW determines whether take would occur and identifies "reasonable and prudent alternatives" for the project and the conservation of special status species. CDFW can authorize take of a State-listed species under Sections 2080.1 and 2081(b) of CDFW code in those cases where it is demonstrated that the impacts are minimized and mitigated. Take authorized under Section 2081(b) must be minimized and fully mitigated. A CESA permit must be obtained if a project will result in take of listed species either during construction or over the life of the project. Under CESA, CDFW is responsible for maintaining a list of Threatened and Endangered species designated under State law (CDFG Code 2070). CDFW also maintains lists of Species of Special Concern, which serve as "watch lists." Pursuant to the requirements of CESA, a State or local agency reviewing a proposed project within its jurisdiction must determine whether any State-listed species may be present in the project area and determine whether the proposed project will have a potentially significant impact upon such species. Project-related impacts to species on the CESA list would be considered significant and would require mitigation. Impacts to species of concern and fully protected species would be considered significant under certain circumstances.

CEQA (Subsections 21000-21178) requires that CDFW be consulted during the CEQA review process regarding impacts of proposed projects on Rare or Endangered species. These "special status" species are defined under CEQA Guidelines, Subsection 15380(b) and (d), as those listed under the ESA and CESA, and species that are not currently protected by statute or regulation but would be considered Rare, Threatened, or Endangered under these criteria, or by the scientific community. Therefore, species that are considered Rare or Endangered are addressed in this study regardless of whether they are afforded protection through any other statute or regulation. The California Native Plant Society (CNPS) inventories the native flora of California and ranks species according to rarity; plants on Lists 1A, 1B, and 2 are considered special status species under CEQA.

Although Threatened and Endangered species are protected by specific Federal and State statutes, CEQA Guidelines Section 15380(d) provides that a species not listed on the Federal or State list of protected species may be considered Rare or Endangered if it can be shown to meet certain specified criteria. These criteria have been modeled after the definition in the ESA and the section of the California Fish and Game Code dealing with Rare or Endangered plants and animals. Section

15380(d) allows a public agency to undertake a review to determine if a significant effect on species that have not yet been listed by either the FWS or CDFW (i.e., Candidate species) would occur. Thus, CEQA provides an agency with the ability to protect a species from the potential impacts of a project until the respective government agency has an opportunity to designate the species as protected, if warranted.

#### California Native Plant Protection Act

The California Native Plant Protection Act of 1977 (CDFG Code Section 1900-1913) requires all State agencies to use their authority to carry out programs to conserve Endangered and otherwise rare species of native plants. Provisions of the Act prohibit the taking of listed plants from the wild and require the project proponent to notify CDFW at least 10 days in advance of any change in land use, which allows CDFW to salvage listed plants that would otherwise be destroyed.

#### California Department of Food and Agriculture (CDFA) Noxious Weed Species List and the California Invasive Plant Council (CIPC) Invasive Plant Inventory list

The CDFA classifies noxious weeds as to the extent of their distribution in the state and the possibility of successful eradication. "A"-rated noxious weeds are prohibited from entry into the state, sale within the state, and are subject to eradication. "B"-rated noxious weeds are prohibited from nurseries and sale by nurseries and can be prohibited and eradicated at the County level at the discretion of the County Agricultural Commissioner. "C"-rated noxious weeds can also be prohibited from sale and eradicated at the discretion of the County Agricultural Commissioner. "Q"-rated noxious weeds are those weeds that are prohibited until more information as to their invasiveness can be determined.

The CIPC has a rating system for invasive species, as follows:

**High**—These species have severe ecological impacts on physical processes, plant and animal communities, and vegetation structure. Their reproductive biology and other attributes are conducive to moderate-to-high rates of dispersal and establishment. Most are widely distributed ecologically.

**Moderate**—These species have substantial and apparent, but generally not severe, ecological impacts on physical processes, plant and animal communities, and vegetation structure. Their reproductive biology and other attributes are conducive to moderate-to-high rates of dispersal, though establishment is generally dependent upon ecological disturbance. Ecological amplitude and distribution may range from limited to widespread.

Limited—These species are invasive, but their ecological impacts are minor on a statewide level or there was not enough information to justify a higher score. Their reproductive biology and other attributes result in low-to-moderate rates of invasiveness. Ecological amplitude and distribution are generally limited, but these species may be locally persistent and problematic.

#### Nesting Birds

California Fish and Game Code Subsections 3503, 3503.5, and 3800 prohibit the possession, incidental take, or needless destruction of birds, their nests, and eggs. California Fish and Game

Code Section 3511 lists birds that are "Fully Protected" as those that may not be taken or possessed except under specific permit.

#### Protection of Wetlands, Waters of the United States, and Waters of the State

Any person, firm, or agency planning to alter or work in "Waters of the U.S. (WUS)," including the discharge of dredged or fill material, must first obtain authorization from the U.S. Army Corps of Engineers (USACE) under Section 404 of the Clean Water Act (CWA; 33 U.S.C. 1344). Permits, licenses, variances, or similar authorization may also be required by other Federal, State, and local statutes. Section 10 of the Rivers and Harbors Act of 1899 prohibits the obstruction or alteration of navigable WUS without a permit from USACE (33 U.S.C. 403). The CDFW requires notification prior to commencement and possibly a Streambed Alteration Agreement pursuant to California Fish and Game Code Subsection 1601-1603, 5650F, if a proposed project would result in the alteration or degradation of a stream, river, or lake in California. The Regional Water Quality Control Board (RWQCB) may require State Water Quality Certification (CWA Section 401 permit) prior to the alteration of or discharge to WUS and the State.

WUS are defined as all waters that are currently used, or were used in the past, or may be susceptible to use in interstate or foreign commerce, including all waters that are subject to the ebb and flow of the tide; all interstate waters including interstate wetlands; all other waters such as intrastate lakes, rivers, streams (including intermittent and ephemeral streams), mudflats, sand flats, wetlands, sloughs, prairie potholes, wet meadows, playa lakes or natural ponds, where the use, degradation, or destruction of which could affect interstate commerce; impoundments of these waters; tributaries of these waters; or wetlands adjacent to these waters (33 CFR Part 328). With nontidal waters, in the absence of adjacent wetlands, the extent of USACE jurisdiction extends to the ordinary high water mark (OHWM)—the line on the shore established by fluctuations of water and indicated by a clear, natural line impressed on the bank, shelving, changes in soil character, destruction of terrestrial vegetation, and/or the presence of litter and debris. Waters of the State are defined as "any surface water or groundwater, including saline waters, within the boundaries of the state (California Water Code Section 13050(e)."

Water quality in California is governed by the Porter-Cologne Water Quality Control Act (Porter-Cologne Act) (California Water Code § 13000 et. seq.) This act delegates responsibility to the State Water Resource Control Board (SWRCB) for water rights and water quality protection and directs the nine statewide RWQCBs to develop and enforce water quality standards within their jurisdiction. The Porter-Cologne Act requires any entity discharging waste or proposing to discharge waste within any region that could affect the quality of the "Waters of the State" to file a "report of waste discharge" with the appropriate RWQCB. The appropriate RWQCB then must issue a permit, referred to as a waste discharge requirement (WDR). WDRs implement water quality control plans and take into consideration the beneficial uses to be protected, the water quality objectives reasonably required for that purpose, other waste discharges, and the need to prevent nuisances (California Water Code Section 13263).

#### Local

#### Lower Colorado River Multi-Species Conservation Program

The Lower Colorado River Multi-Species Conservation Program (LCR MSCP) was created to balance the use of the Colorado River water resources with the conservation of native species and their habitats. The program works toward the recovery of species currently listed under the ESA. It

also reduces the likelihood of additional species listings. Implemented over a 50-year period, the program accommodates current water diversions and power production and will optimize opportunities for future water and power development by providing ESA compliance through the implementation of a Habitat Conservation Plan (HCP) that was finalized in December 2004.

The program area extends over 400 miles of the lower Colorado River from Lake Mead to the southernmost border with Mexico and includes Lakes Mead, Mohave, and Havasu, as well as the historic 100-year floodplain where the proposed project is located, along the main stem of the lower Colorado River. The HCP calls for the creation of over 3,278 ha (8,100 acres) of habitat for fish and wildlife species and the production of over 1.2 million native fish to augment existing populations. The plan will benefit at least 26 species, most of which are State- or Federally listed Endangered, Threatened, or Sensitive species.

The BOR is the implementing agency for the LCR MSCP. Partnership involvement occurs primarily through the LCR MSCP Steering Committee (currently representing 57 entities including State and Federal agencies, water and power users, municipalities, Native American Tribes, conservation organizations, and other interested parties), which provides input and oversight functions in support of LCR MSCP implementation. Program costs are evenly divided between the Federal government and non-Federal partners.

# Imperial County General Plan

The ICGP, which applies to all public and private projects in unincorporated Imperial County, consists of 10 Elements: Land Use, Housing, Circulation and Scenic Highways, Noise, Seismic and Public Safety, Agricultural, Conservation and Open Space, Geothermal/Alternative Energy and Transmission, Water, and Parks & Recreation (Imperial County 2008).

The Conservation Element and Open Space Element of the ICGP provide detailed plans and measures for the preservation and management of biological and cultural resources, soils, minerals, energy, regional aesthetics, air quality, and open space. The purpose of the Conservation and Open Space Element is to promote the protection, maintenance, and use of the County's natural resources, with particular emphasis on scarce resources, and to prevent wasteful exploitation, destruction, and neglect of the State's natural resources. Additionally, the purpose of this Element is to recognize that natural resources must be maintained for their ecological value for the direct benefit to the public, open space for the preservation of natural resources, the managed production of resources, outdoor recreation, and public health and safety.

# 4.4.1.2 Methodology

#### Field Reconnaissance and Pre-Field Literature Search

Tierra's senior biologist, Tim Jordan, conducted a reconnaissance survey of the project area on July 15 and 16, 2014. Special status species (listed in Appendix A of the Biological Resources Evaluation [BRE] attached as Appendix C to this PEA) were assessed for their potential to occur in the project area based on the existing characteristics that were observed. In addition to special status species and their habitats, the project corridors were assessed for general wildlife species, migratory birds, plant species and noxious weeds, sensitive natural communities, and the presence or absence of waterways. The entire area assessed during the reconnaissance survey included the project corridor centerlines with an approximately 15.2-m (50-foot) buffer to either side, which is

comprehensively referred to as the "study area." All areas within the study area were visually assessed during the surveys.

Prior to conducting the reconnaissance surveys, a comprehensive list of regionally occurring special status species and sensitive natural communities was compiled from the list of reported occurrences in the CDFW's California Natural Diversity Database (CNDDB) for the Araz, Bard, Imperial Reservoir, Laguna Dam, Little Picacho Peak, Picacho Peak, Yuma East, and Yuma West 7.5 minute USGS topographic quadrangles (CNDDB 2014) and from a list of Natural Resources of Concern including Federally listed special status species for Imperial County that was obtained from the FWS Information, Planning, and Conservation (IPAC) system

#### Waterway Delineation

A field delineation was conducted to map all waterways and any vegetation associated with the waterways to be crossed in the project area and to assist TDS with identifying waterways to avoid (Appendix D).

# 4.4.1.3 Project Setting

The project area is located in southeastern California on the lower Colorado River in an area primarily used for agricultural cultivation. Several irrigation canals operated by the BOR's Imperial Irrigation District (IID) and Bard Water District (BWD) either cross or run parallel to the project corridors. Elevations in the project area range from approximately 38–43 m (126–140 feet) above mean sea level (AMSL).

# Terrestrial Habitat

While the study area is located within the Colorado Desert, as classified in *A Manual of California Vegetation* (Sawyer 2009), the dominant type of terrestrial habitat present in the project area consists of agricultural land that is being actively cultivated to produce Sudangrass, wheat, cotton, alfalfa, dates, citrus, and other crops. The road shoulders where the proposed telecommunications line is to be installed are mostly devoid of vegetation due to blading activities associated with road maintenance and agricultural activities. Due to this previous disturbance, little to no native vegetation remains in the project area. Complete lists of plants and wildlife species identified in the study area at the time of the surveys can be found in Appendices C and D of the BRE (Appendix C).

# Aquatic Habitat

Aquatic habitat in the study area is limited to that associated with agricultural canals. There are no ponds or ephemeral or perennial waterways within the study area. Grass Carp *(Ctenopharyngodon idella)*, a fish species native to southeastern Russia and northwestern China, has been stocked in the Yuma Main Canal by the Yuma County Water User's Association (YCWUA) since October 2013 for vegetation control purposes.

#### Sensitive Natural Communities

# Riparian Areas

No sensitive natural communities, as defined by CDFW, are present in the study area. However, the margins of unlined canals in the study area, especially the Reservation Main Drain, contain limited riparian vegetation consisting mostly of dense Common Reed (*Phragmites australis*) and invasive

species such as Salt Cedar (Tamarix ramosissima). This vegetation is mostly low-growing, not structurally complex, and does not have a tree overstory.

#### Wetlands

Riverine wetlands may be present along the unlined canals that are crossed by the project corridors. These potential wetlands were not delineated during the field surveys because TDS would be boring beneath all of the canals crossed by the line installations will employ sufficient set-backs from either the canal edges or the extent of associated vegetation, if present, to avoid any potential impacts to wetlands (see Waterway Delineation and Assessment Report in Appendix D).

#### **Special Status Species**

Based on the assessment methodology outlined above, seven special status wildlife species are either known to occur or have the potential to occur in the study area (Table 4.5). Because of the previously disturbed nature of the study area and its lack of native vegetation, no special status plant species were expected to be found during the surveys, and none were identified.

Scientific Name	Common Name	Status (FWS/State/CNPS)
Amphibians		
Incilius alvarius	Sonoran Desert Toad	-/SSC/-
Lithobates yavapaiensis	Lowland Leopard Frog	-/SSC/-
Birds		
Lanius ludovicianus	Loggerhead Shrike	-/SSC/-
Pyrocephalus rubinus	Vermilion Flycatcher	-/SSC/-
Xanthocephalus xanthocephalus	Yellow-headed Blackbird	-/SSC/-
Mammals		
Corynorhinus townsendii	Townsend's Big-eared Bat	-/CT, SSC/-
Sigmodon hispidus eremicus	Yuma Hispid Cotton Rat	-/SSC/-

Table 4.5. Special Status Species with the Potential to Occur in the Study Area

*Key:* SSC = Species of Special Concern, C = Candidate, T = Threatened.

#### **Migratory Birds**

The study area and/or areas adjacent to it were determined to contain suitable habitat for two migratory birds appearing on the American Bird Conservancy's U.S. Watchlist of Birds of Conservation Concern. No bird nests were observed in the project corridors at the time of the surveys; this lack of nests was due to the project corridors being essentially devoid of vegetation large enough to support bird nests. However, areas adjacent to the project corridors and the study area contain trees and other vegetation that may be utilized by migratory birds. A list of bird species appearing on the 2008 FWS Birds of Conservation Concern (BCC) list for Bird Conservation Region (BCR) 33, Sonoran and Mojave Deserts U.S. Portion Only, can be found in Table 4.6.

Least Bittern	Elf Owl
Bald Eagle	Burrowing Owl
Peregrine Falcon	Costa's Hummingbird
Prairie Falcon	Gila Woodpecker
Black Rail	Gilded Flicker
Snowy Plover	Bell's Vireo
Mountain Plover	Gray Vireo
Whimbrel	Bendire's Thrasher
Long-billed Curlew	LeConte's Thrasher
Marbled Godwit	Lucy's Warbler
Red Knot	Yellow Warbler
Gull-billed Tern	Rufous-winged Sparrow
Black Skimmer	Black-chinned Sparrow
Yellow-billed Cuckoo	Lawrence's Goldfinch

Table 4.6. Bird Conservation Region 33 Migratory Bird List

#### **Invasive Species**

Three invasive plant species appearing on the CDFA Noxious Weed Species List and/or the CIPC Invasive Plant Inventory list were identified in the study area. These invasive species are Russian Thistle (*Salsola kali*), Kariba Weed (*Salvinia molesta*), and Salt Cedar (*Tamarix ramosissima*). With the exception of Russian Thistle and a few scattered dryland infestations of Salt Cedar, all of these invasive species were found associated with the irrigation canals crossed by the project corridors. The only aquatic invasive species identified, Kariba Weed, was found in the Reservation Main Drain at the proposed corridor crossings on Fisher, Picacho, and Stalnacker, Roads (Crossings 1, 3, and 11, indicated in Figure 2). Two of the invasive species, Kariba Weed and Salt Cedar, have a "High" rating assigned by the CIPC, and the remaining species, Russian Thistle, has a "Limited" rating.

# 4.4.2 Environmental Effects

# 4.4.2.1 Significance Criteria

An impact related to biological resources was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice.

Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Have a substantial adverse effect, either directly or through habitat modifications, on any species identified as a Candidate, Sensitive, or special status species in local or regional plans, policies, or regulations, or by the CDFW or FWS;
- Have a substantial adverse effect on any riparian habitat or other sensitive natural community identified in local or regional plans, policies, and regulations or by the CDFW or FWS;

- Have a substantial adverse effect on Federally protected wetlands as defined by Section 404 of the CWA (including, but not limited to, marsh, vernal pool, coastal, etc.) through direct removal, filling, hydrological interruption, or other means;
- Interfere substantially with the movement of any native resident or migratory fish or wildlife species or with established native resident or migratory wildlife corridors, or impede the use of native wildlife nursery sites;
- Conflict with any local policies or ordinances protecting biological resources, such as a tree preservation policy or ordinance; or
- Conflict with the provisions of an adopted HCP, Natural Community Conservation Plan, or other approved local, regional, or State HCP.

# 4.4.2.2 Impacts and Mitigation Measures

The proposed project would incorporate measures to avoid and/or minimize impacts to biological resources as detailed in the APMs below. The project would not result in significant impacts to biological resources in the project area.

**APM BIO-1:** All irrigation canals in the project area will be bored beneath and avoided during construction.

**APM BIO-2:** Bore pits will be placed a minimum distance of 5 m (16 feet) beyond either the top of the canal bank or the maximum extent of any vegetation present along the canal's margin.

**APM BIO-3:** All agricultural fields will be avoided during construction.

**APM BIO-4:** No trees will be removed during project construction. If vegetation trimming is required to complete the installations, trimming will be kept to the absolute minimum necessary.

**APM BIO-5:** All equipment and vehicles will be thoroughly cleaned to remove dirt and weed seeds prior to being transported or driven to or from the project area.

Impact BIO-1: Substantial Adverse Effects, Either Directly or through Habitat Modifications, on Any Species Identified as a Candidate, Sensitive, or Special-Status Species in Local or Regional Plans, Policies, or Regulations, or by the CDFW or FWS (Less than Significant).

Sonoran Desert Toad and Lowland Leopard Frog have the potential to occur along the irrigation canals in the project area. Implementation of the proposed project has the potential to impact these two species if individuals come into contact with construction equipment or personnel or if individuals attempt to flee the construction area and are subject to increased chances of predation or other harm. With the implementation of APM BIO-1 and 2, impacts are expected to be reduced to a less-than-significant level.

Loggerhead Shrike and Yellow-headed Blackbird have the potential to occur in the agricultural fields adjacent to the project area. In addition to potentially occurring in the agricultural fields, Townsend's Big-eared Bat has the potential to occur in other vegetated areas adjacent to the project area, such as residential landscaping, while foraging. With the implementation of APM BIO-3 and 4, impacts are expected to be reduced to a less-than-significant level.

Vermilion Flycatcher and Yuma Hispid Cotton Rat have the potential to occur in the agricultural fields adjacent to the project area and along the vegetated irrigation canals within the project area. With the implementation of APM BIO-1–3, impacts are expected to be reduced to a less-than-significant level.

# Impact BIO-2: Substantial Adverse Effects on any Riparian Habitat or other Sensitive Natural Community Identified in Local or Regional Plans, Policies, and Regulations; or by the CDFW or FWS (No Impact).

Dense vegetation along some of the canals in the project area that would be crossed by the proposed installations may provide suitable habitat for wildlife species, and the canals themselves may provide suitable habitat for fish. All of the canals in the project area would be bored beneath during the proposed installations (APM BIO-1); therefore, the project would have no impact on sensitive natural communities.

# Impact BIO-3: Substantial Adverse Effect on Federally Protected Wetlands as Defined by Section 404 of the CWA (including, but not limited to, Marsh, Vernal Pool, Coastal, etc.) through Direct Removal, Filling, Hydrological Interruption, or Other Means (No Impact).

Potentially jurisdictional riverine wetlands may be present along some of the canals in the project area. All of the canals in the project area would be bored beneath during the proposed installations (APM BIO-1); therefore, the project would have no impact on wetlands, if present.

# Impact BIO-4: Interfere Substantially with the Movement of any Native Resident or Migratory Fish or Wildlife Species or with Established Native Resident or Migratory Wildlife Corridors or Impede the Use of Native Wildlife Nursery Sites (Less than Significant).

No natural landscape blocks, potential riparian connections, or interstate connections are present within the project area (Spencer et al. 2010). The project area contains no Arizona Potential Linkage zones (Arizona's Wildlife Linkages 2006). Although native wildlife may move through the project area, the proposed installation would not create a new barrier to such small animal movement given that the proposed alignment is located along existing roadways and proposed installation would consist of buried cables and the installation of 10 equipment cabinets. No evidence of wildlife corridors was observed during the surveys. Migratory birds may be present in the areas surrounding the project corridors. With the implementation of APM BIO-3 and 4, impacts to migratory birds are expected to be reduced to a less-than-significant level.

# Impact BIO-5: Conflict with any Local Policies or Ordinances Protecting Biological Resources, such as a Tree Preservation Policy or Other Protective Ordinance (No Impact).

The proposed project would be in compliance with the ICGP's Conservation Element because all of the proposed installations would be performed in previously disturbed areas along existing roads and no new removal of undisturbed habitat would occur. There would be no impact related to local biological resource–related policies and ordinances.

# Impact BIO-6: Conflict with the Provisions of an Adopted Habitat Conservation Plan, Natural Community Conservation Plan, or Other Approved Local, Regional, or State Habitat Conservation Plan (No Impact).

Due to the presence of invasive plant species in the study area, implementation of the proposed project has the potential to result in the further spread of existing noxious weeds. Invasive species could also be introduced into the study area by construction equipment, vehicles, personnel, or imported fill or other material. Further introduction of invasive plant species could adversely impact the irrigation canals in the project area and their associated riparian areas, where present. However,

with the implementation of APM BIO-1, 2, and 5, the proposed project would be consistent with the conservation objectives of the ICGP because impacts are expected to be reduced to a less-than-significant level.

#### 4.4.3 References

Arizona Game and 2006	l Fish Department (AZGFD) Rana yavapaiensis. Unpublished abstract compiled and edited by the Heritage Data Management System, Arizona Game and Fish Department, Phoenix, Arizona.
Bureau of Reclama	tion (BOR)
2008	Species Accounts for the Lower Colorado River Multi-species Conservation Program. U.S. Department of Interior Bureau of Reclamation, Lower Colorado Division, Boulder City, Nevada.
California Departn	nent of Fish and Wildlife (CDFW)
2014	California Wildlife Habitat Relationships (CWHR) System Life History Accounts and Range Maps. Available at: http://www.dfg.ca.gov/biogeodata/cwhr/cawildlife.aspx.
California Natural	Diversity Database (CNDDB)
2014	Biogeographic Information and Observation System (BIOS). Available at https://map.dfg.ca.gov/bios/. Accessed on July 13, 2014.
Imperial County	
2008	Imperial County General Plan. County of Imperial Planning/Building Department, El Centro, California.
Sawyer John O ar	nd Todd Keeler-Wolf
2009	A Manual of California Vegetation. California Native Plant Society, Sacramento.
Spencer, W.D., P. Parisi, and A. Pettl	Beier, K. Penrod, K. Winters, C. Paulman, H. Rustigian-Romsos, J. Strittholt,M. er.
2010.	California Essential Habitat Connectivity Project: A Strategy for Conserving a Connected
	<i>California</i> . Available at: https://www.wildlife.ca.gov/Conservation/Planning/Connectivity/CEHC. Accessed on April 23, 2015.
Western Regional	Climate Center (WRCC)
2014	Yuma Quartermaster Depot, Arizona—Climate Summary. Available at: http://www.wrcc.dri.edu/cgi-bin/cliMAIN.pl?nm8668. Accessed on June 19, 2014.

# 4.5 Cultural Resources

# 4.5.1 Affected Environment

# 4.5.1.1 Regulatory Setting

#### Federal

# National Historic Preservation Act

The National Historic Preservation Act (NHPA) of 1966 (16 USC Sec. 470), as amended, is the primary Federal law governing the preservation of cultural and historic resources in the United States. The NHPA establishes the Federal government policy on historic preservation and the programs through which this policy is implemented. Section 106 of the NHPA (16 USC Sec. 470f) requires Federal agencies to take into account the effects of their undertakings on any district, site, building, structure, or object that is included in or determined eligible for inclusion in the National Register of Historic Places (NRHP) and to afford the Advisory Council on Historic Preservation (ACHP) a reasonable opportunity to comment on such undertakings (36 CFR Sec. 800.1). Section 106 is applicable to the proposed project because BIA must approve the requested ROW grants for the portions of the project located on the Fort Yuma–Quechan Reservation and BOR must approve the project's canal crossings.

To be eligible for the NRHP, cultural resources must possess integrity and meet at least one of the following four criteria specified in 36 CFR Sec. 60.4. Pursuant to 36 CFR 60.4, these are the criteria by which properties are evaluated:

The quality of significance in American history, architecture, archeology, engineering, and culture is present in districts, sites, buildings, structures, and objects that possess integrity of location, design, setting, materials, workmanship, feeling, and association, and

**A.** That are associated with events that have made a significant contribution to the broad patterns of our history; or

**B.** That are associated with the lives of persons significant in our past; or

**C.** That embody the distinctive characteristics of a type, period, or method of construction, or that represent the work of a master, or that possess high artistic values, or that represent a significant and distinguishable entity whose components may lack individual distinction; or

**D.** That have yielded or may be likely to yield, information important in prehistory or history (National Park Service 2004).

Under Section 106, a project's impacts on historic properties that affect the characteristics that qualify a property for NRHP inclusion are considered an adverse effect on the environment. Examples of adverse effects on historic properties are listed under 36 CFR Sec. 800.5(a)(2) and include, but are not limited to, physical destruction or damage to all or part of a property, change of the character or the use of the property or physical feature within the setting of the property that

contributes to its significance, or introduction of visual, atmospheric, or audible elements that diminish the integrity of significant features of the property. If an adverse effect is identified, the agency shall act pursuant to 36 CFR Sec. 800.6 (36 CFR Sec. 800.5[d][2]) to resolve the adverse effect by developing and evaluating alternatives or modifications to the undertaking that "could avoid, minimize, or mitigate adverse effects on historic properties" (36 CFR Sec. 800.6[a]). Cultural resources that have been determined ineligible for the NRHP in consultation with the State Historic Preservation Office (SHPO) and interested parties require no further consideration unless new discoveries trigger re-evaluations.

Section 106 of the NHPA does not apply to paleontological resources unless they are found in a culturally related context. In addition to the Antiquities Act (16 USC Sec. 431-433) of 1906, the preservation and salvage of fossils and other paleontological resources can be protected under the National Registry of Natural Landmarks (16 USC Sec. 461-467) and NEPA, which directs Federal agencies to "preserve important historic, cultural, and natural aspects of our national heritage."

# Archaeological Resources Protection Act

The Archaeological Resources Protection Act (ARPA) of 1979 (43 CFR Sec. 7) may impose additional requirements on an agency if Federal or Native American lands are involved. Specifically, the Act: (1) prohibits unauthorized excavation on Federal and Native American lands, (2) establishes standards for permissible excavation, (3) prescribes civil and criminal penalties, (4) requires agencies to identify archaeological sites, and (5) encourages cooperation between Federal agencies and private individuals.

# American Indian Religious Freedom Act

The American Indian Religious Freedom Act (AIRFA) of 1978 (42 USC 1996 and 1996a) affirms the right of Native Americans to have access to their sacred places. If a place of religious importance to American Indians may be affected by an undertaking, AIRFA promotes consultation with Indian religious practitioners (this may be done in coordination with Section 106 consultation). Amendments to Section 101 of the NHPA in 1992 strengthened the interface between AIRFA and NHPA by clarifying the following: (1) properties of traditional religious and cultural importance to an Indian Tribe or Native Hawaiian organization may be determined to be eligible for inclusion in the NRHP, and (2) in carrying out its responsibilities under Section 106, a Federal agency shall consult with any American Indian Tribe or Native Hawaiian organization that attaches religious and cultural significance to properties described under (1).

# Native American Graves Protection and Repatriation Act

For activities on Federal lands, NAGPRA (43 CFR Sec. 10) requires consultation with "appropriate" Indian Tribes (including Alaska Native villages) or Native Hawaiian organizations prior to the intentional excavation, or the removal after inadvertent discovery, of several types of cultural items, such as human remains and objects of cultural patrimony. For activities on Native American or Native Hawaiian lands, which are defined by statute, NAGPRA requires the consent of the Indian Tribe or Native Hawaiian organization prior to the removal of cultural items. The law also provides for the repatriation of such items from Federal agencies and Federally assisted museums and other repositories.

The 1992 amendment to the NHPA strengthened NAGPRA by encouraging "protection of Native American cultural items...and of properties of religious or cultural importance to Indian Tribes, Native Hawaiians, or other Native American groups" (Sec. 112[b][3]) and by stipulating that a Federal "...agency's procedures for compliance with Section 106 ...provide for the disposition of Native American cultural items from Federal or Tribal land in a manner consistent with Sec. 3(c) of the Native American Graves Protection and Repatriation Act..."

The final rule of the NAGPRA regulations, effective May 14, 2010, added procedures for the disposition of culturally unidentifiable Native American human remains in the possession or control of museums of Federal agencies. The rule also amended sections of NAGPRA related to purpose and applicability of regulations, definitions, inventories of human remains and related funerary objects, civil penalties, and limitations and remedies.

# Paleontological Resources Protection Act

The Paleontological Resources Protection Act, as provided in Title VI, Subtitle D, Paleontological Resources Preservation of the Omnibus Public Land Management Act of 2009 (Public Law 111-011), requires the secretaries of the interior and agriculture to manage and protect paleontological resources on Federal land using scientific principles and expertise. The law, which applies only to Federal lands, reaffirms the authority of Federal land managing agencies to implement many of the policies for managing paleontological resources, such as issuing permits for collecting paleontological resources, and maintaining confidentiality of locality data. The law provides authority for the protection of significant paleontological resources on Federal lands, including criminal and civil penalties for fossil theft and vandalism.

# State

# California Environmental Quality Act

CEQA recognizes cultural resources as a part of the environment. A historic resource is defined by CEQA as the following:

- A resource listed on or determined to be eligible by the State Historical Resources Commission for listing on the California Register of Historical Resources (Public Resources Code Sec. 5024.1, Title 14 CCR, Section 4850 et seq.).
- A resource included in a local register of historic resources, as defined in Sec. 5020.1 (k) of the Public Resource Code, or identified as significant in a historic resource survey meeting the requirements of Sec. 024.1(g) of the Public Resources Code.

# California Public Resources Code

California Public Resources Code (PRC) Sec. 5024.1 establishes the California Register of Historic Resources (CRHR), sets forth the criteria to determine significance (detailed above), defines eligible properties, and lists nomination procedures. As described in Subsection (d), resources that are automatically listed in the CRHR include those listed in or formally determined eligible for listing in the NRHP ("historic properties") and California Historical Landmarks from Number 770 onward. The CRHR criteria for eligibility are virtually identical to those of the NRHP. Cultural resources may be listed in or eligible for the CRHR if they have significance and integrity. Cultural resources are significant if they meet any of the following criteria:

• Criterion 1—Association with events that have made a significant contribution to the broad patterns of California's history and cultural heritage, or the United States (CCR Title 14, Sec. 4852[b][1]);

- Criterion 2—Association with the lives of persons important in our past (CCR Title 14, Sec. 4852[b][2]);
- Criterion 3—Embody the distinctive characteristics of a type, period, region, or method of construction, or represent the work of an important creative individual, or possess high artistic values (CCR Title 14, Sec. 4852[b][3]); or
- Criterion 4—Yield, or may be likely to yield, information important in prehistory or history (CCR Title 14, Sec. 4852[b][4]).

A resource must retain adequate integrity to be eligible for listing in the CRHR. Integrity is the authenticity of a resource's physical identity evidenced by the survival of characteristics that existed during the resource's period of significance. Integrity must be judged with reference to the particular criteria under which the resource is eligible for listing in the CRHR (14 CCR 4852[c]). Integrity assessments are generally made with regard to the retention of the following:

- Location—Where the historic property was constructed or the place where the historic event occurred.
- Design—The combination of elements that create the historic form, plan, space, structure, and style of a property. This includes organization of space, proportion, scale, technology, ornamentation, and materials. This is applicable to larger properties for the historic way in which the buildings, sites, and structures are related.
- Setting—The physical environment of a historic property. It refers to the historic character of the property. It includes the historical relationship of the property to surrounding features and open space. These include topographic features, vegetation, simple manmade paths or fencing, and the relationship between buildings, structures, or open space.
- Materials—The physical elements that were combined during a particular period of time and in a particular pattern or configuration to form the historic property.
- Workmanship—The physical evidence of the crafts of a particular culture or people during a given period in history. It may be expressed in vernacular methods of construction and plain finishes or in highly sophisticated configuration and ornamental detailing.
- Feeling—The property's expression of the aesthetic or historic sense of a particular period of time. It results from the presence of physical features that, taken together, convey the property's historic character.
- Association—The direct link between an important historic event or person and a historic property. A property retains association if it is the place where the event or activity occurred and is sufficiently intact to convey that relationship to an observer. Like feeling, association requires the presence of physical features that convey a property's historic character.

PRC Sec. 5097.5 states that any unauthorized removal or destruction of archaeological or paleontological resources on sites located on public land is a misdemeanor. "Public lands" is defined as "lands owned by, or under the jurisdiction of, the State, or any City, County, district, authority, or public corporation, or agency thereof."

PRC Sec. 5097.9 prohibits the interference with the free expression of Native American religion as provided in the United States Constitution and the California Constitution, and cause of severe or irreparable damage to any Native American sanctified cemetery, place of worship, religious or ceremonial site, or sacred shrine on public property, except on a clear and convincing showing that the public interest and necessity so require.

PRC Sec. 5097.97 promotes preservation of certain Native American cultural places located on public property, including a sanctified cemetery, place of worship, religious or ceremonial site, or sacred shrine, by ensuring access to these places by Native Americans.

PRC Sec. 5097.98 requires the Native American Heritage Commission (NAHC), upon notification by a County coroner, to notify the most likely descendants regarding the discovery of Native American human remains; enables the descendants, within 48 hours of the notification by the commission, to inspect the site of the discovery of Native American human remains and to recommend to the landowner or the person responsible for the excavation work means for treating or disposition, with appropriate dignity, the human remains and any associated grave goods; requires the owner of the land upon which Native American human remains were discovered, in the event that no descendant is identified, or the descendant fails to make a recommendation for disposition, or the landowner rejects the recommendation of the descendant, to reinter the remains and burial items with appropriate dignity of the property in a location not subject to further disturbance.

PRC Sec. 5097.99 prohibits obtaining or possessing Native American artifacts or human remains taken from a grave or cairn and sets penalties for those actions.

PRC Sec. 5097.991 states that it is the policy of the State that Native American remains and associated grave artifacts shall be repatriated.

PRC Sec. 5097.993–5097.994 (Native American Historic Resources Protection Act) states that it is unlawful to maliciously excavate, remove, destroy, injure, or deface a Native American historic, cultural, or sacred site that is listed or may be eligible for listing in the CRHR pursuant to PRC Sec. 5024.1, including any historic or prehistoric ruins, any burial ground, any archaeological or historic site, any inscriptions made by Native Americans at such a site, any archaeological or historic Native American rock art, or any archaeological or historic feature of a Native American historic, cultural, or sacred site on public land.

PRC Sec. 21083.2 states that if a project may affect a resource that has not met the definition of a historical resource set forth in Sec. 21084, then the lead agency may determine whether a project may have a significant effect on "unique" archaeological resources; if so, an Environmental impact Report (EIR) (or, if applicable, an EIR/EIS or, if authorized, a Substitute Environmental Document [SED]) shall address these resources. If the potential for damage to unique archaeological resources can be demonstrated, such resources must be avoided; if they cannot be avoided, mitigation measures will be required. The law also discusses excavation as mitigation, discusses the costs of mitigation for several types of projects, sets time frames for excavation, defines unique and non-unique archaeological resources, and sets financial limitations for this section.

PRC Sec. 21084.1 indicates that a project may have a significant effect on the environment if it causes a substantial adverse change in the significance of a historical resource; the section further defines "historical resource" and describes what constitutes a "significant" historical resource.

#### Local

The ICGP identifies areas of varying sensitivity for cultural resources and establishes policy for promoting the protection of important cultural resources (Imperial County 2008).

# 4.5.1.2 Project Setting

# Ethnography

The Quechan are a Native American people inhabiting the region around the confluence of the Gila and Colorado Rivers. The name "Quechan" literally means "those who descended" (Bee 1983:97). The name "Yuma" is the Spanish name for the Quechan and likely derives from the Akimel O'odham/Tohono O'odham name for them, *yumi*. They are one of the several Yuman-speaking groups in southern California and western Arizona. For convenience, ethnologists, beginning with Kroeber in 1943 (Stewart 1983a), have placed the Yuman people into four broad geographical groups. The Delta Yumans include such people as the Cocopah in the Colorado delta area; the Upland Arizona Yumans include the Walapai, Havasupai, and Yavapai; and the California Yuman-speakers consist of southern Californian groups such as the Kumeyaay (or Kamia) and Tipai-Ipai (or Diegueño). The fourth group, the River Yumans, comprise two closely related peoples, the Mohave and the Quechan. The Mohave and Quechan were culturally similar and, traditionally, were allied in opposition to several other groups in the area, including the Halchidhoma, the Maricopa, and the Cocopah (Stewart 1983b:56).

The following brief ethnographic account attempts to form a model of Quechan culture in pre-Reservation times (i.e., prior to 1884) while tracing the impacts from Euroamerican interaction with the Quechan people historically.

# History and Early Sources

The early records of contact between the Spanish and the Yuman Tribes that lived along the Lower Colorado are sparse. The earliest records, those of the Hernando de Alarcón and Melchior Diaz expeditions in the 1540s, do not mention the Quechan at all (Spicer 1962:262). The first substantial records of the Quechan made by Europeans were during Juan de Oñate's 1604 expedition of the Colorado River via the Bill Williams Fork (Bean and Brakke Vane 1978:5–44). The next contact with the Spanish occurred during Father Eusebio Kino's expeditions to ascertain whether California was an island or peninsula beginning in 1698 (Spicer 1962:263–264). Kino was apparently well-received by the different Yuman groups on the Colorado and Gila Rivers. Kino's last visit to the Quechan was in 1702, during his final expedition to determine California's geographical status.

The next visit from the Spanish did not occur until 1748, when the Jesuit missionary Father Jacobo Sedelmayr visited the area. However, unlike Kino, he was greeted with hostility by the Quechan. Part of the reason for this hostility was likely related to widespread epidemics among the Lower Colorado Tribes from diseases that had been introduced by Europeans. In addition, the Spanish slave trade (a practice later adopted by the Quechan) was also causing increasing hostilities elsewhere in the region (Bean and Brakke Vane 1978:5–44). In 1771, the Spanish had become fixated on establishing a permanent route between Sonora and Alta California via the Colorado River and Gila River confluence region, or what would eventually come to be known as the Yuma Route or Yuma Crossing. Spanish presence in the area accordingly intensified. The explorations for this route were led by General de Anza. At the same time, Father Franciso Garcés was busy trying find a route through Yuma country to the Hopi region for missionizing purposes, and was also conducting vigorous missionary activity among the Quechan.

Over the next 10 years, Spanish influence on the Quechan and other Lower Colorado Tribes was great due to these activities, but also because of the introduction of wheat as a winter crop and domesticated livestock (particularly poultry). The Spanish established two settlements near the crossing, the pueblos of Yuma and Xuksi'l, consisting of farmers, priests, and soldiers; these settlers allowed their cattle to graze in the Quechan fields, effectively destroying their crops (Bee 1983:94). This would occur again in 1849 during the California Gold Rush, when vast numbers of people traveled through the crossing (Bean and Brakke Vane 1978:5–47). Warfare related to the ongoing slave trade continued, as did epidemics; syphilis was introduced to the area during the 1774 De Anza expedition (Spicer 1962:264; Bean and Brakke Vane 1978:5–44).

In the summer of 1781, the Quechan successfully revolted against the Spanish, destroying both settlements and killing 95 settlers, soldiers, and missionaries (including Garcés) and taking 76 people captive (Bean and Brakke Vane 1978:5–45). The route from Sonora to Alta California via the Colorado-Gila confluence area was effectively closed off, and the Quechan remained relatively isolated until 1827, when the Quechan opened the crossing to Mexican travelers taking the slave trade road between Caborca, Sonora, and southern California (Bean and Brakke Vane 1978:5–46).

Because of the sporadic contacts between the Spanish and the Quechan, and because of the success of the revolt of 1781, the Quechan retained many of their cultural traditions and lifeways despite the Spanish enculturation of the 1770s. Nevertheless, during the course of the nineteenth century, the Quechan became increasingly subjected to Euroamerican political, religious, and economic impacts. These included the influx of would-be miners following the discovery of gold in California in 1848, the establishment of Fort Yuma in 1852, the arrival of the railroad in 1877, the establishment of the Reservation and Catholic school in the 1880s, the 1893 introduction of the Federal government's land allotment system (resulting from a local application of the Dawes Act of 1887), and irrigation projects (Bean and Brakke Vane 1978:5–48; Smith 2010; Bee 1983:94–95).

# Territory and Settlement

The Quechan account of their origin states that they, like most of the other Lower Colorado Tribes and other Tribes farther to the west (such as the Kumeyaay in the San Diego area), came from the sacred mountain of Avikame (Newberry Mountain, near Needles, California). It is here that they were created by a creator being known as *Kwikumat* or *Kukumat*. From here, they migrated south. The lands regarded as traditional by the Quechan encompass an area extending from Needles to the Gulf of California. An anthropological model hypothesizes that the Quechan, as a tribal identity, formed between the thirteenth and eighteenth centuries when several patrilineal bands formed into a tribal affinity. Group proximity during horticultural activities, linguistic affiliation, and warfare may account for this formation (Bee 1983:86).

Geographically, the Quechan were organized into a number of rancherias, each consisting of several hundred people, organized into extended family groups. The rancherias were distributed along the Colorado River north and south of the Gila confluence and along the Gila (according to some Spanish accounts, as far as 42 km [26 miles] east of the confluence). The internal structure of each rancheria changed throughout the year, with each extended family moving to their river bottomlands during the summer farming season and returning to high ground in the winter and during spring flooding. The rancherias also shifted up and down the rivers in response to food shortages and warfare (Bee 1983:87–89). Because of the warm climate, substantial housing was uncommon. Families dwelt in dome-shaped arrowweed houses and ramadas both on high ground and near their fields during the growing season. In each rancheria, one or two larger and more substantial houses

were occupied by the leading families. These houses could accommodate other rancheria members in extreme cold (Bee 1983:89–90).

#### Subsistence

Throughout their history (and presumably prehistory), the Quechan were primarily gatherers and horticulturalists, something attested to by the early Spanish chroniclers (Bee 1983:86). Wild game was not a primary source of nutrition, as the harsh desert conditions beyond the Colorado River's floodplains limited the viability of hunting. Cultivated foods included maize, tepary beans, various melons, pumpkins, and wild grass seed; other foods, such as watermelons, black-eyed beans, and wheat, were introduced by Euroamerican immigrants. Interestingly, watermelons, a crop that spread extremely rapidly among North American Native populations upon its introduction, had been adopted by the Quechan prior to Kino's visit in the late seventeenth century (Rea 1997:299).

The Quechan practiced a diversified horticultural strategy, and planting of several food crops occurred at different times of year. Maize and melons were planted in February and were not dependent on floodwater farming. Other crops were planted after the spring flooding of the Colorado River. Winter wheat was sowed in the autumn and harvested just before the floods. The wild grasses, which provided seeds to be ground into meal, were sown in less fertile soils. The other main wild foods were mesquite and screw bean pods, which were probably the primary source of nutrition during years of crop failure (Bee 1983:86–87).

As discussed earlier, both cultivated and wild foods were affected by the arrival of Euroamericans, who would allow (or could not prevent) cattle to graze in Quechan fields. In 1893, a long-term impact was made on Quechan horticulture by an agreement (based on the Dawes Severalty Act of 1877) that persuaded Quechan farmers to limit their land holdings to 2 ha (5 acres) per person. All remaining land was then sold at public auction. This was a direct move by non-Natives to acquire the fertile bottomlands of the Colorado River that the Quechan had farmed for centuries. The allotments were increased to 4 ha (10 acres) in 1912. Meanwhile, the Yuma Project had been initiated by the U.S. Reclamation Service (later the BOR) in 1904 and had the effect of disrupting the annual flooding and silt deposition of the Colorado River. By the 1920s and 1930s, farming was no longer a viable occupation, with many Quechans becoming wage workers in Yuma. After years of claiming that agreement was signed under duress and that the U.S government had not fulfilled its terms, 10,117 ha (25,000 acres) of land that had belonged to the original 1884 reservation were restored to the Quechan Tribe in 1978 (Bee 1983:94–95). Today, most of the farmland is leased to non-Native farmers.

# Kinship and Polity

Socially, the Quechan were organized into patrilineal clans. The clans were exogamous units, with clan names borne exclusively by women. Some clan names may have originated from other Tribes, such as the Mohave, Maricopa, or the Kumayaay. The rancherias were agamous; that is, anyone could marry outside the rancheria, but men most frequently married women from their own rancheria. Consequently, settlement was in practice bilocal, an important factor for the extended family as the primary economic unit (Bee 1983:89). Clan membership did not necessarily correspond to rancheria affiliation. Clan functions were largely disregarded by the 1960s, and many Quechans had forgotten their affiliation by that time (Bee 1983:90–91).

In general, the clan and rancheria were the basic social units among the Quechan, with the extended family the economic unit, as mentioned above. Tribal consciousness, when all the people identified

as "Quechan" rather than as members of the smaller-scale social units of clan and rancheria, occurred during warfare, harvest gatherings, and annual mourning ceremonies (Bee 1983:92).

Early European sources described two main leadership positions among the Quechan, one leading civil affairs and one in charge of warfare. However, it seems that these roles may have been largely traditional rather than consisting of any real political power. In practice, decisions were made by the leaders of individual rancherias, who probably consulted in council for matters of concern on the tribal level (Bee 1983:92–93). Although some degree of inheritance may have been a factor in determining leaders, competence was a more powerful attribute. Competence depended upon public approval, but also upon personal power bestowed by special dreams (Bee 1983:92–93). The dreams of a leader or candidate for leadership were evaluated by a group of elders, and the individual was required to experience dreams appropriate to his office, although he was also required to be an effective leader.

# Warfare

Warfare was a cornerstone of Quechan culture. Two types of warfare were distinguished: the war party and the small raiding party (Bee 1983:93). The raiding party was focused on creating havoc and capturing horses or captives. Conflicts involving the war party consisted of a village raid followed by an arranged battle in which the opposing parties faced one another in two lines, ending in a hand-to-hand melee (McCorkle 1978:698). Bee (1983:93) points out that this had greater resemblance to a brutal team sport, where the two sides would agree upon weapons to be used and wait to attack until both sides had fallen into formation. The arsenal consisted of a "potato masher" war club of mesquite wood (typically a tapered cylinder mounted on a handle), wooden spears with fire-hardened tips, and bows. Because of their distinctive war club, the Quechan are referred to by the Spanish word "Garroteros," literally, "clubbers" (Bee 1983:97; Kroeber 1925:782).

Warfare among all the Yuman Tribes was closely intertwined with myth and ceremony, although casualties were real and occasionally heavy. An account of the first war party is given in the central creation myth. Traditionally, the function of warfare among the Lower Colorado Tribes was connected to tribal prestige and ritual, rather than conflict over resources or similar, comparatively mundane concerns. For example, when a sorcerer was killed, this was an act that often precipitated group conflict. This is again connected to the importance of dreams in Yuman culture: dreams of success in battle were highly valued and became incorporated into song cycles. In addition, like the rancheria leaders, war leaders, ceremonial managers, and shamans, obtained their position through dreams (McCorkle 1978:698–699).

The Quechan and Mohave (to whom they are closely related culturally and linguistically) did not usually fight one another, but both engaged in conflicts with the Maricopa and Cocopah, who were sometimes allied with the Pima. There was likely a long history of warfare among the Yuman Tribes that predated the arrival of Europeans. However, warfare may have increased in scale and intensity during the eighteenth and early nineteenth centuries for economic reasons—a departure from the tradition of "ritual" warfare (Bee 1983:93). The motivation for waging war appears to have been related to the taking of captives to trade to the Spanish and other Tribes for horses and other goods. It appears, however, that land acquisition was still not a motivation for war.

# Death and Mourning

Mourning, along with dreaming and warfare, was one of the three most important aspects of the Quechan lifeway. Upon an individual's death, all of his or her belongings, including the family home,

were destroyed or given away. This sometimes left the deceased's family destitute, and they would be provided for by friends or the rancheria leaders (Bee 1983:89). Inheritance was therefore never an important factor in pre-Reservation life. Individual family garden plots were also abandoned, to be used later by non-family members. The *keruk* ceremony, the central mourning ceremony of the Yuman Tribes, including the Quechan, was held after the death of an important leader or after an accumulation of deaths to be honored by the families of the deceased (Bee 1983:93). The *keruk* is alternatively known in older literature as *nyimits* (Kroeber 1925) or *nimits* (Curtis 1906).

A central component of the *keruk* ceremony was a mock battle, prepared for and carried out in the same way as an actual conflict. It also was a reenactment of the battle that was fought following the death of the creator deity Kwikumat. The ceremony also involved the singing of songs commemorating the creation of the world, public mourning, and the destruction of the deceased's property. The ceremony was intertribal and lasted several days, forming an occasion for large-scale social interaction wherein goods were exchanged, marriages were arranged, and enmittees were resolved.

The *keruk* appears to have been associated with a pilgrimage trail between Pilot Knob (approximately 10.86 km [6.75 miles] west of modern Winterhaven) and Newberry Mountain (the sacred mountain *Avikame*). Altschul and Ezzo (1995) have noted that the practice of the *keruk* seems to have intensified during the eighteenth and nineteenth centuries, contemporaneous with the intensified conflicts resulting from the horses-for-slaves trade introduced by the Spanish and with an influx of people migrating from the desiccating Lake Cahuilla. They suggest that the *keruk* and the associated pilgrimage was a unifying force transcending conflicts between inimical Tribes. Altschul and Ezzo likewise suggest that the intaglios along the trail, which are executed in different styles, were the locations of *keruk* rites unique to and performed by different Tribes. The *keruk* has continued into modern times in modified form (Bee 1983:96–97).

#### Historic Context

# Spanish Period

The first entry into what is now Arizona by people of European descent came in the late 1530s. A group of four men, including Álvar Nuñez Cabeza de Vaca, who survived a 1528 shipwreck on the coast of the Gulf of Mexico and then wandered across the Southwest before finally reaching Spanish-held territory in Sonora in 1536, may have passed through the state, although this has been questioned in recent years. Marcos de Niza, a priest dispatched as an advance scout for an expedition into the lands through which the Cabeza de Vaca party supposedly passed, likely explored the eastern part of the state in 1539, although his activities, too, have been called into question by modern researchers. The first European to unequivocally enter Arizona was Francisco Vasquéz de Coronado, who passed through the state on his way to the Pueblo area in New Mexico in 1540. As an adjunct to Coronado's expedition, Hernando de Alarcón was sent by sea up the west coast of Mexico with the intention of linking up with Coronado at some unspecified place. Alarcón discovered the mouth of the Colorado River and a crossing spot at Yuma, but his visit would not lead to any permanent Spanish presence in western Arizona. A few months later, the spot was visited by a second Spanish expedition led by Melchior Díaz, who traveled overland from Sonora via a trail that he would name the Camino del Diablo in order to meet up with Alarcón. Díaz was too late to meet up with Alarcón, but found a message left by his countryman. Alarcón and Díaz described the lower Colorado River area as a war-torn region and mentioned native groups they

identified as the Quiquima or Quicoma and Koxwan or Ciana (koxkha'n). It is not clear who these people were, but they are thought to be the Quechan or Kouanas (Howell 2014).

Over the course of the sixteenth and seventeenth centuries, the Spanish pushed their northern frontier inexorably northward from central Mexico. While they penetrated into present-day New Mexico in the late sixteenth century, establishing a colony along the Rio Grande north of present day Albuquerque in 1598, no comparable presence was established in Arizona until roughly a century later, and this settlement (at least initially) took on a very different form. In the 1680s, Jesuit missionaries, led by the Austrian Eusebio Francisco Kino, began to establish missions in Baja California and northern Sonora, the Sonoran missions ultimately extending north of the modern International Border into Arizona. Most of the Sonoran missions were located along a north-south axis, which, north of the border, corresponds to the Santa Cruz River Valley. One exception, the most remote of the Sonoran missions, was Nuestra Señora de Loreto y San Marcelo de Sonoyta, located about 80.5 km (50.0 miles) southeast of Dateland. This community was (and is) located on the Camino del Diablo pioneered by Díaz 150 years earlier. The Camino del Diablo never became a heavily traveled route, but it was periodically used by missionaries to move overland between the Sonoran and Baja California missions. In 1774, military officer Juan Batista de Anza used the trail to lead a party of 200 colonists overland to California. The colonists settled at Monterrey while Anza himself and a small scouting party proceeded north and reconnoitered the sites for what would become the Presidio of San Francisco and the Mission San Francisco de Asís (Howell 2014).

Kino had visited the confluence of the Gila and Colorado Rivers during expeditions in 1700 and 1701. Kino was the first to refer to the people inhabiting the region, who called themselves the Kwichyana or Kuchiana, as the Yuma or Yuman. The misnomer "Yuma" derived from the missionaries' misunderstanding of the word "*yab-may-o*," meaning "son of a captain" or chief. Following these visits, interaction between the Spanish and the Quechan increased significantly. Nearly a century later, two missions and accompanying settlements were established north of the confluence. The Spanish recognized the strategic importance of the Colorado River crossing at Yuma and consequently desired to remain on good relations with the Quechan. However, disputes over resources between settlers and natives led to a native uprising in 1801. Following the uprising, interactions between Europeans and the Quechan were minimal until the American period (Howell 2014).

# American Period

Following a relatively short interval (A.D. 1821–1848) during which California and the Southwest was controlled by newly independent Mexico, the United States gained possession of most of Arizona with the Treaty of Guadalupe Hidalgo; they gained the remainder with the Gadsden Purchase of 1853. California attained statehood in 1850, becoming the 31st state. The 1850s were particularly tumultuous for the Yuman speaking peoples along the lower Colorado River. With the onset of the California Gold Rush following the discovery of gold at Sutter's Mill in 1848, hostilities erupted as increasing numbers of Euroamerican fortune hunters headed west into California. In the lower Colorado River region, the conflicts between Native Americans and would-be miners resulted in the development of Camp Yuma in 1852, after which time the Quechan lost control of the lands around the Yuma Crossing. In 1858, the Mohave War began following a Mohave attack on the Beale's Road immigrant trail (the Battle of Beale's Crossing). This led to the establishment of Fort Mohave near Topoc, the second major U.S. military outpost on the Colorado River, in 1859. In 1860, the U.S. Army defeated the Mohave in the last major conflict in the lower Colorado River region (Howell 2014).

The military post of Fort Yuma had originally been established in 1849 as Camp Calhoun, later becoming known as Camp Independence and then Camp Yuma. The initial purpose of the camp was to protect the nascent settlement of Colorado City (which would eventually become Yuma) and its strategically located river crossing from the Quechan, who were hostile to the incursion of the settlers. The cost of maintaining the post led to a brief period of abandonment in 1851, but it was re-established in 1852 as thousands of gold seekers began passing through the Yuma Crossing. While the California Gold Rush was the primary impetus for the growth of Colorado City, the settlement expanded when it was recognized that bringing goods via ship to the mouth of the Colorado River and distributing them from the fort was an effective means of getting supplies to other military outposts across the Southwest. This led to the establishment of the U.S. Army Quartermaster Depot, which was in operation from the 1860s until the 1880s (Howell 2014).

Colorado City burgeoned as the result of being both a seaport and a major crossing point on the river for travelers and immigrants heading west. After virtual destruction resulting from major flooding in 1862, Colorado City was rebuilt and renamed Arizona City. Following the Civil War, rather elaborate plans were made for the city's continued development as a commercial center. Arizona City was formally incorporated in 1871 and renamed once again as Yuma in 1873. In 1876, the Yuma Territorial Prison was constructed on a hill across from the fort, where it operated for 33 years until it was relocated to Florence, Arizona, because of overcrowding (Arizona State Parks 2014). In 1877, the first locomotive to cross the Colorado River entered Arizona at Yuma, inaugurating the long-anticipated establishment of the railroad in the state. Four years later, the Southern Pacific Railroad connected with the Texas Pacific Railroad east of El Paso (Howell 2014).

In 1884, the Fort Yuma Reservation was established for the Quechan on the western (California) side of the river. Prior to this time, the Quechan occupied six rancherías situated above the Colorado floodplain, moving to family farm plots on the floodplain during the growing season after the spring floods and until autumn. It is estimated that the Quechan derived 30–50 percent of their subsistence from agriculture, supplementing a mixed foraging and hunting economy. Quechan families gradually abandoned this lifeway following the establishment of the Reservation, where they were allocated 4-ha (10-acre) plots of farmland under the Dawes Severalty Act of 1887, which in turn opened up the remainder of the traditional lands for settlement by non-natives. In 1893, the extent of the reservation was drastically reduced by the U.S. government, which limited reservation lands to 2 ha (5 acres) per living person. Much of the original reservation land was returned to the Quechan in the 1970s (Howell 2014).

Fort Yuma itself continued as a military installation until 1883, when its management was transferred to the U.S. Department of the Interior. The end of the Civil War and the declining conflicts with Native Americans further rendered the fort unnecessary. In addition, the arrival of the railroad in 1877 had obviated the need for the military's use of the quartermaster's as a supply distribution hub. Military operations in the Yuma region would remain dormant until the establishment of the Yuma Proving Grounds during World War II (Howell 2014).

Much of the subsequent history of Yuma pertains to agriculture and the management of the Colorado River. The Yuma Project, an ambitious endeavor to irrigate the lower Colorado River valley, was initiated by the U.S. Reclamation Service (later the BOR) in 1904. The Reclamation Service took over the abandoned Fort Yuma facilities as its headquarters. The first project was the Laguna Dam, which was constructed from 1905–1909. Laguna Dam, located about 21 km (13 miles) northeast of Yuma, gave rise to the construction of several canals, including the Yuma Main Canal

(AZ X:6:67[ASM]) and its laterals and the the East Main (AZ X:6:65[ASM]) and West Main Canals (AZX:6:63[ASM]), both of which split from the Yuma Main in the town of Yuma after diversion beneath the river via the Colorado River Siphon (Stene 1996:8-9). Construction on the Colorado River Siphon (AZ X:6:40[ASM]) began in 1909 and was completed three years later. A 4.2-mdiameter (14.0-foot-diameter) tunnel was excavated through the sandstone underlying the river for a distance of nearly 305 m (1,000 feet). The tunnel was lined with concrete and was connected to two 22.5-m-deep (74.0-foot-deep) vertical shafts on either side of the waterway. The Laguna Dam successfully weathered the severe flooding of 1912 and continued diverting water until 1948, when it was superseded by the Imperial Dam (completed 8 km [5 miles] upstream from the Laguna Dam in 1938) and the All-American Canal. The All-American Canal replaced the Alamo Canal, a significant segment of which flowed through Mexico. In order to establish a canal that was located exclusively on U.S. lands, the All-American Canal was constructed by the BOR beginning in the 1930s. By 1942, it became the sole water source for Imperial Valley. The All-American Canal feeds the BWD, which was established in 1927 by water users from the Reservation Division of the Yuma Project. The BWD maintains the Reservation Division, which consists of 3,058 ha (7,556 acres) of land on the Fort Yuma–Quechan Reservation, and the Bard Division, which consists of 2,881 ha (7,120 acres) of private land (Howell 2014).

To encourage travel along the proposed Ocean-to-Ocean Highway (U.S. Highway 80) that would connect southern California with the rest of the United States, the Ocean-to-Ocean Bridge was constructed across the Colorado River at Yuma in 1915. Construction of the bridge was a joint effort of the Office of Indian Affairs and the states of California and Arizona, and it was fervently promoted by Yuma's business community. When completed, it was the only highway bridge crossing the Colorado River for some 1,931 km (1,200 miles). For a time during the Great Depression, a checkpoint was established by the State Police on the California side of the bridge to prevent the massive influx of people migrating west in search of employment. If the "Okies" or "Arkies" had no money or lacked proof of a job waiting in California, they were not allowed to enter the state. Many of those who were turned away set up camp in Yuma, and a neighborhood still bears the unofficial designation "Okietown." The bridge continued as a crossing point for vehicular traffic until 1988, when it was determined to have become structurally unsound. However, at some point, the bridge was reopened to vehicles, as it currently serves as an access point to the Fort Yuma–Quechan Reservation. The bridge is now listed on the NRHP (Howell 2014).

Following the United States' entry into World War II, combat training centers were established across the desert Southwest. The harsh desert conditions were considered ideal to prepare soldiers for combat overseas, particularly in North Africa. Camp Young, located in the Mojave Desert between Indio and Desert Center, California, served as headquarters of the Desert Training Center (DTC). Major General George S. Patton was Camp Young's first commanding officer and was assigned the task of selecting other desert locations for additional training areas. Ten other camps were established across the California-Arizona Maneuver Area (CAMA). Over a million men trained at the DTC/CAMA from 1942–1944, when the camps were closed. Camp Pilot Knob (in California) and Camp Laguna (in Arizona) were located in the Yuma vicinity. In 1943, the Yuma Test Branch was established downriver from the Laguna Dam for the purpose of testing portable combat bridges. The Yuma Test Station became the main artillery and armament testing range in the United States. It was later renamed the Yuma Proving Ground and remains an important military installation today (Howell 2014).

#### Paleontology

The geology of the project area consists of alluvial deposits dating from the late Holocene to historic times. Holocene deposits are generally considered too young to contain fossilized remains.

#### **Research Methods**

Prior to fieldwork, a Class I records search was performed. The Class I search examined all previously conducted surveys and previously recorded sites and historic properties within a 1.6-km radius (1.0-mile-radius) buffer zone extending from the project footprint. Although the project's area of potential effects (APE) is located only on the California side of the state line, the buffer zone extends into Arizona as well. The Class I research was completed through consultation with the California Historical Resources Information System (CHRIS) for the California portion of the buffer and via the Arizona State Museum's (ASM's) AZSITE online database for the Arizona portion. In addition, a Sacred Lands File (SLF) request was filed with the California Native American Heritage Commission (NAHC), and U.S. General Land Office (GLO) maps for the relevant Township and Range designations within both California and Arizona were also checked for indications of historic properties in the vicinity of the APE.

#### **Records Search**

#### California

The Class I records search found that 43 surveys have been previously conducted and 9 sites have been previously recorded within the California portion of the 1.6-km (1.0-mile) buffer zone surrounding the project area (see Appendix E, Cultural Resources Report, Tables 1 and 2, Figure B.1). In addition, one historic address (the Fort Yuma Train Depot) is present within the buffer zone (see Appendix E, Cultural Resources Report, Table 3).

Three linear, non-canal sites are present within the buffer. One of these sites, CA-IMP-7158, the historic Pilot Knob-Tap Drop 4 161kV Transmission Line, crosses the APE at two points. The line is supported, at least in the vicinity of the APE, by wooden towers and is currently in use. The line has been upgraded and maintained since its construction in the 1940s. The line crosses the APE near the intersection of Picacho Road and Indian Rock Road and again along Cocopah Road (see Appendix E, Cultural Resources Report, Photo 14). Another site, CA-IMP-3456, is described as a "road course NE and SW" and is apparently based on a GLO surveyor's notes from 1856. According to the site card, this site is now in Arizona because of a change in the course of the Colorado River. However, no indications of the site exist in the AZSITE database. Finally, a portion of the historic Southern Pacific Railroad (SPRR) passes through the buffer and crosses the APE along First Avenue. The SPRR (which was purchased by the UPRR in the 1990s) was constructed beginning in the 1870s and ran from the Los Angeles area to Yuma and subsequently further into Arizona. The line has been in active use since its original construction. Over the past several decades, a number of surveys in southern California have recorded segments of the SPRR and various features related to it. One such feature is the railroad bridge over the Colorado River, located adjacent to the Ocean-To-Ocean Bridge. This and several other railroad bridges in the vicinity (such as the bridges that cross the Yuma Main Canal and the All-American Canal) are subsumed under site number CA-IMP-3424.

The remaining three sites are historic canals, each presently in active use. The canals consist of the Yuma Main Canal (CA-IMP-6830), the Reservation Main/Cocopah Canal (CA-IMP-6832), and the Reservation Main Drain Canal (CA-IMP-6824).

#### Arizona

The Class I records search found that 18 surveys were previously conducted and 22 sites were previously recorded within the Arizona portion of the 1.6-km (1.0-mile) buffer zone surrounding the project area (see Appendix E, Cultural Resources Report, Tables 4 and 5, Figure B.2). There are also 22 historic properties and 3 historic districts listed on the NRHP within the buffer zone (see Appendix E, Cultural Resources Report, Tables 6 and 7, Figure B.3). At least two of the properties, the Ocean-to-Ocean Bridge and the Gandolfo Theater, are cross-listed as archaeological sites and historic properties. These properties lie within Yuma or along the Colorado River.

# GLO Maps

GLO maps for the relevant Township and Range designations within both California and Arizona were checked for indications of historic properties in the vicinity of the APE. The maps were accessed via the Bureau of Land Management (BLM) GLO Records website (BLM 2014). All maps on which the APE is located were dated February 6, 1857. The APE itself crosses few properties: a "Cottonwood" along what today would be Picacho Road and an "Indian Field" on the northern end of the APE at Stalnacker Road and Flood Road, which is still a cultivated area today. Within the 1.6-km (1.0-mile) buffer, historic properties include Fort Yuma; the "Settlement of Captain Ankrum," which corresponds approximately to the location of modern Winterhaven; and "Western's House." Several sections note that "there are some Indian villages in this Section."

# Native American Heritage Commission Coordination

Tierra sent a Sacred Lands File and Native American Contacts List request to the Native American Heritage Association (NAHC) on September 15, 2014. NAHC responded on September 21, 2014, stating that their records search failed to indicate the presence of Native American cultural resources in the immediate project area.

# Field Survey

Tierra archaeologists, accompanied by a Quechan Tribal monitor, performed a Class III cultural resources survey of the proposed project area on July 15 and 16, 2014, and returned to the project area on March 12, 2015, to survey the minor alterations made to the project route in February of 2015.

No new prehistoric archaeological sites were observed during the surveys. One property, the Walapai Canal (Primary Site Number P-13-014813), was newly recorded as a historic site. In addition to the canal, several isolated occurrences were recorded. Although not considered an archaeological site, the Fort Yuma–Quechan Indian Reservation Cemetery was also noted as an important cultural landmark in proximity to the APE. The site records on file at the South Coastal Information Center (SCIC) for the Yuma Main Canal (CA-IMP-6830), the Reservation Main/Cocopah Canal (CA-IMP-6832), and the Reservation Main Drain Canal (CA-IMP-6824) were updated to reflect observations made where the canals cross the current APE. No new cultural resources were identified during the March 2015 survey. All of these properties are described below.

# Isolated Occurrences

Ten isolated occurrences were observed (see Appendix E, Cultural Resources Report, Table D.1, Figure D.1). All of the lithic artifacts (n = 6) could only be tentatively identified as flaked stone. The fact that these isolated occurrences were in each case discovered on road shoulders or near the margins of cultivated fields (that is, highly disturbed areas) raises two issues. First, it is possible that

in some cases an item may have been produced by machinery (such as road grading equipment or tractors) impacting naturally occurring rocks. Second, in all cases, it is highly unlikely that the artifacts are in their original locations or contexts. One artifact, a possible quartzite tool (IO 5), is the item most likely to be an actual artifact (see Appendix E, Cultural Resources Report, Photo 21). Three artifacts were identified as historic or possibly historic glass; at one location, the glass was accompanied by a white earthenware plate fragment. One isolated occurrence consists of a roadside memorial shrine (IO 10) located at the southwest corner of the intersection of Picacho Road and Arnold Road. It does not appear to be historic, but it was recorded with the intent of documenting its location for avoidance.

#### Walapai Canal (P-13-014813)

The Walapai Canal (assigned primary site number P-13-014813) was constructed between 1908 and 1910 (Stene 1996:9). The Walapai branched from the Yuma Main Canal at the Siphon Drop Power Plant, near the point where the Yuma Main splits from the All-American Canal. From there, it flows 3.10 km (1.93 miles) to its southern terminus. Today, the Walapai Canal appears on maps as the Walapai Lateral (see see Appendix E, Cultural Resources Report, Figure D.1).

The APE crosses the Walapai Canal along Arnold Road (see Appendix E, Cultural Resources Report, Photo 22). At the crossing point, the canal is of earthen construction, but there is a concrete distribution box at this location. The canal south of this point was not explored or recorded, but this distribution box appears to form the southern terminal end of the canal, except for an extension to its south measuring a few hundred feet in length paralleling First Avenue. The box measures approximately 9.1 m (30.0 feet) long by 1.8 m (6 feet) wide. It is not clear when the box was constructed, but it uses modern metal gates for its distribution openings; slots remain from the wooden gates that it once used. The canal itself is trapezoidal in cross-section (and close to triangular) and measures approximately 5.5 m (18.0 feet) at its top width with an estimated depth of about 1.5 m (5.0 feet).

#### Cemetery

It was noted that the APE passes near the Fort Yuma–Quechan Indian Reservation Cemetery located at the interchange of Quechan Drive, Picacho Road, and Sapphire Lane. The APE does not encroach upon the cemetery; however, the cemetery was noted to allow for the recommendation of monitoring in the vicinity during the construction work (see APM CR-1 below).

#### The Yuma Main Canal (CA-IMP-6830)

The APE crosses the Yuma Main Canal (also known as the California Main Canal) at a point along Arnold Road to the west of the Arnold Road/Picacho Road intersection (see Appendix E, Cultural Resources Report, Figure B.1). Arnold Road is bridged at the canal crossing. Today, the Yuma Main Canal continues to convey a large volume of water from the All-American Canal to the south (see Appendix E, Cultural Resources Report, Photos 15 and 16). The Yuma Main Canal is a large earthen canal. It was constructed as a diversion canal originating from the Laguna Dam. Construction of the canal began in 1909 and was completed by 1912. The Yuma Main originally diverted water from the Laguna Dam, but this diversion was discontinued in 1941 following the construction of an earthen dike across the canal. After this time, the canal began to divert water from the Siphon Drop Spillway along the All-American canal. The Yuma Main continued through the Reservation Division to the Colorado River Siphon, where it passed beneath the river into Yuma and the Arizona side, and to the Valley Division of the Reclamation Service's (later the BOR) Yuma Project. In Yuma, the Yuma Main was split into the East and West Main Canals (Howell 2014).

In Arizona, the Yuma Main Canal, the Colorado River Siphon, the East Main Canal, and the West Main Canal have all been recorded as archaeological sites (AZ X:6:67, X:6:40, X:6:65, and X:6:63[ASM], respectively). The canals (but not the siphon) have all been determined individually eligible for inclusion on the NRHP by the Arizona SHPO. However, it does not appear that the California reach of the Yuma Main Canal has been officially recorded as a historic site or been evaluated for its NRHP status.

At the crossing at Arnold Road, the canal measures roughly 38 m (125 feet) in width. Because the canal currently conveys a large volume of water, it was not possible to determine the canal's other dimensions or its shape in cross-section. However, according to the existing Historic Resources Inventory Record for this property, the canal bottom averages 15 m (50 feet) in width, and the sides slope 1.25:1 with a water depth of about 2.7 m (9.0 feet).

# Reservation Main/Cocopah Canal (CA-IMP-6832)

Construction on the Reservation Main/Cocopah Canal began in 1907; construction on an extensive system of laterals from the Reservation Main commenced the following year. The Reservation Main originally split from the Yuma Main Canal at Indian Heading. The Mojave and Cocopah Canals were split from the Reservation Main. The canal continues to convey a moderate volume of water. Today, the Reservation Main flows westward along Heyser Road and turns south at the interchange of Heyser Road, Stalnacker Road, and Avenue E, where it joins the Cocopah Canal (Howell 2014).

The APE does not cross the Reservation Main Canal proper, but it does come within close proximity of it at the road interchange (see Appendix E, Cultural Resources Report, Figure B.1). However, the APE does cross the Cocopah Canal along Ross Road and it parallels the canal along Cocopah Road. The APE also crosses the Cocopah Canal at Picacho Road (see Appendix E, Cultural Resources Report, Photo 17), Ross Road (see Appendix E, Cultural Resources Report, Photo 18), and the intersections of Flood Road and Haughtelin and Arnold Roads. Because the Cocopah Canal (along with the Mojave Canal, which is not crossed by the APE) was historically a diversion of the Reservation Main, it is considered a component of the same system and was not recorded as a separate site. Much of the Cocopah Canal has been lined with concrete, but portions of it remain earthen, such as at its crossing at Picacho Road.

# Reservation Main Drain Canal (CA-IMP-6824)

The Reservation Main Drain Canal spans the Fort Yuma–Quechan Reservation and serves as a drainage for field runoff (see Appendix E, Cultural Resources Report, Figure B.1). It empties into the Colorado River about 0.8 km (0.5 miles) downstream from the SPRR Bridge. It was constructed between 1912 and 1914 and was designed to drain excess water from the very flat lands in the river valley, which have a high water table. This waterway may also be indicated as a "Ditch" in Sections 23 and 26 on a BLM plat of Township 16 South, Range 22 East, SBB&M, dated September 7, 1951. However, only a segment of the ditch appears on the map. The APE crosses the Reservation Main Drain along Picacho Road (see Appendix E, Cultural Resources Report, Photos 19 and 20), Arnold Road, Fisher Road, and Stalnacker Road. At each location, the canal is of earthen construction with a top width of approximately 7.6 m (25.0 feet). The canal is in active use and it was not possible to estimate its bottom width, but the Historic Resources Inventory Record indicates that its bottom width is 4.3 m (14.0 feet) and its average water depth is 0.9 m (3 feet) (Howell 2014).

# 4.5.2 Environmental Effects

The proposed project involves the use of existing infrastructure in the subject area. The proposed project alignment is located within existing public ROWs that have been previously disturbed. The proposed installation involves minimal ground disturbance, as required for installing underground conduit and cables. Therefore, there is a low probability for the proposed project to affect cultural resources in the subject area. Nevertheless, cultural resources could be discovered during any ground-disturbing activities conducted for the proposed project.

For a built resource to be listed in or be considered eligible for the NRHP or the CRHR, it must retain the essential character-defining features that enable it to convey its historic identity. These features are those that define both why a property is significant and the period during which it acquired its significance. Furthermore, each type of property depends on certain aspects of integrity, more than others, to express its historic significance. Determining which of the aspects is most important to a particular property requires an understanding of the property's significance and its essential physical features from the resource's period of significance.

Impacts on cultural resources could potentially occur if the project were to result in any of the following:

- Substantial adverse changes in the significance of a historical resource either listed or eligible for listing on the NRHP, the CRHR, or a local register of historic resources.
- Substantial changes in the significance of a unique archaeological resource, destruction of a unique paleontological resource or site, or disturbance of human remains, including those interred outside of formal cemeteries. Paleontological resource sensitivity is defined as follows.
- Paleontologic sensitivity is defined as the potential for a geologic unit to produce scientifically significant fossils. This is determined by rock type, past history of the rock unit in producing significant fossils, and fossil localities that are recorded from that unit. Paleontologic sensitivity is derived from the fossil data collected from the entire geologic unit, not just from a specific survey.

# 4.5.2.1 Significance Criteria

An impact related to cultural resources was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice.

Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Cause a substantial adverse change in the significance of a historical resource as defined in CEQA § 15064.5.
- Cause a substantial adverse change in the significance of an archaeological resource pursuant to CEQA § 15064.5.
- Directly or indirectly destroy a unique paleontological resource or site or unique geologic feature.
- Disturb any human remains, including those interred outside of formal cemeteries.

#### 4.5.2.2 Impacts and Mitigation Measures

The proposed project would incorporate measures to avoid and/or minimize impacts to cultural resources, as detailed in the APMs below. The project would not result in significant impacts to cultural resources in the project area.

**APM CR-1:** The Pilot Knob-Tap Drop 4 161kV Transmission Line will be avoided during construction.

**APM CR-2:** The UPRR will be bored beneath and avoided during construction.

**APM CR-3:** All construction activities will be monitored by a qualified archaeologist and/or Tribal member.

**APM CR-4:** If human remains are discovered or recognized in any location other than a dedicated cemetery, TDS will suspend further excavation or disturbance of the site and any nearby areas reasonably suspected to overlie adjacent human remains until the County coroner has been informed and has determined that no investigation of the cause of death is required.

**APM CR-5:** If human remains of Native American origin are discovered on Federal land during ground-disturbing activities, pursuant to the NAGPRA, the contractor will:

- Notify the county coroner or the sheriff;
- Notify, in writing, the responsible Federal agency; and
- Cease activity in the area of discovery and protect the human remains.

**APM CR-6:** In the event that fossil remains are encountered, either by the cultural resources monitor or by construction personnel, qualified paleontological specialists will be contacted. Construction within 30.5 m (100.0 feet) of the find in non-urban areas and 15 m (50 feet) in urban areas will be temporarily halted or diverted until a qualified vertebrate paleontologist examines the discovery.

# Impact CR-1: Cause a Substantial Adverse Change in the Significance of a Historical Resource as Defined in CEQA § 15064.5 (Less than Significant).

The proposed project would cross the historic Pilot Knob-Tap Drop 4 161kV Transmission Line (CA-IMP-7158), the Southern Pacific Railroad (today the Union Pacific Railroad) (CA-IMP-3424), the Yuma Main Canal (CA-IMP-6830), the Reservation Main/Cocopah Canal (CA-IMP-6832), the Reservation Main Drain (CA-IMP-6824), and the Walapai Canal (P-13-014813). All six of these sites have been recommended eligible for inclusion in the NRHP under Criterion A and SHPO's concurrence for the BIA's recommended "No Adverse Effect" determination has been received (see Appendix E) regarding the proposed project's potential impacts on these resources. Impacts due to the proposed project would be less than significant because all six sites have been in continuous service since their inception and are regularly maintained. In addition, the transmission line would be avoided during construction (APM CR-1), the railroad would be bored beneath (APM CR-2), and all canals in the project area, including the four canal sites listed above, would be bored beneath during construction (see APM BIO-1 and 2).

It is possible that undiscovered historical resources may be present in the project area and, if present, these resources could be impacted during the ground-disturbing activities associated with the proposed installations. In order to maintain these potential impacts to a less than significant level, APM CR-3 would be implemented during construction. Therefore, impacts to historical resources would be less than significant.

# Impact CR-2: Cause a Substantial Adverse Change in the Significance of an Archaeological Resource Pursuant to CEQA § 15064.5 (Less than Significant).

There are no archaeological sites present in the proposed project area and the isolated occurrences identified are considered to be "nonunique" archaeological resources as defined by CEQA §15064.5(c)(4) and §21083.2(h). According to these statutes, a "nonunique archaeological resource need be given no further consideration" and "the effects of the project on those resources shall not be considered a significant effect on the environment" (California Association of Environmental Professionals 2014:35, 134). As such, the documentation of the isolated occurrences is considered complete and the proposed project would have no impact on these resources.

It is possible that undiscovered archaeological resources could be present in the project area. If present, these resources could be impacted during the ground-disturbing activities associated with the proposed installations. Depending on the nature of the materials and the extent of the disturbance and/or damage, impacts could be significant. In order to maintain these potential impacts to a less than significant level, APM CR-2 would be implemented during construction.

# Impact CR-3: Directly or Indirectly Destroy a Unique Paleontological Resource or Site or Unique Geologic Feature (No Impact).

The proposed project would have no impact on paleontological resources because the alluvial deposits present are too geologically young to contain such resources. Likewise, the proposed project would have no impact on unique geologic features because none are present in the project area.

# Impact CR-4: Disturb any Human Remains, including Those Interred Outside of Formal Cemeteries (Less than Significant).

Although it would be unlikely for human remains to be disturbed during construction, APM CR-2 would be implemented during construction to ensure that potential impacts are kept to a less than significant level.

#### 4.5.3 References

1995

1978

Altschul, Jeffrey H., and Joseph A. Ezzo

Ceremony and Warfare Along the Lower Colorado River During the Protohistoric Period. In *Proceedings*, Volume 8, pp. 133–145. Society for California Archaeology, Chico.

Bean, Lowell John, and Sylvia Brakke Vane

Persistence and Power, A Study of Native American Peoples in the Sonoran Desert and the Devers-Palo Verde High Voltage Transmission Line. Cultural Systems Research, Inc., Menlo Park, California.

Bee, Robert L. 1983	Quechan. In <i>Southwest</i> , edited by Alfonso Ortiz, pp. 86–98. Handbook of North American Indians, Vol. 10, William C. Sturtevant, general editor. Smithsonian Institution, Washington, D.C.
Curtis, Edward S. 1906	The Yuma. Available at: http://www.worldwisdom.com/public/viewpdf/default.aspx?article- title=The_Yuma_Tribe_by_Edward_Curtis.pdf. Accessed on January 21, 2015.
Howell, Joseph 2014	A Class III Cultural Resources Survey for a Proposed Buried Telecommunications Fiber-Optic Line near Winterhaven, in Imperial County, California. Tierra Archaeological Report No. 2014-141. Tierra Right of Way Services, Ltd., Tucson.
Imperial County 2008	Imperial County General Plan. Imperial County Planning/Building Department, El Centro, California.
Kroeber, A.L. 1925	Handbook of the Indians of California. Bulletin 78 of the Bureau of American Ethnology of the Smithsonian Institution. Reprinted 1976. Dover Publications, Inc., New York.
McCorkle, Thoma 1978	s Intergroup Conflict. In <i>California</i> , edited by Robert F. Heizer, pp. 694–700. Handbook of North American Indians, Vol. 8, William C. Sturtevant, general editor. Smithsonian Institution, Washington, D.C.
Rea, Amadeo M. 1997	At the Desert's Green Edge: An Ethnobotany of the Gila River Pima. University of Arizona Press, Tucson.
Smith, Ian Michael 2010	From Subsistence to Dependence: The Legacy of Reclamation and Allotment on Quechan Indian Lands, 1700–1940. Master's thesis, on file at the University of Montana, Missoula.
Spicer, Edward H. 1962	Cycles of Conquest: The Impact of Spain, Mexico, and the United States on the Indians of the Southwest, 1533–1960. University of Arizona Press, Tucson.
1971	Persistent Cultural Systems. Science 174:795-800.
Stewart, Kenneth I 1983a	M. Yumans: Introduction. In <i>Southwest</i> , edited by Alfonso Ortiz, pp. 1–3. Handbook of North American Indians, Vol. 10, William C. Sturtevant, general editor. Smithsonian Institution, Washington, D.C.

1983b Mohave. In *Southwest*, edited by Alfonso Ortiz, pp. 55–70. Handbook of North American Indians, Vol. 10, William C. Sturtevant, general editor. Smithsonian Institution, Washington, D.C.

# 4.6 Geology, Soils, and Seismic Potential

# 4.6.1 Affected Environment

#### 4.6.1.1 Regulatory Setting

#### Federal

#### Clean Water Act Section 402p

Amendments to the CWA in 1987 added Section 402p, which created a framework for regulating municipal and industrial storm water discharges under the National Pollutant Discharge Elimination System (NPDES) program. In California, the State Water Resources Control Board (SWRCB) is responsible for implementing the NPDES program. Pursuant to the State's Porter-Cologne Water Quality Control Act, it delegates implementation responsibility to California's nine Regional Water Quality Control Boards (RWQCBs). The Colorado River Basin RWQCB has jurisdiction over the non-Tribal portion of the project area, whereas the EPA is responsible for implementing the NPDES program on the Tribal portion of the project area.

Under the NPDES Phase II Rule, any construction project disturbing 0.4 ha (1.0 acre) or more must obtain coverage under the State's Construction General Permit (CGP) for stormwater discharges associated with construction activity. The purpose of the Phase II Rule is to avoid or mitigate the effects of construction activities, including earthwork, on surface waters. To this end, CGP applicants are required to file a Notice of Intent to Discharge Stormwater with the RWQCB that has jurisdiction over the construction area and to prepare a Stormwater Pollution Prevention Plan (SWPPP) stipulating Best Management Practices (BMPs) that would be in place to avoid adverse effects on water quality.

#### State

# Alquist-Priolo Earthquake Fault Zoning Act

The legislature of the State of California passed the Alquist-Priolo Geologic Hazards Zone Act in 1972, renamed the Alquist-Priolo Earthquake Fault Zoning Act in 1994. The intent of the legislation was to limit the hazards of fault surface rupture to occupied structures. Active faults are those with evidence of displacement within the past 11,000 years (Holocene time). Those faults with evidence of displacement during Pleistocene time (11,000–2,000,000 years before present) are generally considered potentially active. In 1974, the California Division of Mines and Geology (currently known as the California Geological Survey) began establishing special study zones along known active faults termed earthquake fault zones. Starting in 1976, the California Division of Mine and Geology initiated the Fault Evaluation and Zoning Program to study faults identified in the Alquist-Priolo Earthquake Fault Zoning Act as "sufficiently active and well defined" to be considered for further evaluation. Fault Evaluation Reports were prepared for each earthquake fault zone (EFZ) summarizing data on fault location, age of activity, orientation, and probable magnitude of displacement.

#### Seismic Hazards Mapping Act

The Seismic Hazards Mapping Act addresses non-surface fault rupture earthquake hazards, including liquefaction and seismically induced landslides. Passed by the State Legislature in 1990, this law was codified in the PRC as Division 2, Chapter 7.8A, and became operative in April 1991.

#### Local

The Seismic and Public Safety Element of the ICGP identifies goals and policies that minimize the risks associated with natural and manmade hazards, and it specifies land use planning procedures that should be implemented to avoid hazardous situations. The purpose of the Seismic and Public Safety Element is directly concerned with reducing the loss of life, injury, and property damage that might result from disaster or accident.

#### 4.6.1.2 Project Setting

The proposed project is located within the Sonoran Desert section of the Basin and Range physiographic province. Within the Basin and Range Province, the Earth's crust (and upper mantle) has been stretched up to 100 percent of its original width. The entire region has been subjected to extension that thinned and cracked the crust as it was pulled apart, creating large faults. Along these roughly north-south-trending faults, mountains were uplifted and valleys fell, producing the distinctive alternating pattern of linear mountain ranges and valleys of the Basin and Range province.

The basin and range topography and the dry climate create a number of impressive features and landscapes that can be found throughout the province. These include pediments, alluvial fans, bajadas, bolsons, Inselbergs, playas, mud flats, salt flats, lakes, sand dunes, canyons, and the Rio Grande Rift. The vast region of the Basin and Range is divided into five distinct sections: Great Basin Section, Sonoran Desert Section, Salton Trough Section, Mexican Highland Section, and the Sacramento Section (NPS 2014). According to the Arizona Geological Survey, the entire project area is located on young river terrace and floodplain deposits associated with the historical Colorado River floodplain ( $Qy_2r$  map unit) (Youberg et.al. 2011).

Soils in the project area are of the Indio silt loam (13), Holtville clay (12), Gadsden clay (8), Lagunita silt loam (19), Kofa clay (17), Ripley silt loam (24), and Lagunita loamy sand (18) map units (NRCS 2013a). These soils are well drained to somewhat excessively drained and formed from mixed alluvium. The surface layer consists mostly of clay and silt loam and occasionally loamy sand. The Lagunita loamy sand and silt loam map units are not classified as prime farmland by the NRCS. All of the remaining soil types are classified as prime farmland if irrigated, but Gadsden and Holtville clays and Indio silt loam need to be reclaimed of excess salts and sodium before the prime farmland designation would apply (NRCS 2013b).

Most of the project corridors are located on clay soils with a relatively high shrink-swell potential. Soils with high shrink-swell potentials, also known as expansive soils, are primarily comprised of clay particles. Clay increases in volume when water is absorbed and shrinks when dry. Expansive soils can damage building foundations, concrete slabs, and road pavement as a result of swelling forces that reduce soil strength. In general, much of the near surface soils in the agricultural areas of the Imperial Valley, including the project site, consist of clays that are moderately to highly expansive (NRCS 1980).

The wind erodibility of these soils ranges from moderate to high and the K-factor for erodibility ranges from 0.1 to 0.55. The K-factor, which can range from 0.02 for the least erodible soils to 0.62 for the most erodible, is an index that quantifies the relative susceptibility of a soil to erosion by surface water flows. Medium-texture soils, including the Gadsden, Holtville and Kofa clays found in the project area, all have a moderate K-factor of 0.32, and fine-textured silty soils, such as the Indio and Ripley silt loams, have relatively high K-factors of 0.55 and 0.43, respectively. The coarse-textured Lagunita loamy sand in the project area has a low K-factor of 0.1 and is the least erodible soil present (NRCS 1980).

The principal fault system in Imperial County is the San Andreas Fault, located east of the proposed project area in the vicinity of the Salton Sea. The Algodones Fault is the closest major fault in this system in relation to the project area and is approximately 11.3 km (7.0 miles) to the southwest (Olmsted et. al. 1973). However, the proposed project area is not located in a mapped Alquist-Priolo Earthquake Fault Zone, and the County of Imperial is not identified as having any Seismic Hazards Zones according to the United States Geological Survey (USGS).

# 4.6.2 Environmental Effects

# 4.6.2.1 Significance Criteria

An impact related to geology, soils, or seismic potential was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice.

Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Expose people or structures to potential adverse effects, including the risk of loss, injury, or death involving:
  - Strong seismic ground shaking;
  - Seismic-related ground failure, including liquefaction; or
  - Landslides.
- Result in substantial soil erosion or the loss of topsoil.
- Be located on a geologic unit or soil that is unstable or that would become unstable as a result of the project and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction, or collapse.
- Be located on expansive soil, as defined in Table 18-1-B of the Uniform Building Code (1994), creating substantial risk to life or property.
- Have soils incapable of adequately supporting the use of septic tanks or alternative waste disposal systems where sewers are not available for the disposal of wastewater.

# 4.6.2.2 Impacts and Mitigation Measures

The proposed project would incorporate the following APMs and would not result in significant impacts on geology, soils, or seismic potential in the project area.

**APM GEO-1**: TDS will require the contractor to manage construction-induced sediment and excavated spoils in accordance with the requirements of the SWRCB and EPA NPDES permits for stormwater runoff associated with construction activities.

**APM GEO-2**: Prior to the onset of construction, TDS or its authorized contractor will complete a SWPPP that outlines BMPs to control discharges from construction areas.

**APM GEO-3**: No construction-related materials, wastes, spills, or residues will be discharged from the project.

**APM GEO-4**: The staging of construction materials, equipment, and excavation spoils will be performed outside of drainages.

**APM GEO-5**: Excavated or disturbed soil will be kept within a controlled area surrounded by a perimeter barrier that may entail silt fence, hay bales, straw wattles, or a similarly effective erosion-control technique that prevents the transport of sediment from a given stockpile.

**APM GEO-6**: All stockpiled material will be covered or contained in such a way that eliminates offsite runoff from occurring.

**APM GEO-7**: Upon completion of construction activities, excavated soil will be replaced and graded so that post-construction topography and drainage matches pre-construction conditions.

APM GEO-8: Surplus soil will be transported from the site and disposed of appropriately.

# Impact GEO-1: Expose People or Structures to Potential Adverse Effects, including the Risk of Loss, Injury, or Death Involving Strong Seismic Ground Shaking; Seismic-Related Ground Failure, including Liquefaction; or Landslides (No Impact).

The proposed project area is not located in an Alquist-Priolo earthquake fault zone, and Imperial County does not have any seismic hazard zones. Because these hazard zones are not present, and the majority of the proposed facilities to be installed would be buried, the proposed project would not expose people or structures to risks resulting from seismic activity. There would be no impacts.

#### Impact GEO-2: Result in Substantial Soil Erosion or the Loss of Topsoil (Less than Significant).

Plowing construction and the excavation of bore pits and DLC vault sites associated with the proposed project would loosen soil, which could contribute to soil erosion from wind and storm events. Per APM GEO-2, a SWPPP will be prepared that will detail BMPs to be implemented that would minimize or eliminate the potential soil erosion that could result from construction. Therefore, soil erosion and the loss of topsoil resulting from the proposed project would be held to less than significant levels.

# Impact GEO-3: Be Located on a Geologic Unit or Soil that is Unstable, or That Would Become Unstable as a Result of the Project, and Potentially Result in On- or Off-Site Landslide, Lateral Spreading, Subsidence, Liquefaction, or Collapse (No Impact).

The proposed project would involve the installation of buried fiber-optic lines and ancillary equipment including DLC sites consisting of buried vaults and aboveground equipment cabinets. Per APMs GEO-1–7, a SWPPP will be prepared prior to construction that will detail BMPS to be followed related to management of runoff, excavation and stockpiling, and post-construction site restoration. All soils disturbed during construction would be stabilized following construction by compacting to accepted engineering standards. Because of this, and the lack of topographical relief in the project area that would be conducive to landslides, there would be no impacts from on- or

off-site landslides, lateral spreading, subsidence, liquefaction, or collapse resulting from the proposed project.

# Impact GEO-4: Be Located on Expansive Soil, as Defined in Table 18-1-B of the Uniform Building Code (1994), Creating Substantial Risk to Life or Property (No Impact).

The proposed fiber-optic line installations would be located in an area having expansive soils with a high shrink-swell potential. Because the majority of the project's components would be buried, disturbed soils would be compacted following construction, and none of the aboveground installations would include large structures, there would be no impacts resulting in substantial risks to life or property due to the expansive soils present in the project area.

# Impact GEO-5: Have Soils Incapable of Adequately Supporting the Use of Septic Tanks or Alternative Waste Disposal Systems Where Sewers are Not Available for the Disposal of Wastewater (No Impact).

The proposed project does not include the installation of septic tanks or other waste disposal systems; therefore, there would be no impacts related to disposal of wastewater.

#### 4.6.3 References

National Park Service (NPS)

2014 Geologic, Energy, and Mineral Resources. Earth Science Concepts: Basin and Range Physiographic Province. Available at: http://www.nature.nps.gov/geology/education /concepts/concepts\_basinrange.cfm. Accessed on January 28, 2015.

National Resource Conservation Service (NRCS)

1980	Soil Survey, Yuma-Wellton Area, Parts of Yuma County, Arizona and Imperial County
	California. U.S. Department of Agriculture Soil Conservation Service in
	cooperation with the Arizona Agricultural Experiment Station and the California
	Agricultural Experiment Station, Washington D.C.

- 2013a NRCS Soil Survey, Yuma-Wellton Area, Parts of Yuma County, Arizona, and Imperial County California. Available at: http://websoilsurvey.nrcs.usda.gov/app/. Accessed on June 13, 2014.
- 2013b Web Soil Survey Map Unit Descriptions. Available at: http://websoilsurvey.nrcs.usda.gov/. Accessed on August 22, 2014.

Olmsted F.H, O.J. Loeltz, and Burdge Irelan

 1973 Geohydrology of the Yuma Area, Arizona and California, Water Resources of the Lower Colorado River – Salton Sea Area. Geological Survey Professional Paper 496-H. U.S. Department of the Interior Geological Survey, Washington D.C.

Youberg, Ann, J.E. Spencer, and P.A. Pearthree

2011 Geologic Map of the Yuma East 7.5-minute Quadrangle, Yuma County, Arizona. Digital Geologic Map DGM-86, version 1.0, scale 1:24,000. Arizona Geological Survey, Tucson.
#### 4.7 Hazards and Hazardous Materials

#### 4.7.1 Affected Environment

#### 4.7.1.1 Regulatory Setting

#### Federal

The EPA is the principal Federal regulatory agency responsible for the safe use and handling of hazardous materials. The key Federal regulations pertaining to hazardous wastes are described below. Other applicable Federal regulations are contained primarily in Titles 29, 40, and 49 of the CFR.

#### Toxic Substances Control Act

The Toxic Substances Control Act of 1976 (15 United States Code 2601 et seq.) authorizes the EPA to track industrial chemicals produced within or imported into the United States. Under this act, the EPA screens and tests industrial chemicals that pose a potential health hazard to humans or the environment. This act grants the EPA the authority to control and ban newly developed industrial chemicals and other chemicals that pose a risk in order to protect public and environmental health.

## Resource Conservation and Recovery Act/Comprehensive Environmental Response, Compensation, and Liability Act of 1980

The Resource Conservation and Recovery Act (RCRA) and the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 (CERCLA) address handling, disposal, and spill contingency measures for hazardous substances. The National Oil and Hazardous Substances Pollution Contingency Plan (NCP 40 CFR Part 300) specify the requirements for spill response activities. These laws and regulations apply to the proposed project installation activities conducted within the subject area.

#### Federal Aviation Administration Regulations

The FAA regulates the use of aircraft. The FAA requires a lift plan for the use of helicopters in populated areas. The lift plan serves to identify staging areas and flight paths that present the least potential to affect populated areas. The FAA regulates the flight distances for loaded and unloaded helicopters. Unloaded large helicopters (also called sky cranes) cannot fly within 46 lateral m (150 lateral feet) of an occupied structure at elevations where downdrafts can occur. Loaded sky cranes cannot fly within 91 lateral m (300 lateral feet) of an occupied structure. If the required distances cannot be maintained during the flight, structures must be unoccupied.

#### State

California hazardous materials and wastes regulations are equal to or more stringent than Federal regulations. The EPA has granted the State primary oversight responsibility to administer and enforce hazardous waste management programs. State regulations require planning and management to ensure that hazardous materials are handled, stored, and disposed of properly to reduce risks to human health and the environment. Several key State laws pertaining to hazardous materials and wastes are discussed below.

#### Hazardous Materials Release Response Plans and Inventory Act of 1985

The Hazardous Material Release Response Plans and Inventory Act, also known as the Business Plan Act, requires businesses using hazardous materials to prepare a plan that describes business facilities, inventories, emergency response plans, and training programs. Hazardous materials are defined as raw or unused materials that are part of a process or manufacturing step. They are not considered to be hazardous waste. Health concerns pertaining to the release of hazardous materials, however, are similar to those relating to hazardous waste.

#### Hazardous Waste Control Act

The Hazardous Waste Control Act created the State Hazardous Waste Management Program, which is similar to, but more stringent than, the Federal RCRA program. The act defines "hazardous wastes" as waste products with properties that make them dangerous or potentially harmful to human health or the environment. Hazardous wastes can be the byproducts of manufacturing processes or simply discarded commercial products, such as cleaning fluids or pesticides. The act is implemented by regulations set forth in CCR Title 26, which describes the following required parameters for the proper management of hazardous waste:

- Identification and classification.
- Generation and transport.
- Design and permitting of recycling, treatment, storage, and disposal facilities.
- Treatment standards.
- Operation of facilities and staff training.
- Closure of facilities and liability requirements.

These regulations list more than 800 materials that may be hazardous and establish criteria for identifying, packaging, and disposing of them. Under this act and CCR Title 26, a generator of hazardous waste must complete a manifest that accompanies the waste from the generator to the transporter to the ultimate disposal location. Copies of the manifest must be filed with the California Department of Toxic Substances Control (DTSC).

#### California Occupational Safety and Health Administration Standards

Worker exposure to contaminated soils, vapors that could be inhaled, or groundwater containing hazardous constituents is subject to the monitoring and personal safety equipment requirements established in Title 8 of the California OSHA regulations. The primary intent of the Title 8 requirements is to protect workers, but compliance with some of these regulations also reduces potential hazards to non-construction workers and project vicinity occupants through required controls related to site monitoring, reporting, and other activities.

#### California Environmental Protection Agency

CEPA implements and enforces a statewide hazardous materials program established by Senate Bill 1082 (1993) to consolidate, coordinate, and make consistent the administrative requirements, permits, inspections, and enforcement activities for the following environmental and emergency management programs for hazardous materials.

- Hazardous Materials Release Response Plans and Inventories (Business Plans).
- California Accidental Release Prevention Program.
- Underground Storage Tank Program.
- Aboveground Petroleum Storage Act Requirements for Spill Prevention, Control, and Countermeasure Plans.

- Hazardous Waste Generator and On-Site Hazardous Waste Treatment Programs.
- California Uniform Fire Code: Hazardous Materials Management Plans and Hazardous Material
- Inventory Statements.

#### Local

#### Certified Unified Program Agency

A Certified Unified Program Agency (CUPA) is a City or County agency certified by DTSC to conduct the Unified Program established by Senate Bill 1082 (as explained under CEPA). The Imperial County CUPA Department of Toxic Substances Control is the CUPA with jurisdiction in the vicinity of the project area.

#### Imperial County General Plan

<u>Fire</u>

The Imperial County Fire Prevention and Explosives Ordinance, Section 53101-53300, contains provisions for the purpose of prescribing regulations governing conditions hazardous to life and property from fire or explosion. Such measures in this ordinance include the following:

- Storage of flammable materials
- Storage of Radioactive materials
- Permit required for sale and use of fireworks
- Abatement of weeds and other vegetation

#### Hazards and Hazardous Materials

The ICGP has goals and objectives related to hazards and hazardous materials (Imperial County 2008). These goals and objectives are listed below.

- Goal 3: Protect the public from exposure to hazardous materials and wastes.
  - Objective 3.1—Discourage the transporting of hazardous materials/waste near or through residential areas and critical facilities.
  - Objective 3.2—Minimize the possibility of hazardous materials/waste spills.
  - Objective 3.3—Discourage incompatible development adjacent to sites and facilities for the production, storage, disposal, and transport of hazardous materials/waste as identified in the County General Plan and other regulations.
  - Objective 3.4—Adopt and implement ordinances, policies, and guidelines that assure the safety of County ground and surface waters from toxic or hazardous materials and wastes.

#### Winterhaven Urban Area Plan

The Winterhaven Urban Area Plan identifies the goals, policies, and standards that will guide the physical growth of the Winterhaven Urban Area, which consists of the Townsite of Winterhaven and surrounding areas. A goal and associated objectives in the plan related to hazards and hazardous materials include:

- Goal 2: Minimize potential hazards to public health, safety, and welfare and prevent the loss of life and damage to health and property resulting from both natural and human-related phenomena.
  - Objective 2.1—Ensure the adequacy of existing emergency preparedness and evacuation plans to deal with identified hazards and potential emergencies.
  - Objective 2.3—Minimize injury, loss of life, and damage to property by implementing all State codes where applicable.
  - Objective 2.4—Prevent and reduce death, injuries, property damage, and economic and social dislocation resulting from natural hazards, including flooding, land subsidence, earthquakes, other geological phenomena, levee or dam failure, urban and wildland fires, and building collapse by appropriate planning and emergency measures.

#### 4.7.1.2 Project Setting

The subject area is located within the existing ROW of public roads; therefore, a Phase 1 Site Assessment of the subject area corridor was not conducted. An Allands Data and Research, Inc. (Allands), Regulatory Database Corridor Study was prepared for the project alignment (Appendix F). Results of the Allands report indicate that there are three underground storage tanks (USTs) within 0.40 km (0.25 miles) of the project alignment. However, none of the USTs listed are reported to be in violation of any environmental regulations or pose a threat to public health and/or safety.

No other known regulated or unregulated hazardous waste generators, leaking tanks, toxic spills, or other sites affecting the environment are located in the proposed project area. There is no listed Superfund or other National Priorities List (NPL) site in the vicinity of the project area.

The nearest schools to the project area are Bill M. Manes High, San Pasqual Valley High School, San Pasqual Unified Middle School, San Pasqual Vocational Academy, and the San Pasqual Valley Elementary School, all located near the intersection of Arnold and Baseline Roads at 676 Baseline Road, Winterhaven, California 92283. These schools are located within 0.40 km (0.25 miles) of the project area.

The nearest public airport is the Yuma International Airport, approximately 4 km (6 miles) south of the project area.

According to the CAL FIRE Imperial County Fire Hazards Severity Zones map, the proposed project alignment is located within areas that have not been zoned because they are located within areas of local responsibility; however, surrounding areas are mapped as Local Responsibility Area (LRA) Moderate fire danger (California Department of Forestry and Fire Protection 2007). According to the ICGP, the potential for a major fire in the unincorporated areas of the County is generally low (Imperial County 2008).

#### 4.7.2 Environmental Effects

#### 4.7.2.1 Significance Criteria

An impact related to public health and safety was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice. Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Create a significant hazard to the public or the environment through the routine transport, use, or disposal of hazardous materials.
- Create a significant hazard to the public or environment through reasonably foreseeable upset and accident conditions involving the release of hazardous materials into the environment.
- Emit hazardous emissions or handle hazardous or acutely hazardous materials, substances, or waste within 0.40 km (0.25 miles) of an existing or proposed school.
- Be located on a site that is included on a list of hazardous materials sites compiled pursuant to Government Code Section 65962.5 and, as a result, would create a significant hazard to the public or the environment.
- Result in a safety hazard for people residing or working in an area subject to an airport land use plan or an area within 3.2 km (2.0 miles) of a public airport or private airstrip.
- Impair implementation of or physically interfere with an adopted emergency response plan or emergency evacuation plan.
- Expose people or structures to a significant risk of loss, injury, or death involving wildland fires, including where wildlands are adjacent to urbanized areas or where residences are intermixed with wildlands.

#### 4.7.2.2 Impacts and Mitigation Measures

Measures to avoid and minimize impacts from hazards and hazardous materials have been included in the APMs listed below. With implementation of the standard construction protocols and existing regulations, the proposed project would not result in significant impacts related to hazards and hazardous materials in the subject area of this PEA.

**APM HAZ-1**: TDS will ensure proper labeling, storage, handling, and use of hazardous materials in accordance with best management practices and OSHA's HAZWOPER requirements.

**APM HAZ-2**: TDS will ensure that employees are properly trained in the use and handling of hazardous materials and that each material is accompanied by a MSDS.

**APM HAZ-3**: Any small quantities of hazardous materials stored temporarily in staging areas will be stored on pallets within fenced and secured areas and protected from exposure to weather. Incompatible materials will be stored separately, as appropriate.

**APM HAZ-4**: All hazardous waste materials removed during construction will be handled and disposed of by a licensed waste disposal contractor and transported by a licensed hauler to an appropriately licensed and permitted disposal or recycling facility to the extent necessary to ensure the area can be safely traversed.

**APM HAZ-5**: Significant releases or threatened releases of hazardous materials will be reported to the appropriate agencies.

### Impact HAZ-1: Creation of a Significant Hazard to the Public or the Environment through the Routine Transport, Use, or Disposal of Hazardous Materials (Less than Significant).

Construction of the proposed project would involve small quantities of commonly used materials, such as fuels and oils, to operate construction equipment. However, because standard construction BMPs would be implemented to reduce the emissions of pollutants, this impact is considered less than significant (APM GEO-1 and 2). Spills of small quantities of hazardous wastes, such as waste oil, could be generated during construction and maintenance activities. However, potential impacts from accidents involving the release of small quantities of hazardous materials would be minimal due to the implementation of the proposed APMs (APM GEO-1 and 2; APM HAZ-1–5). Spill clean-up kits would be provided and kept on-site during construction, and equipment would remain in good working order to prevent spills. Therefore, impacts would remain less than significant.

## Impact HAZ-2: Creation of a Significant Hazard to the Public or Environment through Reasonably Foreseeable Upset and Accident Conditions Involving the Release of Hazardous Materials into the Environment (Less than Significant).

Potential impacts that could result from the proposed project include the risk of an oil or hazardous materials release from vehicle collisions, fires, damage to utility lines, and the general risks associated with installation. Construction activities would involve the operation of construction equipment and support vehicles within the project site. Construction of the project could also result in spills from accidents or the improper handling or disposal of fuels or hazardous materials, which could expose workers and the public to levels of hazardous materials in excess of OSHA and other applicable regulations. In addition to spills, small quantities of hazardous wastes, such as waste oil, could be generated during maintenance activities. However, potential impacts from accidents involving the release of small quantities of hazardous materials would be minimal due to the implementation of APMs. Spill clean-up kits would be provided and kept on-site during construction, and equipment would remain in good working order to prevent spills. Therefore, impacts would remain less than significant.

## Impact HAZ-3: Reasonable Anticipation to Emit or Handle Hazardous or Acutely Hazardous Materials, Substances, or Waste within 0.40 km (0.25 Miles) of an Existing or Proposed School (Less than Significant).

There are three schools located within 0.4 km (0.25 miles) of the proposed project. Given the types of materials used during construction (fuel, oils) and the minimal quantities that may be used, it is unlikely that any school would be affected by an accidental release of hazardous materials. However, potential impacts from accidents involving the release of small quantities of hazardous materials would be minimal due to the implementation of APMs. Spill clean-up kits would be provided and kept on-site during construction, and equipment would remain in good working order to prevent spills. Therefore, impacts would remain less than significant.

#### Impact HAZ-4: If the Project is Located on a Site that is Included on a List of Hazardous Materials Sites Compiled Pursuant to Government Code Section 65962.05 and, as a Result, Create a Significant Hazard to the Public or the Environment (No Impact).

According to the Allands Report, three sites of potential environmental concern are located within or adjacent to the project alignment. However, based on regulatory status, none of the sites are considered to represent a recognized environmental condition. In addition, the project alignment is not located on a Superfund or other NPL site. Therefore, the proposed project would not result in a significant hazard to the public or the environment through exposure to such sites. No impact is associated with this concern.

## Impact HAZ-5: If the Project Results in a Safety Hazard for People Residing or Working in an Area Subject to an Airport Land Use Plan or an Area within 3.2 km (2.0 Miles) of a Public Airport or Private Airstrip (No Impact).

The nearest public airport to the project alignment is the Yuma International Airport, located approximately 4 km (6 miles) southeast of the proposed project in Yuma, Arizona. The proposed project does not include installation of any new utility poles or increasing the height of the existing aerial distribution lines. Therefore, impacts associated with public airports are not anticipated.

### Impact HAZ-6: Impair Implementation of or Physically Interfere with an Adopted Emergency Response Plan or Emergency Evacuation Plan (Less than Significant).

Because project construction would occur within public road ROWs, traffic would need to be controlled and coordinated. Typically, traffic control would be set up for the day's work operation. One lane of traffic may need to be closed during work activities. During such periods, flaggers would be used to direct traffic in the construction zone. Delays to motorists would typically average 1–2 minutes. Traffic control measures would conform to CDOT specifications.

## Impact HAZ-7: Expose People or Structures to the Risk of Loss, Injury, or Death involving Wildland Fires, including Where Wildlands are Adjacent to Urbanized Areas or Where Residences are Intermixed with Wildlands (No Impact).

There are no wildlands present in or adjacent to the project area; consequently, there would be no impact related to the risk of loss, injury, or death involving wildland fires as a result of proposed project.

#### 4.7.3 References

California Department of Forestry and Fire Protection

2007 Fire Hazard Severity Zones in Imperial County. Draft. CAL FIRE, Sacramento.

#### Imperial County

2008

*Imperial County General Plan.* Imperial County Planning/Building Department, El Centro, California.

#### Allands Data and Research

2015 *Allands Regulatory Database Search Corridor Study.* Allands Data and Research, Goodyear, Arizona.

#### 4.8 Hydrology and Water Quality

#### 4.8.1 Affected Environment

#### 4.8.1.1 Regulatory Setting

#### Federal

Section 404 of the CWA authorizes the USACE to review and approve permit applications for the discharge of dredge and fill material within WUS, including wetlands. Section 10 of the River and Harbors Act requires project proponents to obtain a permit from USACE for construction or fill activities affecting the course, location, condition, or capacity of navigable waters. Section 401 of the CWA established national water quality goals and created the National Pollutant Discharge Elimination System (NPDES) to regulate water discharges and subsequent impacts to water quality.

Section 401 also provides States the opportunity to review and provide comment on Section 404 permit applications through a certification process for determinations of water quality standards compliance.

#### State

Acting under the leadership of the State Water Resources Control Board, RWQCBs protect the beneficial uses of surface water and groundwater in California under the Porter-Cologne Water Quality Control Act, with a focus on water quality. The RWQCBs regulate all pollutant or nuisance discharges that may affect either surface waters or ground Waters of the State. In cases where the waters are excluded from regulation under the CWA, the RWQCBs may still exercise jurisdiction over discharges into Waters of the State, pursuant to the Porter-Cologne Act in cases where the waters are excluded from regulation under the Federal CWA. In the absence of a legally approved formal protocol for delineating Waters of the State, all potential WUS as well as all isolated waters are considered Waters of the State. Stormwater discharges in the project area are regulated by the Colorado River Basin RWQCB.

#### Local

The Conservation/Open Space and Water Elements of the ICGP outline goals and objectives for the protection of water quality in the County (Imperial County 2008). Preservation of water resources in the Conservation/Open Space Element of the GP has the goal of conserving, protecting, and enhancing the water resources in the planning area with the following objectives applicable to the proposed project:

- Objective 8.1—Protect all bodies of water (e.g., the Salton Sea) and watercourses for their continued use and development.
- Objective 8.4—Ensure the use and protection of the rivers and other waterways in the County. Ensure proper drainage and provide accommodation for storm runoff from urban and other developed areas in manners compatible with requirements to provide necessary agricultural drainage.
- Objective 8.5—Protect and improve water quality and quantity for all water bodies in the County.
- Objective 8.6—Eliminate potential surface and groundwater pollution through regulations as well as educational programs.

Protection of surface waters in the Water Element of the GP has the goal of maintaining the longterm viability of the Salton Sea, Colorado River, and other surface waters in the County by protecting and sustaining wildlife and a broad range of ecological communities with the following objectives applicable to the proposed project:

- Objective 2.1—The continued viability of the agricultural sector as an important source of surface water for the maintenance of valuable wildlife and recreational resources in the County.
- Objective 2.2—A balanced ecology associated with the riparian and ruderal biological communities important as breeding and foraging habitats for native and migratory birds and animals occurring within the County.
- Objective 2.3—Preservation of riparian and ruderal habitats as important biological filters as breeding and foraging habitats for native and migratory birds and animals.

#### 4.8.1.2 Project Setting

The project area is located within the Yuma Valley Groundwater Basin (7-36), which is part of the Lower Colorado Watershed (Hydrologic Unit Code (HUC) 150301017) (EPA 2014), which is in turn part of the larger Colorado River hydrologic region. Historical data indicates that groundwater levels east and south of the All-American Canal, which includes the project area, have remained largely unchanged from 1962 through 2002 and range from 1.5– 6.1 m (5.0–20.0 feet) below the surface (CDWR 2004).

There are no perennial or ephemeral natural streams in the project area; however, 11 irrigation canals operated by the BOR's IID and BWD are crossed by the project corridors at 17 locations (see Appendix D).

Review of Federal Emergency Management Agency (FEMA) Flood Insurance Rate Map (FIRM) panels 06025C1900C, 06025C1925C, 06025C2250C, and 06025C2275C indicates that all of the project corridors are located in areas mapped as Zone X (FEMA 2015). Zone X areas are located outside the FEMA Special Flood Hazard Area because they are above the elevation of the 0.2 percent annual chance flood and have minimal flood hazard risk.

#### 4.8.2 Environmental Effects

#### 4.8.2.1 Significance Criteria

An impact related to hydrology and water quality was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice.

Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Violate any water quality standards or waste discharge requirements.
- Substantially deplete groundwater supplies or interfere substantially with groundwater recharge such that there would be a net deficit in aquifer volume or a lowering of the local groundwater table (e.g., the production rate of pre-existing nearby wells would drop to a level which would not support existing land uses or planned uses for which permits have been granted).
- Substantially alter the existing drainage pattern of the site or area, including through the alteration of the course of a stream or river, in a manner that would result in substantial erosion or siltation on- or off-site.
- Substantially alter the existing drainage pattern of the site or area, including through the alteration of the course of a stream or river, or substantially increase the rate or amount of surface runoff in a manner that would result in flooding on- or off-site.
- Create or contribute runoff water that would exceed the capacity of existing or planned stormwater drainage systems or provide substantial additional sources of polluted runoff.
- Otherwise substantially degrade water quality.
- Place housing in a 100-year flood hazard area as mapped on a Federal Flood Hazard Boundary or FIRM or other flood hazard delineation map.

- Place structures within a 100-year flood hazard area that would impede or redirect flood flows.
- Expose people or structures to a significant risk of loss, injury, or death involving flooding as a result of a failure of a levee or dam.
- Contribute to inundation by seiche, tsunami, or mudflow.

#### 4.8.2.2 Impacts and Mitigation Measures

The proposed project would incorporate APMs that would avoid or minimize impacts to water quality and waste discharge (see Section 4.6.2.2, APMs GEO-1–GEO 7). With incorporation of these APMs, there would be no significant impacts to hydrology and water quality.

#### Impact HYD-1: Violate Any Water Quality Standards or Waste Discharge Requirements (No Impact).

The proposed project would involve ground disturbance that has the potential for increasing sediment transport in the project area. Prior to the installations, TDS would obtain a NPDES permit from the Colorado River Basin RWQCB and develop a SWPPP including BMPs that would be implemented during construction (APM GEO-1 and GEO-2). These BMPs would include structural controls such as straw wattles and silt fencing, which would serve to contain sediment from disturbed areas that could be transported by storm events. Therefore, the proposed project would not violate water quality standards and there would be no impact.

Impact HYD-2: Substantially Deplete Groundwater Supplies or Interfere Substantially with groundwater Recharge such that there would be a Net Deficit in Aquifer Volume or a Lowering of the Local Groundwater Table (e.g., the production rate of pre-existing nearby wells would drop to a level which would not support existing land uses or planned uses for which permits have been granted) (No Impact).

During the proposed fiber-optic installations, fugitive dust from disturbed areas will be controlled by the application of water. The proposed project would not require substantial amounts of water during construction and would require no water during operation. Therefore, there would be no impact to groundwater supplies.

## Impact HYD-3: Substantially Alter the Existing Drainage Pattern of the Site or Area, including through the Alteration of the Course of a Stream or River, in a Manner that would Result in Substantial Erosion or Siltation On- or Off-Site (No Impact).

The proposed project involves the installation of buried fiber-optic lines; following the installations, the ground surface contours would be restored to their pre-construction condition (APM GEO-7). Therefore, drainage patterns would remain as they currently are, and no impacts to surface water flow would occur.

## Impact HYD-4: Create or Contribute Runoff Water that would Exceed the Capacity of Existing or Planned Stormwater Drainage Systems or Provide Substantial Additional Sources of Polluted Runoff (No Impact).

#### Impact HYD-5: Otherwise Substantially Degrade Water Quality (No Impact).

Prior to commencing the line installations, a SWPPP would be developed that will provide guidelines for implementing BMPs to control sediment transport (APMs GEO-1–7). These BMPs would ensure that no impacts from runoff water occur during construction and that water quality in the vicinity of the project area is maintained. There would be no impact.

Impact HYD-6: Place Housing in a 100-Year Flood Hazard Area as Mapped on a Federal Flood Hazard Boundary or Flood Insurance Rate Map or Other Flood Hazard Delineation Map (No Impact).

Impact HYD-7: Place Structures within a 100-Year Flood Hazard Area that would Impede or Redirect Flood Flows (No Impact).

### Impact HYD-8: Expose People or Structures to a Significant Risk of Loss, Injury, or Death Involving Flooding as a Result of a Failure of a Levee or Dam (No Impact).

The proposed project does not include the placement of housing. All of the proposed fiber-optic line installations would be buried, and the only aboveground structures to be installed would be DLC cabinets, splice boxes, and line markers. None of these structures, either above or below ground, would redirect flood flows, and the project area is not located in a flood hazard area. Therefore, there would be no impacts.

#### Impact HYD-9: Contribute to Inundation by Seiche, Tsunami, or Mudflow (No Impact).

The proposed project area is located inland and in an area with relatively flat topography; therefore, the proposed project would not contribute to the risk of inundation by seiche, tsunami, or mudflow. There would be no impact.

#### 4.8.3 References

2004

California Department of Water Resources (CDWR)

California Groundwater Bulletin 118, Yuma Valley Groundwater Basin. Available at: http://www.water.ca.gov/pubs/groundwater/bulletin\_118/basindescription/7-36.pdf. Accessed on January 26, 2015.

Environmental Protection Agency (EPA)

2014 Surf your Watershed. Available at: http://cfpub.epa.gov/surf/ county.cfm?fips\_code=06025. Accessed on December 19, 2014.

Federal Emergency Management Agency (FEMA)

2015 FEMA Food Map Service Center. Available at: http://msc.fema.gov/portal. Accessed on January 19, 2015.

#### Imperial County

2008

*Imperial County General Plan.* Imperial County Planning/Building Department, El Centro, California.

#### 4.9 Land Use and Planning

#### 4.9.1 Affected Environment

#### 4.9.1.1 Regulatory Setting

#### Federal

No Federal plans or policies related to land use or planning apply to the project.

#### State

#### California Public Utilities Commission

The CPUC has jurisdiction over the siting and design of the proposed project because the CPUC authorizes the construction and maintenance of investor-owned public utility facilities.

#### Local

The CPUC has primary jurisdiction over the proposed project because it authorizes the construction, operation, and maintenance of public utility facilities. Although the CPUC has the authority to preempt local agency permitting of the proposed project, they have not issued any decision broadly preempting such permitting. Therefore, the proposed project would have to meet local permitting requirements.

The entire project area is located within unincorporated Imperial County; however, portions of the project area are also located within the Winterhaven Urban Area and the Fort Yuma–Quechan Reservation. "Urban Areas" within unincorporated Imperial County are ICGP designations that provide for a range of permitted land uses within specific geographic areas (Imperial County 2008). Winterhaven has prepared a separate Land Use Plan (or "Urban Area Plan") that includes additional guidance on planning policy within the Winterhaven Urban Area boundary.

#### Imperial County General Plan and Zoning Regulations

Imperial County incorporates planning into their long-term development strategy through the implementation of the ICGP, which provides policies and objectives as well as specific land use designations, to guide the "distribution, general location, and extent of uses of land for housing, business, industry, open space, agriculture, and public facilities" within unincorporated Imperial County (Imperial County 2008).

The following local land use goals, objectives, and policies apply to the proposed project alignment:

- Goal 8: Coordinate local land use planning activities among all local jurisdictions and State and Federal agencies.
  - Objective 8.8—Ensure that the siting of future facilities for the transmission of electricity, gas, and telecommunications is compatible with the environment and County regulation.
  - Objective 8.9—Require necessary public utility ROWs when appropriate.

The following local land use goals, objectives, and policies apply to the land surrounding the proposed project alignment:

- Goal 1: Preserve commercial agriculture as a prime economic force.
- Goal 2: Diversify employment and economic opportunities in the County while preserving agricultural activity.
- Goal 3: Achieve balanced economic and residential growth while preserving the unique natural, scenic, and agricultural resources of Imperial County.
  - Objective 3.8—Utilize nonagricultural land as a resource to diversify employment opportunities and facilitate regional economic growth. Uses

must be consistent with each site's resource constraints, the natural environment, and the County Conservation and Open Space Element.

The project alignment is located within an existing transportation corridor. The project alignment is located adjacent to areas primarily carrying the zoning designations of Indian Reservation and Agriculture–General (A-2) with a small area zoned Light Commercial (C-1) located at the intersection of Perez Road and Ross Road. The portion of the alignment located within the Winterhaven Urban Area is located adjacent to Low-Density Residential, Medium-Density Residential, High-Density Residential, General Commercial, and Government/Special Public. Land use within the alignment is used as a transportation corridor. Land use adjacent to the project corridor is primarily agricultural, with the areas adjacent to the corridor in the Winterhaven Urban Area including government offices, commercial areas, and residential areas.

#### Winterhaven Urban Area Plan

The Winterhaven Urban Area Plan does not include any goals or objectives specifically related to telecommunicationd facilities.

#### 4.9.1.2 Project Setting

The project area is located within unincorporated Imperial County and includes the communities of Winterhaven, Bard, and Ross Corner as well as portions of the Fort Yuma–Quechan Reservation. The majority of the project area is used for agriculture, with small areas of residential and commercial properties located in the communities of Winterhaven, Bard, and Ross Corner. Existing development within the project area can be characterized as rural, sparse, and mostly limited to residences and buildings associated with agriculture. The communities of Winterhaven, Bard, and Ross Corner include more dense residential and commercial development.

The project area includes a school complex located near the intersection of Arnold and Baseline Roads that includes elementary, middle, high, and vocational schools. There are no public recreational facilities or designated open spaces in the project area; however, the school complex includes sports facilities.

#### 4.9.2 Environmental Effects

#### 4.9.2.1 Significance Criteria

An impact related to land use and planning was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice.

Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Physically divide an established community.
- Conflict with any applicable land use plan, policy, or regulation of an agency with jurisdiction over the project (including, but not limited to the General Plan, specific plan, local coastal program, or zoning ordinance) adopted for the purpose of avoiding or mitigating an environmental effect.
- Conflict with any applicable HCP or Natural Community Conservation Plan.

#### 4.9.2.2 Impacts and Mitigation Measures

#### Impact LU-1: Physically Divide an Established Community (No Impact).

The proposed project would be constructed along an existing public transportation corridor. The subject area is currently used as a public roadway, and other utilities are currently installed in these corridors. The use of this alignment for telecommunication network facilities is consistent with the current use of the subject area. The proposed project would retain existing land use designations.

Because the proposed telecommunication facilities would be built entirely within the existing utility corridor, the proposed project would not result in the physical division of an established community. There would be no impact.

# Impact LU-2: Conflict with any Applicable Land Use Plan, Policy, or Regulation of an Agency with Jurisdiction over the Project (including, but Not Limited to the General Plan, Specific Plan, Local Coastal Program, or Zoning Ordinance) Adopted for the Purpose of Avoiding or Mitigating an Environmental Effect (No Impact)

The CPUC has primary jurisdiction over the proposed project because it authorizes the construction, operation, and maintenance of public utility facilities. Although the CPUC has the authority to preempt local agency permitting of the proposed project, they have not issued any decision broadly preempting such permitting. Therefore, the proposed project would have to meet local permitting requirements. The proposed project would be co-located within existing utility ROWs, and project construction, design, and operational characteristics would be in compliance with the applicable Zoning Regulations. Because TDS would be required to acquire all necessary permits and conditions of approval from local jurisdictions, such as encroachment permits, and provide CPUC with appropriate documentation, there would be no impact.

## Impact LU-3: Conflict with Any Applicable Habitat Conservation Plan or Natural Community Conservation Plan (No Impact)

The proposed project alignment is located in an area addressed by the LCR MSCP; however, the proposed project does not conflict with the plan. There would be no impact to any applicable HCP or Natural Community Conservation Plan.

#### 4.9.3 References

Imperial County

- 2008 *Imperial County General Plan.* Imperial County Planning/Building Department, El Centro, California.
- 1996 *Winterhaven Urban Area Plan.* Imperial County Planning/Building Department, El Centro, California.

#### 4.10 Mineral Resources

#### 4.10.1 Affected Environment

#### 4.10.1.1 Regulatory Setting

#### Federal

The Mining and Mineral Policy Act of 1970 declared that the Federal government's policy is to encourage private enterprise in the development of a sound and stable domestic mineral industry. The Act also encourages orderly economic development of mineral resources and includes research and reclamation methods.

#### State

The Surface Mining and Reclamation Act of 1975 (SMARA) mandated the initiation by the State Geologist of mineral land classification in order to help identify and protect mineral resources in areas within the State subject to urban expansion or other irreversible land uses which would preclude mineral extraction. SMARA also allowed the State Mining and Geology Board (SMGB), after receiving classification information from the State Geologist, to designate lands containing mineral deposits of regional or statewide significance. Mineral commodities are mapped within jurisdictional boundaries, such as Counties, using the California Mineral Land Classification System.

The objective of classification and designation processes is to ensure, through appropriate lead agency policies and procedures, that mineral deposits of statewide or of regional significance are available when needed. The SMGB, based on recommendations from the State Geologist and public input, prioritizes areas to be classified and/or designated. Areas that are generally given highest priority are those areas within the State that are subject to urban expansion or other irreversible land uses that would preclude mineral extraction.

Classification is completed by the State Geologist, in accordance with the SMGB's priority list, by defining Mineral Resource Zones (MRZs) (defined below). Classification of these areas is based on geologic and economic factors without regard to existing land use and land ownership. The following MRZ categories are used by the State Geologist in classifying the State's lands:

- MRZ-1 Areas are where adequate geologic information indicates that no significant mineral deposits are present or where it is judged that little likelihood exists for their presence. This zone is applied where well-developed lines of reasoning, based on economic-geologic principles and adequate data, indicate that the likelihood for occurrence of significant mineral deposits is nil or slight.
- MRZ-2a Areas are underlain by mineral deposits where geologic data show that significant measured or indicated resources are present. Areas classified MRZ-2a contain discovered mineral deposits that are either measured or indicated reserves as determined by such evidence as drilling records, sample analysis, surface exposure, and mine information. Land included in the MRZ-2a category is of prime importance because it contains known economic mineral deposits.
- MRZ-2b Areas are underlain by mineral deposits where geologic information indicates that significant inferred resources are present. Areas classified MRZ-2b contain discovered deposits that are either inferred reserves or deposits that are

presently sub-economic as determined by limited sample analysis, exposure, and past mining history.

- MRZ-3a Areas contain known mineral deposits that may qualify as mineral resources. Further exploration work within these areas could result in the reclassification of specific localities into the MRZ-2a or MRZ-2b categories. MRZ-3a areas are considered to have a moderate potential for the discovery of economic mineral deposits.
- MRZ-3b Areas contain inferred mineral deposits that may qualify as mineral resources. Land classified MRZ-3b represents areas in geologic settings that appear to be favorable environments for the occurrence of specific mineral deposits. MRZ-3b is applied to land where geologic evidence leads to the conclusion that it is plausible that economic mineral deposits are present.
- MRZ-4 Areas are where geologic information does not rule out either the presence or absence of mineral resources. It must be emphasized that MRZ-4 classification does not imply that there is little likelihood for the presence of mineral resources, but rather that there is a lack of knowledge regarding mineral occurrence.

#### Local

The Conservation and Open Space Element of the ICGP contains goals and objectives to preserve mineral resources in the County. Figure 5 (Mineral Resources) of the Conservation and Open Space Element indicates the general location of known mineral resources in the County.

#### 4.10.1.2 Project Setting

A wide variety of minerals are found throughout Imperial County. Gold, gypsum, sand, gravel, lime, clay, and stone have the highest economic value and are presently extracted for profit in the County. Industrial materials are also readily available, including kyanite, mineral fillers (clay, limestone, sericite, mica, and tuff), salt, potash, calcium chloride, manganese, and sand (Imperial County 2008).

The proposed project is not located in a mapped MRZ area (CDOC 2015), and according to the ICGP mineral resources map, there are no known mineral resources in the vicinity of the project area. Given that the project area is located in the historic floodplain of the Colorado River, the only likely mineral resources present would be sand and gravel.

#### 4.10.2 Environmental Effects

#### 4.10.2.1 Significance Criteria

An impact related to mineral resources was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice.

Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Result in the loss of availability of a known mineral resource that would be of value to the region and the residents of the state.
- Result in the loss of availability of a locally important mineral resource recovery site delineated on a local General Plan, specific plan, or other land use plan.

#### 4.10.2.21mpacts and Mitigation Measures

Impact MIN-1: Result in the Loss of Availability of a Known Mineral Resource that would be of Value to the Region and the Residents of the State (No Impact).

## Impact MIN-2: Result in the Loss of Availability of a Locally Important Mineral Resource Recovery Site Delineated on a Local General Plan, Specific Plan, or Other Land Use Plan (No Impact).

The proposed project would involve the installation of buried fiber-optic lines within existing road ROWs. There would be no impacts to known mineral resources because none are located in the project area.

#### 4.10.3 References

California Department of Conservation

2015 Surface Mining and Reclamation Act Mineral Land Classification Maps. Available at: http://www.quake.ca.gov/gmaps/WH/smaramaps.htm. Accessed on April 17, 2015.

#### Imperial County

2008

Imperial County General Plan. Imperial County Planning/Building Department, El Centro, California.

#### 4.11 Noise

Sound (noise) occurs when an ear senses pressure variations or vibrations in the air. Noise is unwanted sound. A person's brain associates a subjective element to a sound, and an individual reaction is formed. Studies indicate that the most pervasive sources of noise in our environment today are those associated with transportation. The source of most outdoor noise is mainly caused by machines and transportation systems, motor vehicles, aircrafts, and trains.

The magnitude of noise is described by its sound pressure. Since the range of sound pressure varies greatly, a logarithmic scale is used to relate sound pressures to a common reference level, the decibel (dB). Because sound perception depends on the context in which the sound was generated and the characteristics of the sound, such as frequency duration, noise measurement refinements have been developed. These include the A-weighted decibel scale (dBA), which is weighted towards the portions of the sound frequency spectrum to which the human ear is most sensitive. Most equipment noise levels are expressed using the dBA scale.

Sound levels are often expressed in terms of an average noise level over time. The most commonly used short-term average is  $L_{eq}$ , the equivalent noise level. When  $L_{eq}$  is used, a time for averaging may be stated, such as 15 minutes, 1 hour, 8 hours, or 24 hours. If no time is stated, a one-hour average is assumed.  $L_{eq}$  is usually used in the description of noise near a point source or group of sources, such as a tractor or a construction site. Policies and ordinances that regulate noise at the source are usually stated in terms of  $L_{eq}$ .

The threshold of human hearing is assigned a dB level of zero. A normal conversation at a distance of 1.0–1.5 m (3.0–5.0 feet) produces about 60 dB. The conversation is not 60 times louder than the hearing threshold, it is a million times louder because the decibel scale is logarithmic (60 dB,  $10^6 = 1,000,000$ ). A table of common sound levels can be found in Table 4.7.

Sound Level (dB)	Community/Outdoor	Industry/ Home Indoor	Impression/Effect	
130				
	jet takeoff (at 61 m [200 feet])		threshold of pain (130–140 dB)	
120				
110	chainsaw (at 0.6 m [2.0 feet])	nightclub		
100	pile driver (at 15 m [50 feet])			
90	power mower, heavy truck (at 15 m [50 feet])	boiler room	hearing damage (8-hour exposure)	
80	concrete mixer (at 15 m [50 feet])	garbage disposal	loud/annoying	
70	freeway (at 30.5 m [100.0 feet])	noisy restaurant	shouting required at 0.9 m (3 feet)	
60	air conditioner unit	department store	loud speech required at 0.9 m (3 feet)	
50	light vehicle traffic (at 30.5 m [100.0 feet])	quiet office	normal speech at 0.9 m (3 feet), disturbs sleep	
40	bird calls	library	quiet	
	soft whisper (at 1.8 m [6.0 feet])			
30		quiet bedroom		
20	North Rim of Grand Canyon	recording studio		
10			threshold of hearing	

 Table 4.7. Common Sound Levels

Source: Imperial County General Plan, Noise Element (2008).

#### 4.11.1 Affected Environment

#### 4.11.1.1 Regulatory Setting

Federal, State, and local bodies of government establish laws and regulations to control excessive noise and reduce human noise exposure to a level that is acceptable within their jurisdiction. While Federal and State laws regulate transportation noise, establish "normally" and "conditionally" acceptable exterior noise limits based on land use type, and establish maximum acceptable interior noise limits for residences, no Federal or State provisions regulate noise levels relating to temporary construction activity. Construction noise is generally regulated at the local or County-wide level.

#### Federal

No Federal regulations relating to noise are applicable to this project.

#### State

No State regulations relating to noise are applicable to this project.

#### Local

The Noise Element of the ICGP provides a program for incorporating noise issues into the land use planning process, with a goal of minimizing adverse noise impacts to receptors that are sensitive to noise.

#### Construction Noise Standards

Construction noise, from a single piece of equipment or a combination of equipment, shall not exceed 75 dB  $L_{eq}$  when averaged over an eight-hour period and measured at the nearest sensitive receptor. This standard assumes a construction period, relative to an individual sensitive receptor, of days or weeks. In cases of extended-length construction times, the standard may be tightened so as not to exceed 75 dB  $L_{eq}$  when averaged over a one-hour period.

Construction equipment operation shall be limited to the hours of 7 a.m. to 7 p.m. Monday through Friday and 9 a.m. to 5 p.m. on Saturday. No commercial construction operations are permitted on Sunday or holidays. In cases of a person constructing or modifying a residence for himself/herself, and if the work is not being performed as a business, construction equipment operations may be performed on Sundays and holidays between the hours of 9 a.m. and 5 p.m. Such noncommercial construction activities may be further restricted where disturbing, excessive, or offensive noise causes discomfort or annoyance to reasonable persons of normal sensitivity residing in an area (Imperial County 2008).

#### 4.11.1.2 Project Setting

The majority of the proposed project is located in a rural agricultural area with scattered residences, and concentrated residential areas are present in Winterhaven and Bard. The San Pasqual Valley school complex located at Arnold and Baseline Roads would be considered a sensitive receptor, as would the scattered rural residences and residential areas in Winterhaven and Bard.

Existing noise sources in the proposed project area include agricultural equipment, vehicular traffic, and trains on the UPRR. Typical sound levels for the existing noise sources found in the project area, normalized to a reference distance of 15 m (50 feet), can be found in Table 4.8.

Noise Source	Sound Level <sup>a</sup>
Agricultural equipment	67–82 dBA (Bean 2008)
Light vehicular traffic	56 dBA (Imperial County 2008)
Train (horn at road crossings)	116 dBA maximum (USDOT 2009)
Train (locomotive and cars)	83–91 dBA (USDOT 2009)

Table 4.8. Existing Noise Sources in the Project Area

<sup>a</sup> Sound levels were normalized using the equation:  $dB_x = dB_{ref} + 20 \log (d_{ref} / d_x)$ , where  $dB_x$  is the decibel level at distance "x",  $dB_{ref}$  is the decibel level at the reference distance,  $d_{ref}$  is the reference distance, and  $d_x$  is the distance that the desired decibel level,  $dB_x$ , is to be calculated for.

#### 4.11.2 Environmental Effects

#### 4.11.2.1 Significance Criteria

An impact related to noise was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice.

Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Result in exposure of persons to or generation of noise levels in excess of standards established in the local General Plan, by noise ordinance, or by applicable standards of other agencies.
- Result in exposure of persons to or generation of excessive ground-borne vibration or ground-borne noise levels.
- Result in a substantial permanent increase in ambient noise levels in the project vicinity above levels existing without the project.
- Result in a substantial temporary or periodic increase in ambient noise levels in the project vicinity above levels existing without the project.

#### 4.11.2.2 Impacts and Mitigation Measures

Measures to avoid and minimize impacts from noise have been included in the APMs listed below. With implementation of the standard construction protocols and existing regulations, the proposed project would not result in significant impacts related to noise in the subject area of this PEA.

**APM NOI-1**: All construction equipment operation shall be limited to the hours of 7 a.m. to 7 p.m. Monday through Friday and 9 a.m. to 5 p.m. on Saturday. No construction operations shall occur on Sunday or holidays.

## Impact NOI-1: Result in Exposure of Persons to or Generation of Noise Levels in Excess of Standards Established in the Local General Plan, by Noise Ordinance, or by Applicable Standards of Other Agencies (Less than Significant).

During construction, equipment operation would be the primary noise source associated with construction activities and could affect noise sensitive receptors adjacent to the project area. Section 3.6.5 above lists the typical construction equipment that would be needed for the various construction activities. The construction activities would occur on weekdays only (APM NOI-1) and the anticipated construction schedule for each activity is listed in Section 3.6.6.

The Federal Highway Administration (FHWA) has compiled data regarding the noise-generating characteristics of specific types of construction equipment. The typical average maximum noise levels for construction equipment measured at a distance of 15 m (50 feet) are depicted in Table 4.9. Noise levels from equipment shown in Table 10 decrease with distance from the construction site at a rate of approximately 6 dBA per doubling of distance. The noise levels shown in Table 4.9 represent the construction equipment's averaged maximum noise levels, operating under full load conditions. However, most construction equipment operates in alternating cycles of full power and low power, and during varying periods of time. Consequently, the average sound level at

construction sites is typically less than the equipment's maximum noise levels. Noise generated by construction equipment during the proposed project's construction would occur with varying intensities and durations during the various phases of construction.

Equipment	Maximum Noise Level (dBA) at 15 m (50 feet)
Bulldozer	82
Directional boring machine	83
Backhoe	78
Mud sucker	81
Skid steer loader	79
Medium-duty truck (5 ton)	76
Air compressor	78
Pickup	75

Table 4.9. Construction Equipment Noise Levels

Source: 2011 FHWA Construction Noise Handbook, actual measured sound levels, samples averaged

Noise levels at receiving properties are dependent on several factors, including the number of machines operating within an area at a given time and the distance between the source(s) and receiving properties. The nearest sensitive receptors along the project corridors include residences in Winterhaven that are as close as 4.6 m (15.0 feet). Rural residences in the remaining portions of the project area are no closer than 9 m (30 feet) to the project corridors, and the school complex at Arnold and Baseline is approximately 38 m (125 feet) away from the project corridor at that location. Typically, the average noise level generated from the proposed construction activities would range from 75–83 dBA when measured at a distance of 15 m (50 feet) from the construction area. These noise levels from construction equipment are within the same range as that normally produced by agricultural equipment in the project area.

Noise generated by construction activities, therefore, could result in noise levels, at the closest sensitive receptors, exceeding the County's 75 dB  $L_{eq}$  (8-hour) averaged noise standard indicated above. However, during the installations, construction equipment would be constantly moving and would not remain at any one location for an extended amount of time. In addition, all construction would occur on weekdays and during the hours specified in the County standards. Therefore, while the project would generate noise for a very short term during construction, the noise levels would not exceed the 8-hour thresholds of the local standards. The impact would be less than significant.

## Impact NOI-2: Result in Exposure of Persons to or Generation of Excessive Ground-Borne Vibration or Ground-Borne Noise Levels (Less than Significant).

Most of the proposed project installation would be conducted using plowing construction techniques, which produce only negligible ground-borne vibration. For the areas where the proposed line would be installed using directional boring, some amount of vibration may be generated. As described in the discussion of Impact NOI-1 above, construction activities would take place for a matter of hours a limited number of days at any one location, and construction hours would conform to local regulation (APM NOI-1). The impact would be less than significant.

Impact NOI-3: Result in a Substantial Permanent Increase in Ambient Noise Levels in the Project Vicinity Above Levels Existing without the Project (No Impact).

The proposed project would not result in a permanent increase in ambient noise levels in the project vicinity because the installed facilities, consisting of buried fiber-optic lines, equipment cabinets and vaults, and markers, would produce no noise. There would be no impacts.

## Impact NOI-4: Result in a Substantial Temporary or Periodic Increase in Ambient Noise Levels in the Project Vicinity Above Levels Existing without the Project (Less than Significant).

As discussed in Impact NOI-1 above, noise generated by project construction would be limited to a few hours in a day on several nonconsecutive days at each location. Since existing noise sources in the project area include vehicular traffic, the railroad, and agricultural equipment, with noise generation taking place very close to the identified sensitive receptors, construction equipment noise would not raise ambient noise levels substantially. The impact would be less than significant.

#### 4.11.3 References

#### Bean, Thomas

- 2008
- Noise on the Farm Can Cause Hearing Loss. Available at: http://ohioline.osu.edu/aex-fact/pdf/ AEX\_590\_08.pdf. Accessed on April 9, 2015.

#### Federal Highway Administration

2011 Construction Noise Handbook. Available at: http://www.fhwa.dot.gov/environment /noise/construction\_noise/handbook/handbook09.cfm. Accessed on April 8, 2015.

#### Imperial County

2008

- *Imperial County General Plan.* Imperial County Planning/Building Department, El Centro, California.
- U.S. Department of Transportation (USDOT)

2009 Handbook for Railroad Noise Measurement and Analysis. Available at: http://www.fra.dot.gov/eLib /Details/L03061. Accessed on April 9, 2015.

#### 4.12 Population and Housing

#### 4.12.1 Affected Environment

#### 4.12.1.1 Regulatory Setting

Implementation of the proposed project would occur entirely within existing ROWs and would not involve the acquisition of any property or the relocation of any existing residents, businesses, or other uses. Consequently, Federal and State policies related to relocation assistance and real property acquisition would not apply to this project.

#### State

State law requires each City and County to adopt a General Plan for its future growth. This plan must include a housing element that identifies housing needs for all economic segments and provide

opportunities for housing development to meet those needs. At the State level, the Housing and Community Development Department estimates the relative share of California's projected population growth that would occur in each county presented by the Department of Finance's demographic research unit.

Each City and County must update its General Plan housing element on a regular basis (usually every five years). Among other things, the housing element must incorporate policies and identify potential sites that would accommodate the City's and County's share of the regional housing need. The applicable County housing element, part of the ICGP, is described below.

#### Local

The Housing Element of the ICGP is a comprehensive assessment of current and future housing needs in the County and proposed actions to facilitate the provision of housing to meet those needs for households of all income levels. Policies contained in this element are an expression of the statewide housing priority to allow for the "attainment of decent housing and a suitable living environment for every Californian," as well as a reflection of the unique needs and concerns of the County community. The purpose of the Housing Element is to establish specific goals and policies relative to the provision of housing and to adopt an action plan toward this end. In addition, the element identifies and analyzes housing needs and resources, as well as constraints to housing development (Imperial County 2013).

#### 4.12.1.2 Project Setting

The majority of the proposed project is located in a rural agricultural area with scattered residences, and concentrated residential areas are present in Winterhaven and Bard. The closest residences in relation to the project corridors are located in Winterhaven at a distance of approximately 4.6 m (15.0 feet). Rural residences in the remaining portions of the project area are no closer than 9.1 m (30.0 feet) to the project corridors

#### 4.12.2 Environmental Effects

#### 4.12.2.1 Significance Criteria

An impact related to population and housing was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice.

Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Induce substantial population growth in an area either directly (for example, by proposing new homes and businesses) or indirectly (for example, through extension of roads or other infrastructure).
- Displace substantial numbers of existing housing units, necessitating the construction of replacement housing elsewhere.
- Displace substantial numbers of people, necessitating the construction of replacement housing elsewhere.

#### 4.12.2.2Impacts and Mitigation Measures

## Impact POP-1: Induce Substantial Population Growth in an Area Either Directly or Indirectly (No Impact)

The proposed project would not induce population growth. Implementation of the project would provide a service to existing rural residents, businesses, and schools. Construction activities would last only a few weeks and would not generate new permanent jobs in the region. There would be no impact related to population growth.

### Impacts POP-2: Displace Existing Housing and/or People, Resulting in Relocation and/or the Construction of Replacement Housing Elsewhere (No Impact)

The proposed project consists of installing telecommunications facilities within existing ROWs along County roads. Project implementation would not displace existing housing or people and therefore would not require relocation or construction of replacement housing elsewhere. There would be no impact related to displacement of housing and/or people.

#### 4.12.3 References

Imperial County

2013

County of Imperial 2014–2021 Housing Element. Imperial County, El Centro, California.

#### 4.13 Public Services/Utilities and Service Systems

#### 4.13.1 Affected Environment

#### 4.13.1.1 Regulatory Setting

There are no applicable Federal or local policies related to public services or utilities for the proposed project.

#### State

CPUC regulates privately owned telecommunications, electric, natural gas, water, railroad, rail transit, and passenger transportation companies in California. CPUC is responsible for ensuring that California utility customers have safe, reliable utility service at reasonable rates, protecting utility customers from fraud and promoting the health of California's economy. CPUC establishes service standards and safety rules and authorizes utility rate changes. CPUC enforces CEQA compliance for utility construction.

#### 4.13.1.2 Project Setting

Police protection in the proposed project area is provided by the Quechan Tribal Police Department and the Imperial County Sherriff. Fire protection is provided by the Winterhaven Fire Department.

The project corridors are located along County and BIA roads, many of which include existing utility easements with aerial electrical distribution lines and buried telecommunications and water lines.

As mentioned in Section 3.3, wired Internet service in the proposed project area is limited to dial-up and is only available in TDS's four existing DSAs. Cellular data service (3G, 4G, and 4GLTE) from Verizon, AT&T, and Sprint is available in portions of the project area, as is HughesNet satellite Internet service.

The San Pasqual Valley School District complex at Arnold and Baseline Roads currently receives Internet connectivity through a microwave link from a station located west of the project area at Pilot Knob. This link provides 54 MBPS Internet service to the school, but the District has expressed a desire for a faster fiber-optic broadband connection (SPVUSD 2008).

#### 4.13.2 Environmental Effects

#### 4.13.2.1 Significance Criteria

An impact related to public services was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice.

Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Result in substantial adverse physical impacts associated with the provision of new or physically altered government facilities or the need for new or physically altered government facilities (the construction of which could cause significant environmental impacts) in order to maintain acceptable service ratios, response times, or other performance objectives for any of the following public services:
  - Fire protection
  - Police protection
  - Schools
  - Parks
  - Other public facilities
- Exceed wastewater treatment requirements of the applicable RWQCB.
- Require or result in the construction of new water or wastewater treatment facilities or expansion of existing facilities (the construction of which could cause significant environmental effects).
- Require or result in the construction of new stormwater drainage facilities or expansion of existing facilities (the construction of which could cause significant environmental effects).
- Require new or expanded entitlements for water supplies if existing water supplies available for the project from existing entitlements and resources are insufficient.
- Result in a determination by the wastewater treatment provider that serves or may serve the project that it has inadequate capacity to serve the project's projected demand in addition to the provider's existing commitments.
- Be served by a landfill with insufficient capacity to accommodate the project's solid waste disposal needs.
- Not be in compliance with Federal, State, and local statutes and regulations related to solid waste.

#### 4.13.2.2 Impacts and Mitigation Measures

Impact PSU-1: Result in Adverse Physical Impacts Affecting Service Ratios, Response Times, or Other Performance Objectives for any Public Service (Fire and Police Protection, Schools, Parks, or Other Public Facilities (Less than Significant Impact).

The proposed project consists of installation and operation of facilities to improve the capacity and reliability of the area's telecommunications system and would therefore have no effects on the demand for schools, parks, or other public facilities. Construction activities are not expected to result in the need for new or physically altered governmental facilities or to affect service ratios, response times, or other performance objectives for any public services. The impact on service ratios, response times, and other performance objectives would be less than significant.

### Impact PSU-2: Exceed Wastewater Treatment Requirements of the Colorado River Basin RWQCB (No Impact).

The proposed project does not include any facilities or uses associated with the generation of wastewater. The proposed project would therefore have no impact on wastewater treatment requirements.

## Impact PSU-3: Require or Result in the Construction of New Water or Wastewater Treatment Facilities or Expansion of Existing Facilities, the Construction of Which Could Cause Significant Environmental Effects (No Impact).

The proposed project would not require or result in the construction of any new water or wastewater treatment facilities or the expansion of existing facilities. There would be no impact.

## Impact PSU-4: Require or Result in the Construction of New Stormwater Drainage Facilities or Expansion of Existing Facilities, the Construction of Which Could Cause Significant Environmental Effects (No Impact).

The proposed project involves the placement of telecommunications facilities within existing utility ROWs and would not generate a need for expansion or construction of new stormwater drainage facilities. There would be no impact.

### Impact PSU-5: Require New or Expanded Entitlements for Water Supplies if Existing Water Supplies Available for the Project from Existing Entitlements and Resources are Insufficient (No Impact).

Construction activities would incorporate standard ICAPCD construction measures specified in Regulation VIII to reduce fugitive dust emissions, including the use of water for dust suppression. Water needed for dust suppression would be provided to the project contractor by local municipal water sources, such as those found in Winterhaven. The contractor would obtain the quantity of water needed for a day's operations prior to arriving on site. Because there would be little ground disturbance associated with the project, only a small amount of water (between 500 and 1,000 gallons per week) would be required. There would be no increase in demand for new or expanded entitlements to provide sufficient water supplies following construction. There would be no impact.

## Impact PSU-6: Result in a Determination by the Wastewater Treatment Provider which Serves or May Serve the Project that it has inadequate Capacity to Serve the Project's Projected Demand in Addition to the Provider's Existing Commitments (No Impact).

Neither construction nor operation of the proposed project would generate additional wastewater in the project area. There would be no impact.

## Impact PSU-7: Be Served by a Landfill with Insufficient Capacity to Accommodate the Project's Solid Waste Disposal Needs (Less than Significant).

Following construction, the proposed project is not expected to generate solid waste. Minimal amounts of solid waste would be generated during construction, and TDS and/or their contractors would recycle this material to the extent possible and/or properly dispose of it. No new landfill capacity would be necessary, and any impacts would be less than significant.

## Impact PSU-8: Conflict with Federal, State, and Local Statutes and Regulations Related to Solid Waste (No Impact).

The proposed project would be implemented in compliance with all Federal, State, and local statutes and regulations related to solid waste. There would be no impact.

#### 4.13.3 References

San Pasqual Valley Unified School District (SPVUSD)

2008 Technology Use Plan, School Years 2008–2011. Technology Committee, San Pasqual Valley Unified School District, Winterhaven, California.

#### 4.14 Recreation

#### 4.14.1 Affected Environment

#### 4.14.1.1 Regulatory Setting

There are no applicable Federal, State, or local policies related to recreation for the proposed project.

#### 4.14.1.2 Project Setting

There are no parks or opportunities for recreational activities in the vicinity of the proposed project area.

#### 4.14.2 Environmental Effects

#### 4.14.2.1 Significance Criteria

An impact related to recreation was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice.

Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Increase the use of existing neighborhood and regional parks or other recreational facilities such that substantial physical deterioration of the facility would occur or be accelerated.
- Include recreational facilities or require the construction or expansion of recreational facilities that might have an adverse physical effect on the environment.

#### 4.14.2.2Impacts and Mitigation Measures

Impact REC-1: Increase the Use of Existing Neighborhood and regional parks or Other Recreational Facilities such that Substantial Physical Deterioration of the Facility would Occur or be Accelerated (No Impact).

The proposed project would not increase the use of any neighborhood or regional parks or any other recreational resources because none are located in the vicinity of the project area. The proposed project would not lead to any increases in population, and therefore would not require construction or expansion of recreational facilities. There would be no impact.

### Impact REC-2: Include Recreational Facilities or Require the Construction or Expansion of Recreational Facilities which Might have an Adverse Physical Effect on the Environment (No Impact).

The proposed project does not include construction or expansion of any recreational facilities. There would be no impact.

#### 4.15 Transportation and Traffic

#### 4.15.1 Affected Environment

#### 4.15.1.1 Regulatory Setting

#### Federal

There are no applicable Federal policies related to transportation and circulation for the proposed project.

#### State

State law requires each City and County to adopt a comprehensive, long-range General Plan, including a circulation element, to guide its physical development. The applicable County circulation documents are described below.

#### Local

#### Imperial County General Plan

The ICGP, as the County's fundamental land use and development policy document, establishes goals and polices related to the county's transportation network. The ICGP contains the following relevant transportation goal (Imperial County 2008):

• Goal 1: The County will provide and require an integrated transportation system for the safe and efficient movement of people and goods within and through the County with minimum disruption to the environment.

#### Winterhaven Urban Area Plan

The Winterhaven Urban Area Plan identifies the goals, policies, and standards that will guide the physical growth of the Winterhaven Urban Area, which consists of the Townsite of Winterhaven and surrounding areas. A goal and associated objectives in the plan related to transportation include:

- Goal 1: The County will provide an integrated transportation system for the safe and efficient movement of people and goods within and throughout the Winterhaven Urban Area with minimum disruption to the environment.
  - Objective 1.1—Maintain and improve the existing road and highway network, while providing for future expansion and improvement based on travel demand and the development of alternative travel modes.
  - Objective 1.2—Ensure safe and coordinated traffic patterns, continuous growth, and promote a planned and consistent development around the township area.
  - Objective 1.3—Finance or seek funding for circulation system maintenance projects.

#### 4.15.1.2 Project Setting

The primary transportation thoroughfares in the region consist of I-8, which is the primary east-west route through Imperial County between San Diego, California, and Yuma, Arizona, and Interstate Business 8 (also called Winterhaven Drive), which provides business access to the Winterhaven community from I-8. The double-track UPRR runs parallel to and north of Winterhaven Drive in the southern portion of the project area. Roads within the project area consist primarily of two-lane minor collector roadways and residential streets.

#### Existing Roadway Network

The proposed project is located in a rural, unincorporated area of the County within the existing ROW of the local roadway system. These roadways provide access to land uses within the local region and connect local streets with the I-8 and State Highway 80 roadways. According to the County's 2013 Transportation Plan Update, there are currently no roadways in the project area identified as problem roadways in the Congestion Management Element of the plan (Imperial County 2013).

#### Transit

The Quechan Tribe, in partnership with the Yuma County Intergovernmental Public Transportation Authority (YCIPTA), provides local fixed-route bus service in Winterhaven and on Fort Yuma–Quechan Reservation lands (Yuma County Intergovernmental Public Transportation Authority 2015). In addition, there is a three-day-per-week route operating between eastern Imperial County (Winterhaven) and Downtown El Centro, California. Services are provided under contract to First Transit, Inc. (Imperial Valley Transit 2015).

The San Pasqual Unified School District provides bus services for the local community for the school day and after-school activities. Buses operate in the morning and afternoon.

#### 4.15.2 Environmental Effects

#### 4.15.2.1 Significance Criteria

An impact related to transportation and traffic was considered potentially significant under CEQA if the project would result in any of the following environmental effects. The criteria are based on Appendix G of the State CEQA Guidelines and professional practice. Appendix G of the State CEQA Guidelines indicates that an impact is considered significant if the project would:

- Conflict with an applicable plan, ordinance, or policy establishing measures of effectiveness for the performance of the circulation system, taking into account all modes of transportation including mass transit and non-motorized travel and relevant components of the circulation system, including but not limited to intersections, streets, highways and freeways, pedestrian and bicycle paths, and mass transit.
- Conflict with an applicable congestion management program, including but not limited to level-of-service standards and travel-demand measures or other standards established by the County congestion management agency for designated roads or highways.
- Result in a change in air traffic patterns, including either an increase in traffic levels or a change in location that results in substantial safety risks.
- Substantially increase hazards due to a design feature (e.g., sharp curves or dangerous intersections) or incompatible uses (e.g., farm equipment).
- Result in inadequate emergency access.
- Conflict with adopted policies, plans, or programs regarding public transit, bicycle, or pedestrian facilities, or otherwise decrease the performance or safety of such facilities.

#### 4.15.2.2 Impacts and Mitigation Measures

Measures to avoid or minimize traffic impacts would be followed during construction in accordance with the APMs below. These measures would comply with or require construction contractors to comply with the relevant emergency access and temporary traffic-control requirements identified by the California Department of Transportation (CDOT) and/or the County where appropriate. With implementation of these APMs and existing regulations, the proposed project would not result in significant impacts related to traffic in the subject area of this PEA. No additional measures are needed.

**APM TRA-1**: TDS will require the project contractor to obtain all necessary local, State, and BIA road encroachment permits prior to construction and will comply with all the applicable conditions of approval.

**APM TRA-2**: As deemed necessary by the applicable jurisdiction, the road encroachment permits may require the contractor to prepare a traffic control plan in accordance with professional engineering standards prior to construction.

**APM TRA-3**: TDS will develop circulation and detour plans to minimize impacts to local street circulation. This will include the use of signing and flagging to guide vehicles through and/or around the construction zone.

APM TRA-4: TDS will schedule truck trips outside of peak morning and evening commute hours.

APM TRA-5: TDS will limit lane closures during peak hours to the extent possible.

**APM TRA-6**: TDS will include detours for bicycles and pedestrians in all areas potentially affected by project construction.

**APM TRA-7**: TDS will install traffic control devices as specified in the *California Department of Transportation Manual of Traffic Controls for Construction and Maintenance Work Zones.* 

**APM TRA-9**: The contractor will coordinate with local transit agencies for the temporary relocation of routes or bus stops in work zones as necessary.

Impact TRA-1: Conflict with an Applicable Plan, Ordinance, or Policy Establishing Measures of Effectiveness for the Performance of the Circulation System, Taking into Account all Modes of Transportation including Mass Transit and Non-Motorized Travel and Relevant Components of the Circulation System, including but not limited to Intersections, Streets, Highways and Freeways, Pedestrian and Bicycle Paths, and Mass Transit (Less than Significant)

The proposed project would not conflict with any applicable circulation plan, ordinance, or policy establishing measures of effectiveness for the circulation system's performance. Construction traffic would be present on a temporary basis and would be similar to ongoing activities occurring in the subject area, including local travel and ranch and farm activities. Therefore, this would be a less-than-significant impact.

Construction activities would occur along existing transit and school bus routes and may require temporary traffic control and temporary closure of one lane of traffic. Although minimal work within travel lanes is anticipated, when the construction zone must take over a travel lane, a lane of traffic would be closed to provide traffic control for the work zone. Lane or shoulder closures would be short-term and would occur only during construction hours. TDS will coordinate with local transit agencies prior to construction (APM TRA-9). Therefore, this would be a less-than-significant impact.

## Impact TRA-2: Conflict with an Applicable Congestion Management Program, including, but not limited to Level-of-Service Standards and Travel-Demand Measures, or Other Standards Established by the County Congestion Management Agency for Designated Roads or Highways (Less than Significant).

Implementation of the proposed project would not conflict with the ICGP, Circulation and Scenic Highways Element, the applicable congestion management program for the area. Construction traffic associated with the proposed project would not be substantial enough to affect local roadway performance levels, and there would be no long-term effect on roadway traffic. This would be a less-than-significant impact.

### Impact TRA-3: Result in a Change in Air Traffic Patterns, including Either an Increase in Traffic Levels or a Change in Location that Results in Substantial Safety Risks (No Impacts).

The proposed project involves the installation of buried telecommunications facilities and does not include installation of any new utility poles. No change in air traffic patterns would be associated with the proposed project. There would be no impact.

## Impact TRA-4: Substantially Increase Hazards Due to a Design Feature (e.g., sharp curves or Dangerous Intersections) or Incompatible Uses (e.g., Farm Equipment) (Less than Significant)

The proposed project consists of the installation of new buried fiber-optic cable. Operation of the project would not involve any hazardous changes to roadways or their uses. Because the project alignment is primarily located within public road ROWs, traffic would need to be controlled and coordinated to avoid a hazardous situation during construction activities.

Construction equipment to be used for the proposed installations would be highly maneuverable and would use existing improved areas for turning around or parking, such as existing roads, field access aprons, driveway aprons, or farm roads. For some construction activities, it may be necessary to close one traffic lane. At least one lane of traffic would be open at all times. Traffic control would be implemented in accordance with CDOT specifications as presented in Chapter 5 of their traffic manual, *Traffic Controls for Construction and Maintenance Work Zones*, even when not on State highways. Flaggers would direct traffic in the construction zone. Delays to motorists would typically average 1–2 minutes. Lane or shoulder closures would be short-term and would occur only during construction hours. In addition, TDS would ensure all APMs are followed to avoid and minimize transportation and traffic effects are implemented. Therefore, this would be a less-than-significant impact.

#### Impact TRA-5: Result in Inadequate Emergency Access (Less than Significant)

Because the project alignment is primarily located within or near public road ROWs, traffic would need to be controlled and coordinated during some construction activities. Although minimal work within travel lanes is anticipated, when the construction zone must take over a travel lane, a lane of traffic would be closed to provide traffic control for the work zone. Lane or shoulder closures would be short-term and would occur only during construction hours. All traffic-control measures would conform to CDOT specifications as presented in Chapter 5of their traffic manual, *Traffic Controls for Construction and Maintenance Work Zones*. Therefore, this would be a less-than-significant impact.

## Impact TRA-6: Conflict with Adopted Policies, Plans, or Programs Regarding Public Transit, Bicycle, or Pedestrian Facilities, or Otherwise Decrease the Performance or Safety of Such Facilities (Less than Significant)

The proposed project would not conflict with any adopted alternative transportation policies, plans, or programs. However, construction activities will occur along existing transit and school bus routes and may require temporary traffic control and temporary closure of one lane of traffic. Although minimal work within travel lanes is anticipated, when the construction zone must take over a travel lane, a lane of traffic would be closed to provide traffic control for the work zone. All traffic control measures would conform to CDOT specifications as presented in Chapter 5 of their Traffic Manual, *Traffic Controls for Construction and Maintenance Work Zones.* TDS would coordinate with local transit agencies for temporary relocation of routes or bus stops in work zones prior to any lane closures (APM-9). Therefore, this impact would be less than significant.

Construction activities on or near the roadway shoulder could temporarily affect bicycle or pedestrian travel within the proposed project alignment. Construction activities in any individual location would be of short duration and would not encroach on the roadway; they therefore would not require redirection of motorists, bicyclists, or pedestrians. In compliance with the *California Manual on Uniform Traffic Control Devices* (MUTCD), bicycle traffic, like motorists, would be provided "reasonably safe passage through the [temporary traffic control] zone" (CDOT 2012). As part of the project construction activities, warning signs and notices would be posted to properly warn bicyclists utilizing the roadway of potential hazards on or near the shoulder (APM TRA-6). This impact would be less than significant.

#### 4.15.3 References

Imperial County	
1996	<i>Winterhaven Urban Area Plan.</i> Imperial County Planning/Building Department, El Centro, California.
2008	<i>Imperial County General Plan.</i> Imperial County Planning/Building Department, El Centro, California.
2013	Imperial County 2013 Transportation Plan. Imperial County Transportation Commission, El Centro, California.
Imperial Valley T	ransit
2015	Imperial Valley Transit Bus Routes. Available at: http://www.ivtransit.com/riderguide/offline/download.pdf. Accessed on April 5, 2015.
Yuma County Int	ergovernmental Public Transportation Authority
2015	YCAT Blue Route 5—Quechan Shuttle (5). Available at: http://www.ycipta.org /routes/5. Accessed on April 10, 2015.

#### 4.16 Growth-Inducing and Cumulative Impacts

#### 4.16.1 Growth-Inducing Impacts

Growth-inducing effects could occur if a project would induce growth either directly or indirectly in the surrounding environment. Typically, the growth-inducing potential of a project would be considered significant if it fosters growth or a population concentration above what is assumed or planned for in local and regional land use plans or in projections made by regional planning groups. Significant growth-inducing impacts could also occur if the project provides infrastructure or service capacity to accommodate growth levels beyond those permitted by local plans and/or policies. Growth and development within Imperial County is managed at the local and County level and is anticipated to occur consistent with general and specific plans prepared and approved by each jurisdiction.

The proposed project is not expected to induce growth. Rather, it would allow TDS to provide broadband telecommunications services to currently underserved areas.

The proposed project could also be considered growth-inducing if growth results from the direct and indirect employment needed to construct, operate, and maintain the project. The proposed project would not require full-time personnel on site, and construction work would be temporary and of short duration. Inspection and maintenance activities would occur only periodically. Therefore, the proposed project would not generate growth associated with direct or indirect employment for construction, operation, or maintenance of the project. There would be no growth-inducing effects associated with the proposed project.

#### 4.16.2 Indirect Impacts

Indirect impacts, also referred to as secondary impacts, are impacts caused by a project that occur later in time or are farther removed in distance, but are still reasonably foreseeable. Indirect effects may include growth-inducing impacts and the impacts that result from this growth related to a change in the pattern of land use, population density, or growth rate and the resulting effects on air and water and other natural systems.

As noted above, the proposed project is not anticipated to induce growth. Rather, it would allow TDS to provide broadband telecommunications services, as required by CPUC, to current and future customers in the area. Growth and development within Imperial County is managed at the local and County level and is anticipated to occur consistent with general and specific plans prepared and approved by each jurisdiction. Therefore, to ensure that adequate telecommunications services are available to serve existing and planned development, the proposed project would be considered an essential utility.

Future development in Imperial County must occur consistent with the applicable General Plan, specific plans, and related environmental documentation, and development in the vicinity of the proposed project area is expected to be minimal. The agricultural land present is considered Prime Farmland (see Section 4.2) and is therefore protected from development into other uses. Likewise, the Winterhaven Area Plan indicates that development in that community is anticipated to be minimal and consist primarily of infill on existing lots. This project would not influence planned or future developments. Development of the proposed project is not anticipated to result in any indirect impacts on land use, population density, growth rate, or natural systems or resources in the project area. No long-term indirect changes or growth of any kind can be reasonably attributed solely to the proposed project.

#### 4.16.3 Cumulative Impacts

Cumulative impacts are defined in State CEQA Guidelines Section 15355 as "two or more individual effects which, when considered together, are considerable or which compound or increase other environmental impacts." A cumulative impact occurs from "the change in the environment, which results from the incremental impact of the project when added to other closely related past, present, and reasonably foreseeable probable future projects. Cumulative impacts can result from individually minor, but collectively significant, projects taking place over a period of time" (State CEQA Guidelines Section 15355[b]).

For the purposes of this PEA, cumulative impacts on resources in the general project vicinity may result from closely related projects either in close physical or temporal proximity that could add incrementally to any potential impacts of the proposed project. The Imperial County Public Works Department Projects Out-to-Bid List was reviewed for relevant present and future projects (Imperial County 2015). The only future project in the vicinity of the proposed project is for sidewalk installations along the east side of Baseline Road between Arnold and San Pasqual Roads (project #5405). The proposed project alignment on Arnold Road would be installed on the north side of Arnold Road, opposite the northern terminus of the sidewalk project. No other projects are located at or near the project site that would add to potential circulation impacts, and as such would not result in cumulatively considerable impacts.

As discussed in Section 4.3, Air Quality, the project area is currently in non-attainment for the criteria pollutants PM10 and  $O_3$ ; however, the estimated emissions levels from the proposed project during construction for both PM10 and ROG are both well below the established ICAPCD thresholds. Consequently, because the proposed project's anticipated emissions of these two criteria

pollutants that are in non-attainment are below what ICAPCD would consider significant, any cumulative impacts would be considered less than significant.

#### 4.16.4 References

Imperial County

2015 Imperial County Public Works Department. Available at: http://www.co.imperial.ca.us/ publicwork/PublicWorkUser/ProjectsOutToBid/ProjectsOutToBid.htm. Accessed on April 20, 2015.

#### 5.0 NEPA-SPECIFIC RESOURCE AREA IMPACT ANALYSIS

The environmental analysis presented in this section takes into consideration the potential effects on the natural environment that may result from BIA's approval of a ROW grant that would allow the Tribal portion of the proposed project to be constructed. Approval of the ROW grant would constitute a Federal action; therefore, the environmental consequence analysis uses the correct NEPA term "Proposed Action," which is analogous to the CEQA "proposed project".

NEPA documents typically require an alternatives analysis where impacts of a Proposed Action are compared to impacts from alternative ways to accomplish the purpose and need of the project. Impacts of not implementing the Proposed Action are known as the No-Action Alternative, which serves as the baseline conditions present in the analysis area from which to make comparisons. Due to the limited potential for adverse impacts associated with the Proposed Action, BIA has determined that a comprehensive alternatives analysis was not required for the Winterhaven project, and only the impacts of the Proposed Action and No-Action Alternative need be considered in this PEA (Chip Lewis, BIA, personal communication, 1/26/2015).

The 2012 BIA NEPA Handbook includes a list of resource areas that require consideration in BIA NEPA documents. An initial screening analysis of these resource areas and how they relate to the Proposed Action is presented in Table 5.1. This analysis was performed to eliminate duplicate analyses in this combined NEPA/CEQA document.

Resource Area	Finding	
Land Resources		
Topography (land forms, drainage, gradients)	Addressed in Section 4.6, Geology, Soils, and	
Soils (types, characteristics)	Seismic Potential	
Geologic Setting, Mineral, and Paleontological Resources	Addressed in Section 4.6; Section 4.10, Mineral Resources; and Section 4.5, Cultural Resources	
Water Resources (surface and ground; quality, quantity, use, rights)	Addressed in Section 4.8, Hydrology and Water Quality	
Air (quality/achievement, visibility)	Addressed in Section 4.3, Air Quality and Greenhouse Gases	
Living Resources		
Wildlife (terrestrial, aquatic, Threatened and Endangered)	addressed in Section 4.4, Biological Resources	

Table 5.1. DIA INEFA Resource Area	Table 5	5.1. BIA	NEPA	Resource	Areas
------------------------------------	---------	----------	------	----------	-------

Resource Area	Finding	
Vegetation (terrestrial, aquatic, riparian, threatened/endangered)		
Ecosystems and Biological Communities)		
Agriculture (livestock, crops, prime and unique farmland)	addressed in Section 4.2, Agricultural Resources	
Cultural Resources		
Historic, Cultural, and Religious Properties		
Archaeological Resources	addressed in Section 4.5, Cultural Resources	
Socioeconomic Conditions		
Employment and Income		
Demographic Trends	addressed in Section 5.3, Socioeconomics	
Lifestyle and Cultural Values	addressed in Section 4.5	
Community Infrastructure (public services, utilities)	addressed in Section 4.13, Public Services/Utilities and Service Systems	
Environmental Justice	addressed in Section 5.2, Environmental Justice	
Resource Use Patterns		
Hunting, Fishing, Gathering	addressed in Section 4.5	
Timber Harvesting		
Agriculture	addressed in Section 4.2, Agricultural Resources	
Mining	addressed in Section 4.10	
Recreation	addressed in Section 4.14, Recreation	
Transportation Networks	addressed in Section 4.15, Transportation and Traffic	
Land Use Plans	addressed in Sections 4.1-4.15	
Other Values		
Wilderness	there is no designated wilderness in or in the vicinity of the project area and no concerns were identified	
Noise and Light	addressed in Section 4.1, Aesthetics, and Section 4.11, Noise	
Visual	addressed in Section 4.1, Aesthetics, and Section 4.11, Noise	
Public Health and Safety	addressed in Section 4.3 and Section 4.7, Hazards and Hazardous Materials	
Climate Change (Greenhouse Gases)	addressed in Section 4.3	
Indian Trust Assets	addressed in Section 5.1, Indian Trust Assets	
Hazardous Materials	addressed in Section 4.7	

#### 5.1 Indian Trust Assets

Indian Trust Assets (ITAs) are legal interests in assets that are held in trust by the United States government for Federally recognized Tribes or American Indian individuals. The trust relationship usually stems from a treaty, Executive Order, or act of Congress. The Secretary of the Interior is the trustee for the United States on behalf of Federally recognized Tribes. "Assets" are anything owned
that holds monetary value. "Legal interests" refers to a property interest for which there is a legal remedy (such as a compensation or injunction) if there is improper interference. Assets can be real property, physical assets, or intangible property rights (such as a lease or right to use something). ITAs cannot be sold, leased, or otherwise alienated without approval from the United States. Trust assets may include lands, minerals, natural resources, and hunting, fishing, and water rights. American Indian reservations, rancherias, and public domain allotments are examples of lands that are often considered ITAs. In some cases, ITAs may be located off trust land.

BIA shares the Indian trust responsibility with other agencies of the Executive Branch to protect and maintain ITAs reserved by or granted to Tribes or American Indian individuals by treaty, statute, or Executive Order.

## 5.1.1 Affected Environment

The portions of the proposed project area on the Fort Yuma–Quechan Reservation are located on Tribal allotments that are Indian Trust Assets. Each of the allotments is approximately 4 ha (10 acres) in size and can have anywhere from 1 to well over 100 Tribal members that have an ownership interest in the allotment.

## 5.1.2 Environmental Consequences

## 5.1.2.1 Proposed Action

The Proposed Action would allow TDS to install and maintain fiber-optic lines on approximately 62 Tribal land allotments through the grant of a 1.5-m-wide (5.0-foot-wide) ROW with a term of 50 years. In total, the ROWs would encompass approximately 1.5 ha (3.8 acres) of Tribal land. Allottees would retain legal ownership and title to their land, which they could continue to use as they see fit, provided that their use does not interfere with TDS's use of the ROW for the fiber-optic lines. Therefore, the Proposed Action would have a minor, long-term, adverse impact on Indian Trust Assets.

## 5.1.2.2 No-Action Alternative

The No-Action Alternative would result in no impacts to Indian Trust Assets in the Proposed Action Area.

# 5.2 Environmental Justice

Title VI of the Civil Rights Act of 1964 and related statutes (including the State-level Environmental Evaluation Group's "Guidance on Title VI and Environmental Justice") ensure that individuals are not excluded from participation in, denied the benefit of, or subjected to discrimination under any program or activity receiving Federal financial assistance on the basis of race, color, national origin, age, sex, or disability. Executive Order 12898, Federal Actions to Address Environmental Justice in Minority Populations and Low-Income Populations, dated February 11, 1994, directs Federal agencies to identify and address the disproportionately high and adverse human health or environmental effects of their actions on minority and low-income populations to the greatest extent practicable and permitted by law. The order also directs each agency to develop a strategy for implementing environmental justice. The order is also intended to promote nondiscrimination in Federal programs that affect human health and the environment, as well as provide minority and low-income communities with access to public information and public participation.

### 5.2.1 Affected Environment

According to 2010 U.S. Census, the majority of the residents in the three Census Block Groups (CBGs) in the Proposed Action area are minorities. Households with incomes below the Federal poverty level in the Proposed Action area range from a high of 44.7 percent for CBG 06025940003 to a low of 23.6 percent for CBG 060259400002, with CBG 060259400001 in the middle at 32.2 percent (CPUC 2014).

### 5.2.2 Environmental Consequences

### 5.2.2.1 Proposed Action

The Proposed Action would be constructed within the ROW of existing public roadways and would not result in adverse human health or environmental effects on the local population. Because the Proposed Action would not disproportionately affect any one socioeconomic group more so than any other, there would be no adverse impact related to Environmental Justice in the Proposed Action area.

### 5.2.2.2 No-Action Alternative

Because the No-Action Alternative would result in no changes to existing conditions, the No-Action Alternative would result in no adverse impacts related to Environmental Justice.

## 5.3 Socioeconomics

### 5.3.1 Affected Environment

#### 5.3.1.1 Employment and Income

According to the U.S. Census Bureau's 2006–2010 American Community Survey (ACS), the unemployment rates in the project area range from a low of 4.1 percent in CBG 60259400003, which includes Winterhaven, to a high of 10.4 percent in 60259400002, which includes the Tribal portion of the project area and a small portion of unincorporated Imperial County. The unemployment rate of CBG 60259400001, located in unincorporated Imperial County, is in the middle at 6.7 percent. According to the ACS 2010 estimates, the median household income in the project area is \$29,111 for CBG 60259400001, \$25,179 for 60259400002, and \$18,929 for CBG 60259400003 (CPUC 2014).

#### 5.3.1.2 Demographic Trends

In 2010, the most populous CBG in the project area, 60259400001, had a population of 1,322. The other two CBGs in project area, 60259400002 and 6025900003, had populations of 944 and 786, respectively (CPUC 2014).

Between 2000 and 2010, the population of the Fort Yuma–Quechan Reservation dropped from 2,376 to 2,197, a decline of 7.5 percent. The majority of residents on the Reservation identify themselves as American Indian and Alaska Native, with 61 percent reporting in 2010. The Hispanic or Latino population on the Reservation was 32 percent in 2010 and residents identifying themselves as white made up 23 percent of the population in 2010 (ARPI 2010).

### 5.3.2 Environmental Consequences

### Proposed Action

The Proposed Action would bring telecommunication services to areas that are currently underserved. The inclusion of or improvement to communication services in the project area would enhance the quality of life of participating residents. The installation of the telecommunications facilities is not likely to immediately create new jobs, but may make the area more desirable to live and work in because of the availability of telecommunication services. The Proposed Action would therefore result in a minor beneficial impact related to Socioeconomics.

### No-Action Alternative

Because the potential benefits of the enhanced telecommunications services would not be brought to an area that is currently underserved, the No Action Alternative would result in a minor adverse impact to socioeconomics.

# 5.4 References

Arizona Rural Policy Institute (ARPI)

2010 Demographic Analysis of the Fort Yuma–Quechan Reservation Using 2010 Census and 2010 American Community Survey Estimates. Arizona Rural Policy Institute, Center for Business Outreach, W.A. Franke College of Business, Northern Arizona University, Flagstaff.

California Public Utilities Commission (CPUC)

2014 California Broadband Availability Mapper. Available at: http://www.broadbandmap.ca/gov/map. Accessed on December 5, 2014.

# 6.0 LIST OF PREPARERS

Tierra, in cooperation with TDS, prepared this PEA. Members of the project team who prepared the document are listed below.

# 6.1 Tierra Right of Way Services, Ltd.

Tom Euler, Environmental Planning/Cultural Resources Director Joseph Howell, Archaeologist Tim Jordan, Senior Biologist Theresa Knoblock, Biologist/Environmental Compliance Lead Barbara Montgomery, Ph.D., Principal Investigator/Project Manager John Murray, Technical Editor

# 6.2 TDS Telecommunications, Inc.

Joe Kirk, Manager Garin Mayer, Engineer Branden Piper, Network Specialist Nate Stanislawski, Manager